THREATS OF EXTREME WEATHER EVENTS

Improving the Resilience of QwaQwa to the Multiple Risks of Climate Change

Volume 1

Report to the Water Research Commission

by

M.F. Avenant¹, J.A. Belle², H. Börnick³, D. Jungmann^{1,4}, B. Opeolu⁵, P. Voua Otomo⁶, N. Davis⁷ and R. Schulze^{7,8}

¹ Centre for Environmental Management, University of the Free State
 ² Disaster Management Training and Education Centre for Africa, University of the Free State
 ³ Institute of Water Chemistry, Technical University Dresden
 ⁴ Institute of Hydrobiology, Technical University Dresden
 ⁵ Faculty of Applied Sciences, Cape Peninsula University of Technology
 ⁶ Department of Zoology, University of the Free State
 ⁷ Isikhungusethu Environmental Services (Pty) Ltd
 ⁸ Centre for Water Resources Research, University of KwaZulu-Natal

WRC Report No. 3091/1/23 ISBN 978-0-6392-0538-0

September 2023



Obtainable from Water Research Commission Bloukrans Building, 2nd Floor Lynnwood Bridge Office Park 4 Daventry Street Lynnwood Manor PRETORIA

orders@wrc.org.za or download from www.wrc.org.za

This report forms part of a set of two reports emanating from the WRC project no. C2019/2020-00239). The other report is *Improving the Resilience of Qwaqwa to the Multiple Risks of Climate Change from a Hydro-Climatic Perspective. Volume 2: of the Project "Threats of Extreme Weather Events: Improving the Resilience of QwaQwa to the Multiple Risks of Climate Change"* (WRC Report No. 3091/2/23).

DISCLAIMER

This report has been reviewed by the Water Research Commission (WRC) and approved for publication. Approval does not signify that the contents necessarily reflect the views and policies of the WRC, nor does mention of trade names or commercial products constitute endorsement or recommendation for use.

© Water Research Commission

EXECUTIVE SUMMARY

BACKGROUND

Recent climate projections for the Southern African region strongly indicate that the occurrence of rainfall extremes and prolonged drought periods may increase in the near future. Such extreme events pose a severe threat to water resources, food security, public health, infrastructure, biological diversity, and ecological systems in the region. Changes to the temporal dynamics of flow regimes in rivers as a result of these projected changes could hold important consequences for both humans and riverine biota.

While current research on climate extremes generally focuses on relatively large catchments with a low spatial resolution, information is needed on smaller catchments at detailed spatial resolution as to whether the climate projections for Southern Africa actually display increasing probabilities of sequential extreme events of, for example, torrential rainfall following directly after an extended drought period. Sequential extremes could have severe consequences by triggering cascades of secondary effects such as high sediment yields with impacts on, for example, infrastructural damage to water purification plants.

To improve the resilience against such impacts and to develop efficient coping strategies, new approaches that integrate the assessment of exposure and vulnerability on the one hand, and offer efficient coping strategies on the other, are needed. However, such adaptation strategies must be based on appropriate and reliable downscaled information and include communication technologies with early warning and decision support systems for the population and decision makers concerned. Such risk reduction is all the more important given that weak governance could further increase the exposure to these risks. This failure could hold dire consequences for already vulnerable communities and sectors and, to a large extent, counter socio-economic development efforts by the government and other non-governmental agencies.

AIMS

The overarching aim of this project was to understand how the cumulative impacts of successive extreme weather events may affect the exposure and resilience of local communities in QwaQwa and how these impacts may be ameliorated through risk reduction planning. The objectives of the project were:

- 1. To determine how the ecological services of river ecosystems in QwaQwa could be impacted by an increase in the occurrence of extreme weather events;
- 2. To develop an understanding of the impact that secondary effects may have on service delivery and infrastructure development in QwaQwa;
- 3. To identify the most important hazards and risks for the QwaQwa community in terms of water delivery, human health, and environmental safety;
- 4. To prepare a tailor-made risk assessment for the QwaQwa area based on downscaled climate and hydrology models specifically developed for the area;
- 5. To develop a tailor-made disaster risk reduction (DRR) strategy and plan for QwaQwa.

STUDY AREA

The focal area of this study comprised the Phuthaditjhaba urban area and the surrounding villages that make up 30 wards in the area. It is situated in the north-eastern Free State province, bordering KwaZulu-Natal to the south-east and Lesotho on the south-west. The area is located in a relatively narrow valley in a section of the Maloti-Drakensberg mountain range, at an elevation of between 1 675 m to more than 2 200 m above sea level. The mountainous area serves as the headwaters for several streams, including the Mphukojwane, Namahadi, and Metsi-Matso rivers. These streams joins to form the Elands River, and later the Wilge River, a major tributary of the Vaal River. The study catchment forms part the northern Drakensberg strategic water source area.

QwaQwa is one of the four towns in the Maluti-a-Phofung Local Municipality and serves as the administrative headquarters of both the Maluti-a-Phofung Local Municipality and the Thabo Mofutsanyana District Municipality. Phuthaditjhaba, the main urban area, is densely populated and surrounded by informal settlements. Socio-economic development in the area is relatively low. Poverty and unemployment are widespread, with the rural villages being more deprived than the urban areas. The general state of water infrastructure and provision is poor, with large portions of the population lacking piped water, waterborne sanitation, and access to electricity. QwaQwa has been experiencing chronic water scarcity for more than two decades, with public protests occurring regularly. Water insecurity problems came to a head during a severe drought in 2014 to 2016 that exposed the vulnerability of communities in QwaQwa to water insecurity.

METHODOLOGY

The project comprised three components, namely climate and hydrological modelling, ecological risk assessment, and disaster risk analysis and reduction; each of which required different methods. The work was conducted over three phases.

During the first phase, sampling sites were selected on the Namahadi, Metsi-Matso, Mphukojwane, and Elands rivers and were described to serve as a focal point for the different components of the study. During the second phase, the emphasis was on the identification of ecological risks (based on field sampling) and risks to the local communities (based on social surveys through semi-structured questionnaires and interviews). The identified risks were assessed in terms of probability, frequency, intensity, and magnitude against the simulated changes in the local climate and streamflow in the study rivers. During the third and final phase, the risk reduction strategy and plan for the QwaQwa area were developed.

RESULTS AND DISCUSSION

Climate and hydrological modelling

A climate and hydrological model was developed to provide appropriate and reliable downscaled information for QwaQwa.

Climate projections indicated a warming trend for QwaQwa in the immediate and distant future, especially for the lower-lying area of the catchment where Phuthaditjhaba is situated. Mean daily minimum and maximum temperatures are expected to increase by approximately 1.5°C by the mid-2030s, and by more than 4°C by the 2070s. The number of frost days is projected to reduce by more than 40% in the higher-lying parts of the catchment, and by more than 70% in the lower-lying areas. The mean annual evaporation in QwaQwa could increase by 7% in the near future, and by more than 20% in the distant future.

QwaQwa is projected to experience higher rainfall during the rainy season (summer), but relatively less rainfall during the dry season (winter). Rainfall increases will, however, not occur uniformly over the catchment. The projections show that the greatest changes in mean annual rainfall will occur in the mountainous headwater areas, which could receive 30 to 40 mm more rain (a 4% increase from the present) in the near future, and between 140 mm and 185 mm more rain (a 20% increase from the present) in the distant future. Rainfall increases could be lower during dry years and higher during wet years. Dry spells are projected to increase in the immediate future, while wet spells could be fewer. This impact could be more pronounced in the higher-rainfall upper parts of the catchments than in the slightly drier lowland areas.

The hydrological model indicated that the bulk of the runoff in the catchment is generated on the mountain slopes, with these areas producing nearly three times more runoff, on average per year, than the lowland areas. The highest runoff generally occurs in January (summer), while August and September (winter) have the lowest runoff. The mean annual runoff in the catchment is projected to increase by approximately 7% in the highlands and 15% in the lowlands in the near future. In the distant future, runoff could increase by 36% in the highlands and by nearly 70% in the lowlands. Stormflows in the catchment generally occur in summer (rainfall season), when they contribute approximately 70% to runoff. Stormflows are projected to increase by between 13% and 21% annually, with the greatest changes expected for the highland areas. The percentage increase in baseflows, which are the major contributors to total runoff during the dry winter season, are projected to be higher in the lowlands than in the highlands.

Accumulated streamflows are projected to increase in the future, although the increases are not expected to occur uniformly over the catchment. The accumulated streamflows vary greatly between the 19 sub-catchments, depending on the size and rainfall of the sub-catchments. Peak discharges in the catchment generally occur in late summer (February) after thunderstorm activity. The projections indicate that peak discharges could increase in the immediate future (2040s), with the most downstream sub-section of the catchment (lower Elands River) being at risk of flooding in future.

Ecological risk assessment

The ecological risk assessment collected data on physical and chemical water quality, ecotoxicology, and river health. The aim of these assessments was to identify potential hazards to human and environmental health associated with extreme weather events in the study area.

The negative impacts of current anthropogenic activities on the rivers and streams were clear. The results showed that the upper reaches of the rivers, situated predominantly in the more rural and less densely populated areas, were less impacted than the middle and lower reaches, situated in the urban and more densely populated parts of the catchment.

Water quality, in general, deteriorated from upstream to downstream, with the water at the most downstream sites on the Elands River being hypertrophic, acutely toxic, and having high levels of manganese, iron, and a range of emergent pollutants such as dioxins, caffeine, pharmaceuticals, and herbicides. The presence of faecal coliform bacteria, used as an indicator of faecal pollution in surface waters, also increased from upstream to downstream in the catchment. Faecal coliform bacteria concentrations were unacceptably high in the lower Namahadi, middle Mphukojwane, and Elands rivers, where concentrations exceeded the recommended Target Water Quality Range (TWQR) set by the Department of Water and Sanitation by 1 700 times.

This study was the first to detect and confirm the presence of organic micropollutants such as sulfamethoxazole, Bisphenol-4, ibuprofen, carbamazepine, atenolol, nevirapine, efavirenz (or its metabolite 8-OH-efavirenz), and the classical indicator of human impact, caffeine, in the catchment. The data showed that the lower Mphukojwane and the Elands rivers were the most affected, with the substances being present in the low to medium range. This study was also the first to conduct a series of in vitro biotests in the catchment. These tests indicated elevated estrogenic activity in the middle Mphukojwane and Elands rivers, as well as highly increased activity of aryl-hydrocarbon in the Elands River, which can induce inflammatory responses upon exposure and may lead to chronic inflammatory diseases, including asthma, cardiovascular diseases, and increased cancer risk.

The ecotoxicological investigation found the river sediment in the Metsi-Matso, Namahadi, and Elands rivers to be contaminated with chromium above the recommended prohibition level. The Elands River was the most polluted, with chromium and zinc concentrations exceeding the recommended thresholds. A number of metals (aluminium, iron, manganese, copper, and zinc) were detected at elevated concentrations in water samples. The Mphukojwane, lower Namahadi, and Elands rivers were the most impacted, with concentrations for aluminium and iron exceeding the TWQR for both domestic use and aquatic ecosystems at these sites. Acute toxicity tests found that toxicity increased from upstream to downstream, with the sites in the middle and lower reaches being more toxic than those in the upper reaches. There is concern about the water of the Mphukojwane River, with the upper site classified as having high acute toxicity (Class IV) and the middle site as very high acute toxicity (Class V).

Riparian and instream habitats were moderately to seriously modified from their natural state. The scores for instream habitat integrity and fish assemblages were lower than the numerical limits set by the Resource Quality Objectives (as published in the *Government Gazette* in 2016) for the catchment at all the sites, except for the upper and middle Namahadi River reaches. Of particular concern is the severe flow modification in the upper Metsi-Matso due to the lack of environmental flows being released from the Metsi-Matso Dam, severe infestation of black wattle trees at the upper Namahadi, and severe erosion and solid waste disposal at the upper Mphukojwane. Also concerning is the apparent rapid

expansion of unlicensed sand mining (mostly for building and brick making) in the instream and riparian zones of streams and rivers in the catchment.

Disaster risk analysis

The disaster risk analysis comprised four sub-components: hazard analysis (hazard identification, probability, frequency, intensity), exposure analysis of key elements, vulnerability or resilience analysis using the community capitals approach, and calculation of the Multi-Hazard Risk Index (MHRI).

The hazard analysis identified four hazards that posed a high risk to communities in QwaQwa, namely water scarcity, drought, floods, and wildfires. All these highly ranked hazards were climate related, which indicates the importance of addressing climate change impacts and climate change-related risks.

The exposure analysis showed that the overall rating of exposure to the identified hazards was high. It highlighted the fact that communities in QwaQwa have very high exposure to water scarcity, and high exposure to the other three hazards (drought, floods, and wildfires). Most of the elements analysed show high exposure to all the hazards, while the population and land use showed very high exposure to water scarcity. Water sources showed very high exposure to wildfires.

The Resilience Index value calculated for QwaQwa was low, which indicated that communities in QwaQwa would struggle to recover from the impacts of extreme weather events and climate shocks. This result emphasises the importance of strengthening resilience in the area by investing in available resources to reduce disaster risk to climate change hazards and impacts.

The MHRI score was high, which indicated that the overall risk to the combination of climate-related hazards identified for QwaQwa was high.

This study took a different approach to traditional vulnerability analyses by using the Community Capitals Framework (CCF). The CCF differs from traditional approaches by identifying a community's unique strengths, and using those to empower communities. For QwaQwa, the social and natural capitals were identified as potential strengths that could be employed to improve resilience to climate-related hazards.

CONCLUSIONS

This study set out to develop an understanding of how the cumulative impacts of extreme weather events may affect the vulnerability of local communities in QwaQwa and how these impacts may be ameliorated through DRR and resilience building. To fully comprehend the extent and impact of climate-related risks in QwaQwa, a downscaled climate and hydrological model was developed for the catchment, an ecological risk assessment was conducted at nine representative sampling sites on the Namahadi, Metsi-Matso, Mphukojwane, and Elands rivers, and a disaster risk analysis was performed. The findings from the three components were then used to develop a tailor-made DRR framework that addresses the specific hazards, exposures, and vulnerabilities of local communities in the study area. The strategies that were developed specifically address the improvement of policy and political issues, local community empowerment, DRR and climate change adaptation, and resilience building to the

climate-related hazards. The study followed a systems thinking approach to recommend strategies that are interrelated and interconnected. Recommendations were also made for further research to build on, and improve on, the current project and for the replicability of similar studies in other communities in South Africa. The innovative approach of assessing community assets, instead of community vulnerabilities to extreme climate change events, stands out in this study. The addition of a downscaled climate and hydrological model and an ecological risk assessment to the conventional disaster risk assessment methodology makes this study unique.

RECOMMENDATIONS

Recommendations for future research, based on the experience gained during this project, include:

Climate and hydrological modelling

- Sustained station observation networks are essential for the long-term analysis of local and regional climate trends, as well as for the calibration of satellite-derived climate products. The installation of local weather stations at various altitudes in different parts of the catchment would greatly contribute to future studies.
- Although the water releases from the Fika-Patso Dam are monitored, it does not reflect the "natural" or "uncontrolled" flow in the catchment. Considering the important contribution of streams in the catchment to the Wilge River (via the Elands River), and ultimately to the Vaal River, the installation of streamflow gauging stations in the catchment should be considered.

Ecological risk assessment

- Information on the aquatic ecology of the streams is sparse, fragmented, and uncoordinated. In light of the projected climate and hydrological changes for the catchment, it is clear that there is a dire need for a coordinated and comprehensive ecological assessment of the streams.
- The Resource Quality Objectives (RQOs) gazetted for the Elands River catchment (Republic of South Africa, 2016) are largely lacking for river water quality. The only category for which numerical limits are available is for nutrients (phosphate, nitrite, and nitrate), most probably due to a lack of information for the other categories (e.g. salts, system variables, toxins, and pathogens). The results from this study highlight the urgent need for setting numerical limits for the other classes and for conducting regular water quality monitoring, especially in the Namahadi, Mphukojwane, and Elands rivers.
- This study found that the RQOs set for the river instream habitats (Category C or higher) are not achieved for most of the middle and lower sites. Particularly concerning is the marked increase in illegal sand mining during the course of this study. This is a serious matter that requires urgent intervention by the relevant authorities, as well as further investigation.

Disaster risk analysis and reduction

- Stage 3 and 4 of the United Nations Office for Disaster Risk Reduction's (2016) risk assessment process, including monitoring and evaluation, ought to be commissioned in another research project by the Water Research Commission.
- Further, and more in-depth, research should be conducted on the 17 hazards identified for QwaQwa, but not only on those related to extreme climate change events, since risks are integrated and systematic, with cascading and compounding effects.
- The approach developed and applied in this study should be tested on other climate hotspots; for example, Nelson Mandela Bay District Municipality in the Eastern Cape or eThekwini in KwaZulu-Natal.

ACKNOWLEDGEMENTS

The project team, under the auspices of the Centre for Environmental Management (CEM), would like to acknowledge the Water Research Commission for supporting and funding this project. We would like to specifically acknowledge Dr Brilliant Petja for his input and support throughout the project.

We also thank the following persons:

The Review Group, consisting of the following members, are thanked for their guidance, support, and continued interest in the project:

Prof. Kingsley Ayisi	University of Limpopo
Dr Joel Botai	South African Weather Service
Dr Gabriel Lekalakala	Department of Water and Sanitation
Mr Robert Maisha	South African Weather Service
Dr Mangi Mgcuba	Department of Water and Sanitation
Dr Chris Moseki	Department of Water and Sanitation
Ms Lerato Mpheshea	Council of Scientific and Industrial Research
Dr Tendai Sawunyama	Inkomati-Usuthu Catchment Management Agency

Ms Gerda Kruger (WRC, project co-ordinator) for her kind assistance, timely reminders, and support throughout the project.

Ms Donné Kolesky for her assistance with logistical support and the financial management of the project.

Ms Ngitheni Nyoka, Ms Fumi Muyambo, and Mr Ndu Kubheka for their hard work and assistance throughout the project.

The University of the Free State for providing basic facilities and logistics for the study, especially the Department of Chemistry (QwaQwa campus) for allowing us the use of their laboratory and equipment for doing the solid phase extractions.

The Cape Peninsula University of Technology for the provision of laboratory infrastructure for the ecotoxicological studies, and Komlan Apetogbor for his support in the laboratory.

The Technical University of Dresden for providing financial support by the *Flexible Funding Programme Internationalisation* to realise additional sampling, and the Institutes of Hydrobiology and Institute of Water Chemistry for assistance. We also thank Lorena Hannig, Juliane Isler, and Oliver Faber for technical support in the laboratory.

CONTENTS

DISCL	AIMER			ii
EXEC	JTIVE SI	UMMARY	/	iii
ACKN	OWLED	GEMENT	S	х
LIST C	of Figur	RES		xvi
LIST C	OF TABL	ES		xviii
LIST C	F ABBR	EVIATIO	NS	xix
LIST C	OF UNITS	S		xxii
CHAP	TER 1: B	ACKGR	OUND	1
1.1	PROJE	CT BACK	GROUND	1
1.2	PROJE	CT OVEF	RVIEW	2
1.3	PROJE	CT TEAM	1	3
1.4	PROJE	CT LIMIT	ATIONS AND CONSTRAINTS	4
1.5	LAYOU	T OF TH	S REPORT	4
CHAP	TER 2: R	RISKS OF	EXTREME CLIMATE EVENTS ON SOCIO-ECOLOGICAL SYSTEMS	5
2.1	INTRO	DUCTION	I	5
2.2	OVERV	IEW OF	DISASTER RISK, EXPOSURE, VULNERABILITY, AND RESILIENCE	5
2.3	EXTRE	ME CLIM	ATE EVENTS AND DISASTERS	6
	2.3.1	Risks of	extreme climate events and disasters	6
	2.3.2	The like	lihood of an increase in the frequency of extreme climate events in	
		Souther	n Africa and the QwaQwa study area	7
	2.3.3	Potentia	I impacts of extreme events on the ecological integrity of rivers	8
		2.3.3.1	Predicting the impacts of extreme events on river ecosystems and river	10
	221	Potontic	use	10
	2.3.4	2244	Extreme best and bestwayee	10
		2.3.4.1	Extreme and floods	10
		2.3.4.2	Droughte (and very law flaws)	 4 4
		2.3.4.3	Mildfiree	11
	0 0 E	2.3.4.4 Managa	windings	IZ
	2.3.5	wanage	ment considerations	12
CHAP.	TER 3: C	ASE STU	JDY: QWAQWA	14
3.1	BACKG	ROUND.		14
3.2	DIFFER	ENT SC	ALES OF SAMPLING AND SITE SELECTION	16
	3.2.1	Scale of	sampling	16
	3.2.2	Site sele	ection	16
3.3	DESCR		DF STUDY AREA	17
	3.3.1	Physica	l environment	17

		3.3.1.1	Climate	17
		3.3.1.2	Geology, soil, and erodibility	18
		3.3.1.3	Vegetation and land use	20
	3.3.2	Socio-e	conomic environment	22
		3.3.2.1	Brief history of QwaQwa	22
		3.3.2.2	Present situation	22
		3.3.2.3	Water crisis in QwaQwa	24
		3.3.2.4	Water infrastructure and services provision	24
		3.3.2.5	Vulnerability and the potential impact of climate change on services	26
CHAP	TER 4: D	EVELOP	MENT OF A DOWNSCALED CLIMATE AND HYDROLOGICAL MODEL	
FOR	QWAQWA	A		27
11				27
4.1 10				21
4.2				27
	121	Historia	al databases used	21
	4.2.1	Climata	al udiabases used	Z1
	4.2.2		Change databases used	20
	4.2.3	1 1 2 2 1		20
		4.2.3.1	Conorel structure of the ACRU model	29
		4.2.3.2	Constant Structure of the ACRU model	30 22
		4.2.3.3	Constant of storminows with the ACRU model	⊃∠
		4.2.3.4	Constantion of pask discharge	34
		4.2.3.3	Concretion of peak discharge	34
		4.2.3.0	Verification of sediment yields with the ACRU model	35
		4.2.3.7	Verification studies on the ACRU model's output	31
		4.2.3.0	Pageling assessments and in situ assessments	31 27
	4.2.4	4.2.3.9	baseline assessments and in situ assessments undertaken in the OuroOur	37
	4.2.4	study ar		ו 37
			The quipery established to primary spatial unit of this study	37
		4.2.4.1	Sub establights within the guinarian used for modelling in this study	
		4.2.4.2	Sub-catchine its within the quinaries used for modeling in this study	
		4.2.4.3	response units (HPLIs) for more detailed modelling in this study	ai 11
	125	Location	a of the compling sites	4 1
10		LUCALIO	Tor the sampling sites	41
4.5	A 2 1	Tompor	aturo	42
	4.3.1	1211		42
		4.3.1.1	Projections for the future	4Z
	122	4.3.1.Z		44
	4.3.2			44
		4.3.2.1	Projections for the future	44
	100	4.J.Z.Z		40
	4.3.3	Raman 1221	Historical conditions	40
		4.3.3.1	Disionical conditions	40
	121	4.3.3.Z		47
	4.3.4		Historical conditions	. 50
		4.3.4.1	Distolical collutions	50
	1 D E	4.3.4.Z		
	4.3.5	SIORMIC	JWS	54
		4.3.5.1	Distolical collutions	54
		4.3.5.2	Projections for the future	54

	4.3.6	Baseflow	/S	54
		4.3.6.1	Historical conditions	54
		4.3.6.2	Projections for the future	54
	4.3.7	Accumul	ated streamflow	57
	4.3.8	Peak dis	charges	59
	4.3.9	Sedimen	t yields	61
		4.3.9.1	Historical conditions	61
		4.3.9.2	Projections for the future	62
4.4	KEY FIN	IDINGS A	ND TRENDS	63
	4.4.1	Rainfall.		63
	4.4.2	Potential	evaporation	64
	4.4.3	Runoff		64
	4.4.4	Baseflow	/5	64
	4.4.5	Stormflov	WS	65
	4.4.6	Peak dis	charge	65
	4.4.7	Sedimen	t yield	65
CHAP	TER 5: E	COLOGIO	CAL RISK ASSESSMENT OF THE HEADWATER STREAMS IN	
QWAC	QWA			66
Г 4				~~~
5.1				60
5.2	DATAC	OLLECII	UN AND ANALYSES	00
	5.2.1	Compon	ents included in the ecological risk assessment	00
	5.2.2	Field sar	npiing	60
	5.2.3	Descripti	on of sampling sites	67
		5.2.3.1		71
		5.2.3.2	Metsi-Matso	73
		5.2.3.3	Mpnukojwane River	74
		5.2.3.4	Elands River	75
	5.2.4	Descripti	on of sampling and analytical methods applied in the study	76
		5.2.4.1	Physical and chemical water quality	76
		5.2.4.2	Ecotoxicological tests	78
		5.2.4.3	River health	82
	5.2.5	Identifica	ition of potential risks to human health and the delivery of ecosystem	~ 4
		services		84
		5.2.5.1	Physical and chemical water quality	84
		5.2.5.2		85
F 0		5.2.5.3		87
5.3	RESUL			87
	5.3.1	Physical	and chemical water quality	88
		5.3.1.1	System variables	88
		5.3.1.2	Metals	88
	500	5.3.1.3	OMPs	88
	5.3.2	Ecotoxic	ological tests	89
		5.3.2.1	River sediment	89
		5.3.2.2		91
		5.3.2.3	IN VITO TESTS	91
	5.3.3	River he	alth	92
		5.3.3.1	Instream and riparian HI	92
		5.3.3.2	Fish community health	92
5.4	KEY FIN	IDINGS A	ND PUTENTIAL RISKS IDENTIFIED	93

	TER 6: VI IUNITIES	ULNERABILITY ASSESSMENT / DISASTER RISK ASSESSMENT OF QWAQV	VA 97
6.1	INTROD	DUCTION	97
6.2	RISK AS	SSESSMENT METHODOLOGY	98
	6.2.1	Understanding risk	98
	6.2.2	Risk equation	98
	6.2.3	The Community Capitals Framework (CCF) and other relevant frameworks	99
	6.2.4	Community Based Resilience Analysis (CoBRA)	100
6.3	DATA C	OLLECTION AND ANALYSES	101
	6.3.1	Data collection	102
		6.3.1.1 Key informant (KI) interviews	102
		6.3.1.2 Household surveys	102
		6.3.1.3 Field observations	102
		6.3.1.4 Stakeholder validation workshop	102
	6.3.2	Data analyses and results	103
		6.3.2.1 Hazard analysis	103
		6.3.2.2 Exposure analysis	105
		6.3.2.3 Resilience analysis	106
		6.3.2.4 Calculation of the Multi-Hazard Risk Index (MHRI)	108
6.4	KEY FIN	NDINGS AND DISCUSSION	109
	6.4.1	Hazard analysis	109
	6.4.2	Exposure analysis	109
	6.4.3	Resilience analysis	109
	TER 7: D	ISASTER RISK REDUCTION (DRR) STRATEGY AND PLAN FOR QWAQWA	111
7.1			111 s
1.2	OWAOV	VA	111
73	REDUC	ING THE RISKS' RISK REDUCTION STRATEGIES	114
7.0	731	Policy recommendations	115
	732	Recommendations for the empowerment of the local communities	110
	733	Recommendations to reduce disaster risks and adapt to climate change	124
	734	Recommendations to reduce disaster risk and huild resilience against the ecolo	naical
	7.0.4	risks identified in the OwaOwa study area (whole catchment)	126
		Tisks identified in the QwaQwa study area (whole catchment)	120
	TER 8: C ARCH AN	ONCLUSIONS, LESSONS LEARNED, AND RECOMMENDATIONS FOR FUTU	RE 134
8.1	INTROD	DUCTION	134
8.2	CONCL	USIONS	134
	8.2.1	Climate and hydrological modelling	134
	8.2.2	Ecological risk assessment	135
	8.2.3	Disaster risk analysis	136
	8.2.4	DRR strategies	137
8.3	LESSON	NS LEARNED	137
8.4	RECOM	IMENDATIONS FOR FURTHER RESEARCH	138
	8.4.1	Climate and hydrological modelling	138
	8.4.2	Ecological risk assessment	138
	8.4.3	Disaster risk analysis and reduction	139

REFERENCES	.140
APPENDIX A: PHOTOGRAPHS OF THE NINE SAMPLING SITES ON THE NAMAHADI, METSI- MATSO, MPHUKOJWANE, AND ELANDS RIVERS	.166
APPENDIX B: LIST OF POTENTIAL POLLUTANTS, POTENTIAL SOURCES, AND ASSOCIATE RISKS	D 171
APPENDIX C: RESULTS FROM THE PHYSICO-CHEMICAL, ECOTOXICOLOGICAL AND RIVER HEALTH ASSESSMENTS OF THE NINE SAMPLING SITES ON THE NAMAHADI, METSI-MATS MPHUKOJWANE, AND ELANDS RIVERS	≀ O, 174
APPENDIX D: PHOTOGRAPHS OF PROMINENT ANTHROPOGENIC IMPACTS ON THE INSTREAM AND RIPARIAN ZONES AT THE SAMPLING SITES AS OBSERVED DURING THE APRIL 2021 FIELD VISIT	176

LIST OF FIGURES

Figure 1.1.	A graphic representation of the project outline
Figure 3.1.	Map showing the position of the study area within the Wilge River catchment15
Figure 3.2.	Map showing the boundaries of the Maluti-a-Phofung Local Municipality15
Figure 3.3.	Topography of the study area18
Figure 3.4.	Mean annual temperature (top) and mean annual precipitation (MAP) (bottom) of the study area
Figure 3.5.	Vegetation (top) and land cover types (bottom) in the study area21
Figure 3.6.	Residential areas in the study area23
Figure 4.1.	General structure and multi-purposeness of the ACRU model
Figure 4.2.	Schematic of major processes represented in the ACRU model
Figure 4.3.	Interrelationships used in ACRU between soil water content and the ratio of $E : E_{tm}$ 32
Figure 4.4.	Schematic of runoff-generating mechanisms in the ACRU model
Figure 4.5.	Water quality sampling points (green rings), as well as additional quinary-related points identified as sub-catchment outlets for hydrological modelling purposes (red squares; left map), and (right map) the quinary catchments with the sub-catchment numbering system
Figure 4.6.	The main rivers in the study area in relation to residential areas of Phuthaditjhaba (left), with the location of the seven water quality sampling points located on the Namahadi and Metsi-Matso rivers flowing north-westwards from the south and east, and the Mphukojwane flowing north-eastwards, merging to form the Elands River in the north
Figure 4.7.	The flow routing among the 19 sub-catchments within the study area, with the locations of the water quality sampling points highlighted in red
Figure 4.8.	Monthly means for historical maximum temperatures for January and historical minimum temperatures for July for QwaQwa43
Figure 4.9.	Mean annual A-pan equivalent reference potential evaporation (mm) across QwaQwa under historical climatic conditions
Figure 4.10.	Historical mean annual rainfall in QwaQwa46
Figure 4.11.	Historical median monthly rainfall in the upper, middle, and lower quinaries47
Figure 4.12.	MAP (mm) across the 19 QwaQwa sub-catchments for the present period, derived from multiple Coupled Model Intercomparison Project Phase 5 (CMIP5) Global Climate Models (GCMs), and the projected increase from the present to the immediate future
Figure 4.13.	Annual precipitation (mm) in the wettest year in 10 across the 19 QwaQwa sub- catchments for the present period, derived from multiple CMIP5 GCMs, and the projected increase from the present to the immediate future
Figure 4.14.	Comparison of mean annual rainfall (mm) across the 19 sub-catchments for the present and immediate future periods as derived from multiple CMIP5 GCMs50
Figure 4.15.	Median monthly runoff from baseline land cover under historical climatic conditions for the upper, middle, and lower quinaries in QwaQwa51
Figure 4.16.	Runoff ratios from baseline land cover conditions for the near and distant futures to the present for the upper, middle, and lower quinaries in the QwaQwa study area

Figure 4.17.	Average annual local runoff (streamflow on map, mm) across the 19 QwaQwa sub- catchments for the present period, derived from multiple CMIP5 GCMs, and the projected increase from the present into the immediate future, all under baseline land cover
Figure 4.18.	Projected changes in mean annual baseflows from the present to the immediate future in mm and as a percentage
Figure 4.19.	Average annual local stormflow (mm), under baseline land cover conditions, across the 19 sub-catchments for the present period, derived from multiple CMIP5 GCMs, and the projected increase from the present into the immediate future
Figure 4.20.	Projected changes in mean annual accumulated streamflows from the present to the immediate future in m ³ /a and as a percentage, all under baseline land cover conditions
Figure 4.21.	Average annual accumulated streamflows, under baseline land cover conditions, across the 19 QwaQwa sub-catchments for the present period, derived from multiple CMIP5 GCMs, and projected increases from the present to the immediate future
Figure 4.22.	Monthly means of the Peak Discharge Index (PDI) under baseline land cover conditions in the upper, middle, and lower quinaries of the QwaQwa study area
Figure 4.23.	Mean annual accumulated peak discharge across the 19 QwaQwa sub-catchments for the present period as derived from multiple CMIP5 GCMs, and projected increases from the present to the immediate future, all under baseline land cover conditions
Figure 4.24.	Annual sediment yields in t/km ² .a under baseline land cover conditions in the driest year in 10, a median sediment production year, and a 1:10 year high for the upper, middle, and lower quinaries in the study area
Figure 4.25.	Monthly means of sediment yields under historical climatic and baseline land cover conditions from the upper, middle, and lower quinaries in the study area
Figure 4.26.	Projected changes in mean annual sediment yields from the present to the immediate future in t/ha.a and as a percentage, all under baseline land cover
Figure 4.27.	Mean annual accumulated baseline sediment yields in tonnes and in t/ha across the 19 sub-catchments for the present period, derived from multiple GCMs, and projected increases from the present to the immediate future in tonnes and in t/ha
Figure 5.1.	Sampling sites on the Metsi-Matso, Namahadi, Mphukojwane, and Elands rivers for ecological sampling
Figure 5.2.	Level 2 ecoregions and geomorphological zones at the study sites
Figure 5.3.	National Freshwater Ecosystem Priority Areas river and wetland conditions in the study area
Figure 6.1.	Disaster risk assessment methodology
Figure 6.2.	Community capitals used to analyse resilience
Figure 6.3.	The CoBRA framework
Figure 6.4.	Steps in undertaking a CoBRA101
Figure 6.5.	Hazard ranking
Figure 6.6.	Selected climate-related hazards scoring105
Figure 6.7.	Average capital scores for the measured community capitals

LIST OF TABLES

Table 1.1.	Project objectives for the study	2
Table 1.2.	The study team involved in the study and their specific fields of expertise	3
Table 2.1.	Potential negative impacts of extreme climate events on physical riverine drivers	9
Table 2.2.	Potential negative impacts of extreme climate events on riverine biota	9
Table 4.1.	Names, coordinates, and altitudes of the seven sampling points as focus points in the climate and hydrological modelling	. 39
Table 4.2.	Mean altitudes and areas of the 19 sub-catchments identified for QwaQwa	.40
Table 4.3.	Percentages of specific quinaries per sub-catchment	.41
Table 4.4.	Location of the sampling sites in relation to the quinaries and sub-catchments	.42
Table 5.1.	Components included in the ecological risk assessment of the streams in the study area	.67
Table 5.2.	Summary of the sampling sites on the Namahadi, Metsi-Matso, Mphukojwane, and Elands rivers	.69
Table 5.3.	River health categories	. 82
Table 5.4.	Level I and Level II acceptable concentration ranges and thresholds used for the screening of dredged sediment proposed for marine disposal, in mg/kg dry weight, in South Africa	. 86
Table 5.5.	Hazard classification system for natural waters	.86
Table 5.6.	Calculation of the class weight scores	. 87
Table 5.7.	Summary of the ecological risks identified in the QwaQwa study area	.96
Table 6.1.	Identified hazards	103
Table 6.2.	Hazard Index and rating	104
Table 6.3.	Overall exposure scores and rating	106
Table 6.4.	Socio-economic characteristics of household survey participants (N = 246)	107
Table 6.5.	Weighting of capitals and calculation of Resilience Index	108
Table 6.6.	Resilience Index ranking	108
Table 6.7.	Maluti-A-Phofung Local Municipality risk register	110
Table 7.1.	Policy recommendations for reducing disaster risk in QwaQwa	115
Table 7.2.	Recommendations for local community empowerment	119
Table 7.3.	Recommendations for reducing disaster risk and developing climate change adaptation in QwaQwa	124
Table 7.4.	Strategies for reducing disaster risks related to water scarcity in QwaQwa	126

LIST OF ABBREVIATIONS

β-NF	β-Naphthoflavone
4-OHT	4-Hydroxytamoxifen
ACRU	Agricultural Catchments Research Unit
AhR	Aryl-hydrocarbon receptor
AIDS	Acquired immunodeficiency syndrome
Al	Aluminium
As	Arsenic
BOD	Biochemical oxygen demand
BPA	Bisphenol A
BP-4	Bisphenol-4
Ca	Calcium
CAS	Chemical Abstracts Service
CCAM	Conformal Cubic Atmospheric Model
CCF	Community Capitals Framework
Cd	Cadmium
CEM	Centre for Environmental Management
CMIP5	Coupled Model Intercomparison Project Phase 5
CoBRA	Community Based Resilience Analysis
COD	Chemical oxygen demand
CoGTA	Department of Cooperative Governance and Traditional Affairs
Cr	Chromium
CR(VI)	Hexavalent chromium
CSIR	Council for Scientific and Industrial Research
Cu	Copper
CV	Coefficient of variation
DEET	N,N-Diethyl-meta-toluamide
DIN EN	Deutsches Institut für Normung Europaische Norm
DMSO	Dimethyl sulfoxide
DO	Dissolved oxygen
DOC	Dissolved organic carbon
DRR	Disaster risk reduction
DWAF	Department of Water Affairs and Forestry
DWS	Department of Water and Sanitation
E2	17ß-Estradiol
EC	Electrical conductivity
EC50	50% effective concentration
ELL	Elands Lower [site code]
ELU	Elands Upper [site code]
EMS	Ethyl methanesulfonate

ESI	Electrospray ionisation
Fe	Iron
Flu	Flutamide
FRAI	Fish Response Assessment Index
GCM	Global Climate Model
hAR	Human androgen receptor
hER	Human estrogen receptor
Hg	Mercury
HGV	Health-oriented Guidance Value
н	Habitat integrity
HIV	Human immunodeficiency virus
HRU	Hydrological response unit
ICP-OES	Inductively Coupled Plasma Optical Emission Spectrometer
IPCC	Intergovernmental Panel on Climate Change
IS	Internal standard
ISO	International Organization for Standardization
ISO/IEC	International Organization for Standardization / International Electrotechnical
	Commission
KI	Key informant
LC50	Lethal concentration causing the death of 50% of a population
LOQ	Limit of quantification
MAP	Mean annual precipitation
MAR	Mean annual runoff
Mg	Magnesium
MHRI	Multi-Hazard Risk Index
MM	Metsi-Matso Middle [site code]
Mn	Manganese
MNvit	Micronucleus in vitro
MPM	Mphukojwane Middle [site code]
MPN	Most probable number
MPU	Mphukojwane Upper [site code]
MU	Metsi-Matso Upper [site code]
MUSLE	Modified Universal Soil Loss Equation
NbS	Nature-based Solutions
NDMC	National Disaster Management Centre
NGO	Non-governmental organisation
Ni	Nickel
NL	Namahadi Lower [site code]
NM	Namahadi Middle [site code]
NU	Namahadi Upper [site code]

OECD	Organisation for Economic Co-operation and Development
OMP	Organic micropollutant
Pb	Lead
PDI	Peak Discharge Index
PE	Percentage effect
QnCDB	Quinary Catchments Database
RCP	Representative concentration pathway
RPMI	Roswell Park Memorial Institute
RQO	Resource Quality Objective
RSA	Republic of South Africa
SAC ₂₅₄	Spectral absorption coefficient at 254 nm
SANS	South African National Standard
SFDRR	Sendai Framework for Disaster Risk Reduction
StatsSA	Statistics South Africa
т	Testosterone
TDS	Total dissolved solids
Tmin	Minimum temperature
TSS	Total suspended solids
TU	Toxicity unit
TWQR	Target Water Quality Range
UFS	University of the Free State
UN	United Nations
UNDP	United Nations Development Programme
UNDRR	United Nations Office for Disaster Risk Reduction
UNEP	United Nations Environment Programme
UNISDR	United Nations International Strategy for Disaster Reduction
USLE	Universal Soil Loss Equation
UV	Ultraviolet
WFI	Water for Injection
WRC	Water Research Commission
WTP	Water treatment plant
WTW	Water treatment works
WWTP	Wastewater treatment plant
YAAS	Yeast-based Anti-androgenic Screen
YAES	Yeast-based Anti-estrogenic Screen
YAS	Yeast-based Androgenic Screen
YDS	Yeast-based Dioxin-like Screen
YES	Yeast-based Estrogenic Screen
Zn	Zinc

LIST OF UNITS

μg	Microgram(s)
µg/L	Microgram(s) per litre
μL	Microlitre(s)
μm	Micrometre(s)
сс	Cubic centimetre(s)
cm	Centimetre(s)
g	Gram(s)
kg	Kilogram(s)
km	Kilometre(s)
km ²	Square kilometre(s)
m	Metre(s)
m ³	Cubic metre(s)
m ³ .s ⁻¹	Cubic metre per second
m³/a	Cubic metres per annum
m³/s	Cubic metres per second
m ³ /s.km ²	Cubic metres per second per square kilometre
m.a.s.l.	Metres above sea level
mbar	Millibar
mg	Milligram(s)
mg/L	Milligram per litre
mg/kg	Milligram per kilogram
ml	Millilitre(s)
ML/d	Million litres per day
ml/min	Millilitre(s) per minute
mm	Millimetre(s)
mm.h ⁻¹	Millimetres per hour
mm/a	Millimetres per annum
mS/cm	MilliSiemens per centimetre
ng/L	Nanograms per litre
nm	Nanometre(s)
psi	Pounds per square inch
t/ha	Tonnes per hectare
t/ha.a	Tonnes per hectare per annum
t/km².a	Tonnes per square kilometre per annum
V	Voltage
v/v	Volume per volume

CHAPTER 1: BACKGROUND

1.1 PROJECT BACKGROUND

Recent climate projections for the Southern African region strongly indicate that the occurrence of rainfall extremes and prolonged drought periods may increase in the near future. Such extreme events pose a severe threat to water resources, food security, public health, infrastructure, biological diversity, and ecological systems in the region, as well as changes to the temporal dynamics of flow regimes in rivers. These projected changes could hold important consequences for both humans (e.g. influencing freshwater water supply) and to riverine biota (e.g. influencing habitat availability and quality).

While current research on extremes generally focuses on relatively large catchments with a low spatial resolution, information is needed at detailed spatial resolution on smaller catchments as to whether the climate projections for Southern Africa actually display increasing probabilities of sequential extreme events of, for example, torrential rainfall following directly after an extended drought period. Sequential extremes could have severe consequences by triggering cascades of secondary effects such as high sediment yields with impacts on, for example, infrastructural damage to water purification plants.

To improve the resilience against such impacts and to develop efficient coping strategies, new approaches that integrate the assessment of exposure and vulnerability on the one hand, and offer efficient coping strategies on the other, are needed. However, such adaptation strategies must be based on appropriate and reliable downscaled information and must include communication technologies with early warning and decision support systems for the population and decision makers concerned. Such adaptation strategies are all the more important given that weak governance could further increase the exposure to these risks, and that both the water management sector and local municipalities in South Africa are poorly prepared for expected medium- and long-term changes associated with climate change in that these sectors are failing to integrate information on climate change into their management strategies and systems in a systematic manner (Mukheibir, 2008; Ziervogel *et al.*, 2010). This failure could hold dire consequences for already vulnerable communities and sectors and, to a large extent, counter socio-economic development efforts by the government and other non-governmental agencies.

Climate change is expected to have a marked impact on the flow regimes of rivers, which could either be reduced or enhanced markedly over the next 30 or so years; changing perennial rivers to nonperennial, and semi-permanent rivers to ephemeral or episodic rivers, or vice versa (Döll and Schmied, 2012; Van Vliet *et al.*, 2013; Gudmundsson *et al.*, 2021; Tanner *et al.*, 2022). It is therefore vital to understand how changes to the timing, seasonality, incidences, and magnitudes of high and low flows, or periods of flow intermittence, could increase the vulnerability of aquatic biota and river ecosystems to climate-induced changes. Of special interest are the potential effects on the ecosystem services (e.g. freshwater for household use, water for irrigation, dilution of pollution, etc.) delivered by these rivers. Much of South Africa lacks large freshwater bodies and many areas rely on groundwater and/or direct abstraction from rivers for freshwater supply. A change, especially a reduction, in annual and seasonal runoff could therefore not only severely affect freshwater biota, but also the communities that rely on these water sources. This study investigated the risk factors associated with the projected impacts of the expected future changes in the occurrence of extreme hydro-climatic events (e.g. floods, highintensity rainfall events, and droughts) on river ecosystems in the QwaQwa area of the Upper Vaal River catchment in the Free State province of South Africa in order to develop a relevant and appropriate risk reduction strategy that would be useful to the local authorities, the water as well as the agricultural and human health sectors, and also to local communities in this area.

1.2 **PROJECT OVERVIEW**

The overarching aim of this study is to understand how river ecosystem services (including water provisioning) may be influenced by the possible cumulative impacts of successive extreme weather events in future, how these impacts may affect the vulnerability and resilience of local communities in the QwaQwa area, and how these effects may be ameliorated through risk reduction planning (see Table 1.1).

No.	Description of objectives
1	To determine how the ecological services of river ecosystems in the QwaQwa area could be
	impacted by an increase in the occurrence of extreme weather events.
2	To develop an understanding of the impact that secondary effects may have on service
	delivery and infrastructure development in QwaQwa.
3	To identify the most important hazards and risks for the QwaQwa community in terms of water
	delivery, human health, and environmental safety.
4	To prepare a tailor-made risk assessment for the QwaQwa area based on downscaled climate
	and hydrology models specifically developed for the area.
5	To develop a tailor-made disaster risk reduction (DRR) strategy and plan for the QwaQwa
	area.

Table 1.1. Project objectives for the study

In order to achieve the aims of the study, the project was divided into three phases: (1) risk analysis, (2) risk assessment, and (3) developing the risk reduction strategy and plan (see Figure 1.1). In the first phase, sampling sites were selected on the Namahadi, Metsi-Matso, Mphukojwane, and Elands rivers and were described to serve as a focal point for the different components of the study. During the second phase, the emphasis was on the identification of ecological risks, based on field surveys, and risks to the local communities, based on social surveys (semi-structured questionnaires and interviews). The identified risks were assessed in terms of probability, frequency, intensity, and magnitude against simulated changes in runoff and accumulated streamflow resulting from projected changes in the climate. During the third and final phase, the risk reduction strategy and plan for the QwaQwa area were developed.



Figure 1.1. A graphic representation of the project outline

1.3 PROJECT TEAM

The multi-disciplinary project team consisted of eight members, who were assisted by four postgraduate students from the University of the Free State (UFS) (see Table 1.2).

PROJECT TEAM			
Name	Role	Expertise	Institution
Dr Marinda Avenant	Project leader	Environmental	Centre for Environmental
		management and aquatic	Management (CEM), UFS
		ecology	
Prof. Johanes Belle	Principal	Environmental and	Disaster Management,
	researcher	disaster management	Training and Education
			Centre for Africa, UFS
Prof. Beatrice	Principal	Water quality and	Faculty of Applied Sciences,
Opeolu	researcher	environmental toxicology	Cape Peninsula University of
			Technology
Dr Patricks Voua	Principal	Environmental toxicology	Zoology and Entomology,
Otomo	researcher		UFS
Dr Dirk Jungmann	Principal	Water quality and	Institute for Hydrobiology,
	researcher	ecotoxicology	Technical University of
			Dresden
Dr Hilmar Börnick	Principal	Chemistry	Institute of Water Chemistry,
	researcher		Technical University of
			Dresden

Table 1.2. The study team involved in the study and their specific fields of expertise

PROJECT TEAM			
Name	Role	Expertise	Institution
Emeritus Prof.	Collaborator	Climate and hydrological	University of KwaZulu-Natal
Roland Schulze		modelling	
Mr Nick Davis	Collaborator	Climate and hydrological	Isikhungsethu Environmental
		modelling	Services
Ms Ngitheni Nyoka	Student	Ecotoxicology	UFS
	researcher (PhD)		
Ms Fumiso	Student	Disaster management	UFS
Muyambo	researcher (PhD)		
Mr Nduduzo	Student	Ecotoxicology	UFS
Kubheka	researcher (MSc)		
Mr Paul Masoabi	Student	Africa studies	UFS
	researcher (BA		
	Hons)		

1.4 PROJECT LIMITATIONS AND CONSTRAINTS

This project commenced on 1 April 2020 while the country was experiencing a national lockdown (level 5) due to a National State of Disaster being declared by President Cyril Ramaphosa on 23 March 2020 in an effort to curb the spread of the COVID-19 virus. The travel and social restrictions, associated with the National State of Disaster, had a significant impact on the sequencing and alignment of the research activities, as well as the overall progress of the study. The social component of the study, in particular, experienced long delays.

1.5 LAYOUT OF THIS REPORT

The remainder of the report is structured as follows:

- Chapter 2 provides a brief literature background on the impacts of extreme climate events on river health.
- Chapter 3 introduces and describes the study area.
- Chapter 4 discusses the development of the downscaled climate and hydrological and provides a brief overview of the results.
- Chapter 5 reports on the ecological risk assessment, including the key findings.
- Chapter 6 provides an overview of the disaster risk analysis process.
- Chapter 7 presents the disaster risk strategies developed for QwaQwa.
- Chapter 8 concludes the report with lessons learned and recommendations for future research.

CHAPTER 2: RISKS OF EXTREME CLIMATE EVENTS ON SOCIO-ECOLOGICAL SYSTEMS

2.1 INTRODUCTION

The Millennium Ecosystem Assessment, initiated by the United Nations (UN) and published in 2005, was a landmark study that assessed the state of ecosystems globally at the start of the third millennium. The study highlighted the interdependence between the state of ecological systems and human wellbeing by showing the consequences of anthropogenic impacts on ecosystems and their ability to deliver ecosystem services. It furthermore identified five drivers that contributed the most to ecosystem change and degradation in the 20th century, namely habitat change, invasive species, overexploitation, climate change, and pollution (Millennium Ecosystem Assessment, 2005). Of these five drivers, two were believed would become more and more significant in future, namely climate change and pollution (especially enriching from nitrogen and phosphorus). The concept of socio-ecological interdependence has since grown in prominence, and in the new series of Intergovernmental Panel on Climate Change (IPCC) reports the interdependence between climate, ecosystems and biodiversity, and human societies is recognised in the assessment of climate impacts and risks, as well as in the development of climate responses (IPCC, 2022).

2.2 OVERVIEW OF DISASTER RISK, EXPOSURE, VULNERABILITY, AND RESILIENCE

Climate change is perceived by many as the greatest threat the world has ever faced (Jenkins, 2021; UN Office of the High Commissioner for Human Rights, 2022). The impacts from a rapidly changing climate are arguably threatening every aspect of human existence and have been described by the Secretary-General of the UN, Mr Antonio Guterres, as a "crisis multiplier" that could threaten international peace and stability (UN Security Council, 2021). Although climate change is a global phenomenon, not everyone will be equally affected by its impacts. It has been shown that lower-income economies and populations may face the biggest climate risks (World Economic Forum, 2020). Southern Africa, in particular, is considered to be very vulnerable to the impacts of climate change, not only due to its geographical location and climate variability (Scholes and Engelbrecht, 2021), but also because the region has been experiencing climate changes that are more rapid and severe than the global mean (Maúre et al., 2018; Fitchett, 2021). The region's vulnerability is further aggravated by socio-economic factors such as widespread poverty, declining food production and food insecurity, habitat degradation and desertification, poor and ageing infrastructure, increasing pressure on natural resources, weak governance and institutions, the prevalence of HIV/AIDS and other diseases such as malaria, armed conflicts, etc., and recently COVID-19 (United Nations Environment Programme [UNEP], 2013; Niang et al., 2014; World Economic Forum, 2020).

The concept of "risk" has become increasingly prominent in the consideration of climate change impacts over the last decade. In the most recent series of reports published by the IPCC, "risk" is used as a central framework, both for understanding how climate change may impact socio-ecological systems across regions, sectors, communities, and generations, and for developing appropriate responses to these impacts (IPCC, 2022). According to the Sixth Assessment Report of the IPCC (2022), risk, which can be defined as "the potential for adverse consequences for human or ecological systems", can arise from the dynamic interaction between climate-related hazards, the exposure to these hazards, and the vulnerability of human or ecological systems to these hazards when exposed to them. Climate-related hazards refer to "the potential occurrence of a natural or human-induced physical event or trend that may cause loss of life, health impacts, as well as damage and loss to property, infrastructure, livelihoods, service provision, ecosystems and environmental resources" (IPCC, 2022). The exposure and vulnerability to such climate-related hazards are not homogenous, and differ within communities and across societies, regions, countries, and even time (IPCC, 2022). Exposure and vulnerability are seen as the key determinants of disaster risk, and climate adaptation should focus on reducing exposure and vulnerability to hazards and building community and system resilience.

The Sendai Framework for Disaster Risk Reduction (SFDRR) 2015-2030 was the outcome of the Third UN World Conference on Disaster Risk Reduction held from 14 to 18 March 2015 in Sendai in Japan. Its main outcome was "[t]he substantial reduction in disaster risk and losses in lives, livelihoods and health and in the economic, physical, social, cultural and environmental assets of persons, businesses, communities and countries" (United Nations International Strategy for Disaster Reduction [UNISDR], 2015) and its goal was to "prevent new and reduce existing disaster risk through the implementation of integrated and inclusive economic, structural, legal, social, health, cultural, educational, environmental, technological, political and institutional measures that prevent and reduce hazard exposure and vulnerability to disaster, increase preparedness for response and recovery, and thus strengthen resilience" (UNISDR, 2015). With seven targets and four priority areas, the SFDRR serves as an international yardstick to measure the extent to which countries in the global community have reduced disaster risks, including those emanating from extreme climate events. The goal, Target 'D', and Priority 3 of the SFDRR emphasise building community and system resilience to multiple hazards.

2.3 EXTREME CLIMATE EVENTS AND DISASTERS

2.3.1 Risks of extreme climate events and disasters

Human-induced global warming has already resulted in multiple changes in the climate system, including increases in the intensity and frequency of weather extremes (Hoegh-Guldberg *et al.*, 2018). Climate models project that extreme events may become even more prominent in the near to distant future. An extreme climate or weather event may be defined as the "occurrence of value of a weather or climate variable above (or below) a threshold bear of the range of observed values for a particular variable" (IPCC, 2012). Although sparse observational and operational records in Southern Africa limit the ability to detect trends in extreme events with high confidence, there is strong evidence of an increasing trend for hot extremes, erratic rainfall patterns with increasing intensity, frequency of dry spells and droughts, and the magnitude of storms producing flood events (Davis-Reddy, 2017). Such events pose a severe threat to water resources, food security, sustainable energy, public health, infrastructure, biological diversity, and ecological systems in the region (Ziervogel *et al.*, 2014; Van Niekerk and Le Roux, 2017; Bradshaw *et al.*, 2022).

The biggest impact of climate change may not be that single events become more severe, but that more compounding extreme events may occur in regions with naturally high climate variability and vulnerability (IPCC, 2012; Clarke *et al.*, 2022). The cumulative impact of successive extreme events could have severe consequences on human and ecological systems; for example, sequential flood events could trigger cascades of secondary effects, such as increased sediment delivery that causes infrastructural damage to water purification plants, and nutrient and pesticide input into surface water systems that lead to deterioration of drinking water quality (Woodward *et al.*, 2016). To improve resilience against such impacts and to develop efficient coping strategies, new approaches that integrate the assessment of exposure and vulnerability on the one hand and offer efficient coping strategies on the other are needed (Gallina *et al.*, 2016; Schanze, 2016). Such adaptation strategies must, however, be based on appropriate and reliable downscaled information, and must include communication technologies with early warning and decision support systems for the population and decision makers concerned. It is for all these reasons that a study of the assessment of climate-related extreme events and suggesting risk reduction and resilience-building strategies in QwaQwa by the South African Water Research Commission (WRC) are necessary.

2.3.2 The likelihood of an increase in the frequency of extreme climate events in Southern Africa and the QwaQwa study area

There is evidence that the frequency and intensity of extreme climate events have been increasing globally since the 1950s (Field, 2012; Abrahams *et al.*, 2013; Leigh *et al.*, 2015; Garner *et al.*, 2015), regionally in Southern Africa (Mason *et al.*, 1999; Kruger and Sekele, 2013; MacKellar *et al.*, 2014; Engelbrecht *et al.*, 2015; Davis-Reddy and Vincent, 2017; Perkins-Kirkpatrick and Gibson, 2017), and locally in the Maloti-Drakensberg region (Mohamed and Mukwada, 2019; Mukwada and Mutana, 2023). Climate projections indicate that these extreme events will most likely become more common and more severe in the future, both globally (Abrahams *et al.*, 2013; IPCC, 2014) and in Southern Africa (Field, 2012; Niang *et al.*, 2014; Archer *et al.*, 2018). According to the IPCC (2022), there is high confidence that Southern Africa could experience more heat waves and warm spells that will last longer than in the past. There is medium confidence that droughts over Southern Africa will intensify due to a reduction in rainfall and an increase in evaporation and evapotranspiration, and low confidence that the intensity of rainfall events will increase in the intermediate and long term (Field, 2012). According to regional climate models, shifts in the frequency and variance of extreme climate variables in South Africa could increase in the intermediate and long term (Field, 2012). According to regional climate models, shifts in the frequency and variance of extreme climate variables in South Africa could increase in the future (Lumsden *et al.*, 2009; Schulze, 2012).

Mohamed and Mukwada (2019) recently reported a significant increase in the long-term (1960-2016) annual mean maximum and minimum temperatures for the Maloti-Drakensberg region. It was also clear from Mohamed and Mukwada's (2019) study that the frequency of temperature extremes increased significantly over this 56-year period. For the Orange-Senqu River basin, future trends with regard to rainfall and droughts are unclear, with earlier studies (Hewitson *et al.*, 2005; Knoesen *et al.*, 2009; Crerar *et al.*, 2011; see also Dallas and Rivers-Moore, 2014) projecting increased rainfall in summer and autumn (including extreme rainfall events) and decreased short-term droughts, especially in the higher-altitude eastern part of the catchment. A recent study by Abiodun *et al.* (2019) found, however,

that drought intensity and frequency could increase over the catchment in the future (under increasing global warming levels) if both rainfall and potential evapotranspiration are considered (instead of only rainfall). Rainfall variability is expected to increase in the future, both within and between years (Knoesen *et al.*, 2009; Crerar *et al.*, 2011; Dallas and Rivers-Moore, 2014), which increases uncertainty of when streamflow could occur.

Mountain regions are considered particularly vulnerable to climate change (Mukwada and Manatsa, 2018). According to Mohamed and Mukwada (2019), such regions have spatially complex patterns. The steep elevation results in vertical gradients of microclimates, which in turn influence species distribution, livelihoods, and the availability of water resources (Mohamed and Mukwada, 2019). Considering the important effect that temperature has on other meteorological variables, the warming trend over the Maloti-Drakensberg region could have a serious impact on water availability through increased evapotranspiration from wetlands, rivers, and dams (Mohamed and Mukwada, 2019). The abrupt temperature shifts noted by Mohamed and Mukwada (2019) could have a negative impact on the already stressed aquatic ecosystems in the QwaQwa area, which would put the already water-stressed QwaQwa community at risk of reduced ecosystem service delivery.

2.3.3 Potential impacts of extreme events on the ecological integrity of rivers

A river's natural flow regime is the key driver of its ecological features and functioning (Poff and Zimmerman, 2010; Arthington, 2012), and alterations to the flow regime could change a river's unique character (Bunn and Arthington, 2002; Postel and Richter, 2003). Streamflow, in turn, is influenced by hydro-climatological processes, and even though river health is determined by many interacting variables (e.g. water temperature, nutrient availability, and intact riparia), changes to the flow regime could alter the physical habitat template of a river, which would result in long-term ecological changes (Garner et al., 2015). Extreme climate events, caused by meteorological anomalies, may therefore act as structuring forces that could change aquatic habitat, biodiversity, and ecological functioning in riverine ecosystems (Humphries and Baldwin, 2003; Heino et al., 2009; Fenoglio et al., 2010; Morrongiello et al., 2011; Abrahams et al., 2013; Filipe et al., 2013; Dallas and Rivers-Moore, 2014; Death et al., 2015; Leigh et al., 2015; Woodward et al., 2016). Understanding the ecological effects of these events on the drivers (e.g. hydrology, fluvial geomorphology, and water quality) of riverine ecosystems is crucially important for predicting how aquatic communities (fish, aquatic macroinvertebrates, riparian vegetation, etc.) and consequently human use of the river would respond to changes. Examples of the negative impacts extreme events may have on river systems are presented in Tables 2.1 and 2.2.

Extreme event	Potential impact on driver
Heatwaves	 Increased water temperatures;
	 Decreased dissolved oxygen (DO) concentrations; and
	 Increased decomposition of organic matter.
Droughts	Loss of surface flow, connectivity broken, formation of isolated pools, and
	eventually dry river beds;
	 Decreased baseflow due to loss of groundwater inflow from aquifers;
	 Increased periods of flow intermittence (zero flow);
	 Increased accumulation of sediment due to flow intermittence;
	 Declining water quality (salinisation) in isolated habitats; and
	Loss of aquatic habitats.
Heavy rainfall and	 Increased catchment and channel erosion;
floods	 Increased runoff, sediment delivery, and turbidity;
	 Increased connectivity of aquatic habitats;
	 Scouring of river substrata and loss of habitat;
	 Removal of in situ basal resources; e.g. macrophytes and detritus; and
	Hypoxic blackwater events due to microbial breakdown of organic matter
	washed into rivers.
Wildfires	 Formation of hydrophobic soils, leading to lower infiltration and increased
	runoff and sediment delivery to streams;
	 Loss of terrestrial and riparian vegetation, which could influence, for
	example, water temperature and energy input into the river; and
	 Increased levels of phosphorous and conductivity in some cases.
Courses Miletelesed of all	(2000), Abushamas at al. (2012), Dellas and Divers Maans (2014), Jasman at al. (2014), Lainh at al.

Table 2.1. Potential negative impacts of extreme climate events on physical riverine drivers

Source: Whitehead *et al.* (2009); Abrahams *et al.* (2013); Dallas and Rivers-Moore (2014); Jaeger *et al.* (2014); Leigh *et al.* (2015); Death *et al.* (2015); Woodward *et al.* (2016)

Table 2.2. Potential negative impacts of extreme climate events on riverine biota

Extreme event	Potential impact on response
Heatwaves	Aquatic species are vulnerable when their upper thermal limits are
	exceeded, especially over a longer period such as a heatwave;
	 Increased algal growth and incidence of cyanotoxins;
	 Increased metabolism, especially in poikilothermic species;
	 Reproduction and sex ratios for some species; and
	Changes in community processes and structures.
Droughts	Longitudinal ecological processes are interrupted; e.g. reduced nutrient
	input from upstream that results in reduced productivity;
	 Restriction of migrating and spawning patterns for fish species;
	Riparian vegetation separated from channel due to lateral drying;
	 Loss of riparian vegetation due to a drop in the groundwater level;
	 Extirpation of fish when refuge pools dry up;
	Switch from lotic to lentic conditions, and eventually from lentic to
	terrestrial conditions (vegetation encroachment into the channel); and
	Loss of sensitive species and domination by more tolerant species.
Heavy rainfall and	Displacement of aquatic organisms;
floods	• Expansion of invasive species (e.g. exotic macrophytes);
	Untimely floods could influence natural reproduction cues and cycles of
	freshwater biota; and
	Changes in food resources.
Wildfires	Responses linked to increased water temperature, nutrient input, and
	sedimentation;
	Reduction in macroinvertebrate richness and density; and
	Dominance by more generalist species.

Source: Heino *et al.* (2009); Fenoglio *et al.* (2010); Abrahams *et al.* (2013); Filipe *et al.* (2013); Dallas and Rivers-Moore (2014); Leigh *et al.* (2015); Death *et al.* (2015); Woodward *et al.* (2016); Tanner *et al.* (2022)

2.3.3.1 Predicting the impacts of extreme events on river ecosystems and river use

The effects of extreme climate fluctuations on riverine ecosystems are poorly understood (Fenoglio *et al.*, 2010; Woodward *et al.*, 2016). Although the physical effects of floods and droughts on river ecosystems are relatively well studied, the biological and indirect impacts (e.g. cost of lost ecosystem services) of these events are less known and uncertain (Filipe *et al.*, 2013; Death *et al.*, 2015). As a result, accurate predictions of biological changes, and the impact these may have on biodiversity and ecosystem services, are difficult to make, and will remain of low confidence without supporting field studies. Other factors that may influence the accuracy of such predictions in South Africa are the high levels of natural variability in some systems (e.g. seasonal and ephemeral systems), a general poor understanding of flow-ecology relationships (for the majority of South African rivers), a poor understanding of life-histories for the majority of aquatic species, a general lack of specialist species in rivers naturally subjected to large disturbances (aquatic communities in these rivers are generally dominated by hardy generalist species), and a deteriorating trend in river health (due to increased anthropogenic pressures and poor river management) that increases the vulnerability of river systems.

2.3.4 Potential impacts of extreme events on water river use and human health

There is high confidence that climate change has the potential to reduce the quantity and quality of freshwater and groundwater resources, especially in semi-arid and arid regions characterised by large spatial and temporal climate variability (Dudgeon *et al.*, 2006; IPCC, 2014; Döll *et al.*, 2015; Kusangaya *et al.*, 2017), such as Southern Africa (Scholes and Engelbrecht, 2021). South Africa depends heavily on surface water sources, which provide 85% of the usable water in the country, and is also the main source of potable water (Colvin *et al.*, 2016). It is therefore of great concern that surface water resources are considered to be at the epicentre of projected climate change impacts (Kusangaya *et al.*, 2017).

Climate change is expected to reduce raw water quality and to pose risks to the quality of drinking water (Döll *et al.*, 2015; Klasic *et al.*, 2017; Van Niekerk and Le Roux, 2017; Wang *et al.*, 2022). According to Döll *et al.* (2015), water quality projections are difficult to make; the following factors, however, may contribute to a deterioration of water quality: increased water temperature; increased sediment, nutrient, and pollutant loadings from heavy rainfall; increased concentration of pollutants during low-flow episodes; and flood damage to water treatment infrastructure that disrupts municipal water treatment. A brief overview of the impacts of extreme events on the quantity and quality of surface water is presented below.

2.3.4.1 Extreme heat and heatwaves

Extreme heat may cause unusually high air and water temperatures that can change the physical and chemical properties of water (Abrahams *et al.*, 2013; Dallas and Rivers-Moore, 2014; Woodward *et al.*, 2016; Wang *et al.*, 2022). Such changes may include lower DO solubility, higher inorganic salt concentrations, increased pH levels, accelerated release of nutrients from sludge, increased algal growth and the release of cyanotoxins from heat-tolerant cyanobacteria, higher microbial activity, etc. Higher water temperatures may also increase the oxygen consumption and metabolism of aquatic biota,

and may cause mortality among riverine benthic invertebrates when individuals are pushed outside of their optimal temperature range (Abrahams *et al.*, 2013; Woodward *et al.*, 2016). Estimating water increases from air temperature projections is, however, complex, since factors such as solar radiation, groundwater input, and shading from riparian vegetation may have an insulating or buffering effect (Dallas and Rivers-Moore, 2014).

2.3.4.2 Storms and floods

Floods generally occur when the volume of water in a river exceeds the capacity of the channel to transport the water downstream. During a flood, the velocity and erosive power of the water increase with growing discharge, which dislodges particles from the river bed and banks and uprooting emerged and submerged vegetation (Abrahams *et al.*, 2013). Floods may therefore increase erosion of the mineral substrate, remove aquatic vegetation, and disperse organic material and debris downstream, before depositing it downstream as the velocity is reduced. Heavy rainfall and subsequent runoff from steep, sparsely vegetated slopes, urban substrates, and overflown sewers may wash large amounts of sediment (as well as pollutants in the sediment such as heavy metals, pesticides, and organic compounds), pollutants (including particulate matter, pathogens, and soluble substances), and organic waste into the river (Wang *et al.*, 2022). The first stormflow of the season is known to be particularly high in pollutants and organic matter and leads to an increase in suspended solids and dissolved organic carbon (DOC) and a decrease in DO concentration (Abrahams *et al.*, 2013). The nutrient loading of water bodies may, in combination with higher air and water temperatures, cause algal and cyanobacteria blooms (Klasic *et al.*, 2017). High flows may also improve water quality by diluting the pollutant load and decreasing electrical conductivity (EC) in the receiving waters.

2.3.4.3 Droughts (and very low flows)

Extreme low flows during droughts can be more detrimental to river systems than high flows, due to a higher risk of ecological change and potentially long-term effects (Abrahams et al., 2013; Woodward et al., 2016). Lower rainfall and increased evaporative demand (linked to warming temperatures) during droughts are projected to reduce soil moisture, runoff, and groundwater recharge, which causes water shortages and low flows in rivers (Scholes and Engelbrecht, 2021). In severe droughts, streamflow may be interrupted, which can cause river fragmentation, longer residence time, and the loss of longitudinal processes (e.g. ameliorating effect of continuous flow on water quality, downstream invertebrate drift, etc.). The lower velocity may cause the deposition of suspended solids, which reduces turbidity. Loss of dilution capacity due to very low or interrupted flows may increase the concentrations of dissolved and particulate matter, as well as other parameters such as water temperature, DO, and EC. Also important to note is that the quality of surface water may change due to a higher proportion of groundwater, which may have lower DO and higher nutrient concentrations (Abrahams et al., 2013). During droughts, the effluents from wastewater treatment plants (WWTPs) have a more pronounced effect on water quality due to the lower dilution and self-purifying capacity of water bodies. This could pose a serious risk to ecological and human health in situations where the WWTPs are not operating optimally.

2.3.4.4 Wildfires

Extreme high temperatures, in combination with droughts and lightning storms, may trigger wildfires (Thompson *et al.*, 2016). Wildfires can generate ash that contains minerals and oxidised organic substances that can increase the total nitrogen, phosphorous, organic carbon, and sediment content of freshwater (Wang *et al.*, 2022). Fires can also accelerate soil erosion and change runoff patterns in catchments by destroying surface vegetation and organic soil material that facilitate water infiltration after rainfall (Scott, 1993; Strydom and Savage, 2016). It has been shown that fires on mountain slopes may reduce infiltration by as much as 50%, which increases the likelihood of flash flooding (Merz, 2004). The increased runoff generated from these burnt slopes increases soil removal, which results in increased sediment delivery to river systems (Strydom and Savage, 2016).

2.3.5 Management considerations

Managing water resources under a changing climate is particularly challenging due to the high levels of uncertainty; not only due to natural variability and uncertainties around future greenhouse gas emissions and downscaled climate modelling, but also related to other factors such as the difficulty of quantifying potential freshwater-related impacts (Döll *et al.*, 2015). According to Döll *et al.* (2015), these uncertainties make it very difficult to quantify the impact of climate change on freshwater systems in a deterministic way. Döll *et al.* (2015) propose that water managers identify potential hazards and risks (related to freshwaters) for a broad range of possible future hydrological changes and then determine freshwater risks by assessing the consequences of the identified hazards, also taking into account the level of exposure and vulnerability.

As drastic as climate change impacts can be for surface waters, it is important to emphasise that climate change is not the only stressor on these systems. Additional stressors to river functioning include anthropogenic interventions such as a growing water demand, pollution, introduction of alien species, water abstraction, inter-basin transfers, large-scale damming, land use changes, habitat fragmentation, sediment mining, removal of riparian vegetation, etc. (Kusangaya *et al.*, 2017; Best, 2019). The implication of this is that the total risk to river systems is often underestimated (Thompson *et al.*, 2021).

For example, the deterioration of surface water quality due to increased anthropogenic activities, population growth, and climate change has led to an increase in the accumulation of waste and pollutants in water bodies (Wu *et al.*, 2017). Major rivers in Africa, Latin America, and Asia have been identified as the worst hit in terms of pollution load, and this has been on the increase since the 1990s (UNEP, 2016). Locally, pollution due to metal and pesticide contamination causes the most concern. In the eastern Free State especially, metal pollution in the Elands River in Phuthaditjhaba and the Wilge River in Harrismith has recently been linked to failing wastewater treatment works and the lack of proper refuse removal services (Moloi *et al.*, 2020).

Polluted water affects public health directly or indirectly. This is because polluted water not only limits its utilisation value, it also places an added economic burden on society through treatment costs incurred for more polluted the water. Poor water quality also negatively affects the living standards,

dignity, and social wellbeing of humans (Council for Scientific and Industrial Research [CSIR], 2010; Apeh and Ekenta, 2012; Cullis *et al.*, 2018). Polluted water affects human health in many ways. Waterborne diseases such as diarrhoea, cholera, and bacterial infections are particularly endemic in many parts of the world, including South Africa. These waterborne diseases, especially those leading to diarrhoea, are suspected to cause three to five million deaths yearly among children. These diseases remain among the leading causes of deaths and disability worldwide and continue to dominate the global burden of water-related diseases (CSIR, 2010; Yang *et al.*, 2012). Additionally, Yang *et al.* (2012) reported that although substantial advances in biomedical sciences and public health measures have facilitated control of many infectious diseases in the past century, the world has witnessed an increasing incidence and geographical expansion of emerging and re-emerging infectious diseases.

CHAPTER 3: CASE STUDY: QWAQWA

3.1 BACKGROUND

QwaQwa, a former homeland area for Sesotho-speaking people at the foot of the Maloti-Drakensberg mountain range in the north-eastern corner of the Free State province, South Africa, has been plagued by water insecurity for more than two decades (see Figure 3.1). Phuthaditjhaba, QwaQwa's urban centre, falls under the jurisdiction of the Maluti-a-Phofung Local Municipality, which is responsible for providing basic services to communities, including water and sanitation (see Figure 3.2). The municipality has, however, a poor record of service delivery. Service delivery protests by communities in the Maluti-a-Phofung Local Municipality have been occurring since 2004, with limited or no access to safe drinking water as one of the main grievances (Mail & Guardian, 2004; Gouws et al., 2011; Mdlalane, 2020). Nearly 80% of houses in the urban centre are connected to the municipal water supply (Statistics South Africa [StatsSA], 2011), but this percentage is expected to be considerably lower for informal settlements on the outskirts of town and rural villages higher up in the mountains. Access to the municipal water network does not necessarily guarantee access to clean water, since prolonged interruptions in the water supply have become a common occurrence in QwaQwa (Sekhele and Voua Otomo, 2023). Water insecurity problems came to a head during a severe drought in 2015/2016 that caused a drastic decline in the water levels of the Fika-Patso and Metsi-Matso reservoirs, the main water supply in QwaQwa, which necessitated the Maluti-a-Phofung Local Municipality to erect communal water tanks and deliver water with hired water trucks (Macupe, 2020; Mocwagae, 2020; Mukwada and Mutana, 2023). Despite these efforts, many poor households that were unable to afford buying drinking water were forced to rely directly on potentially polluted local rivers and streams for access to water (Khalane, 2020). Alarmingly, the frequency and intensity of severe droughts are expected to increase in the catchment in the immediate future (Abiodun et al., 2019; Mukwada and Mutana, 2023). Together with other risk factors such as projected increases in minimum and maximum temperatures, as well as evaporation, lower rainfall during the dry winter season, more dry spells and fewer wet spells (see Chapter 4), reduced streamflow in the headwater streams (Mukwada and Mutana, 2023), continued influx of people into the area (StatsSA, 2011), widespread environmental degradation in the catchment (Avenant, 2022), high levels of poverty and unemployment (Noble and Wright, 2013; Melore and Nel, 2020), poor governance and gross mismanagement of resources (Sekhele and Voua Otomo, 2023), insufficient water infrastructure and maintenance, underperforming WWTPs, and improper sewage sludge disposal (Sekhele and Voua Otomo, 2023), communities in QwaQwa remain highly vulnerable to water insecurity - even during years with average or above-average rainfall.


Figure 3.1. Map showing the position of the study area within the Wilge River catchment



Figure 3.2. Map showing the boundaries of the Maluti-a-Phofung Local Municipality

3.2 DIFFERENT SCALES OF SAMPLING AND SITE SELECTION

3.2.1 Scale of sampling

Data collection for the three components of the study – climate and hydrological modelling, ecological risk assessment, and disaster risk assessment – was conducted at different spatial scales. Whereas the climate and hydrological modelling mainly focused at higher (larger) hierarchical scales, such as quinary catchment, river segment, and hydrological response unit (HRU) level, the ecological work was conducted at lower (smaller) scales, such as river reach, morphological unit, and hydraulic biotope level (based on the hierarchical geomorphological classification for South African rivers by Rowntree and Wadeson, 1999). Due to the fact that the ecological sampling focused on the lowest spatial level in the catchment, the site-selection process was predominantly driven by ecological requirements (see Section 3.2.2).

The purpose of the field sampling was to identify potential risk factors linked to field-based knowledge of the selected rivers in the study area, and to serve as focus points for the downscaled climate and hydrological modelling, as well as the social surveys.

3.2.2 Site selection

A total of nine sampling sites were selected for study on the Metsi-Matso, Mphukojwane, Namahadi, and Elands rivers. These rivers, which drain the mountainous areas around Phuthaditjhaba and the surrounding rural villages, were chosen in order to screen the greater QwaQwa area. The sites were selected to represent both sparsely inhabited and relatively undisturbed areas, and densely urbanised zones with a diversity of anthropogenic activities, as well as to represent both elevated areas on hilly slopes and flatter lower-lying downstream areas.

Based on the project proposal and available budget, it was decided to select two sampling sites on each of the Metsi-Matso, Namahadi, and Mphukojwane rivers: one relatively undisturbed site in the upper reaches and one more impacted site in the middle or lower reaches. An additional site was added on the Elands River to reflect the impacts downstream of where the abovementioned rivers join. Other criteria that guided site selection included the accessibility to sites, suitability for sampling, ecological representativity, the presence of critical habitats, and the availability of historical data on the specific river reaches.

In preparation for site selection, a comprehensive literature review was conducted to identify if, and where, previous studies had been conducted in the study area. This was important for obtaining background information on the catchment and various rivers, to collect existing information and ecological data, and to allow for continuity of study where possible. Next, a range of maps representing the physical and ecological characteristics of the catchments (e.g. topography, climate, land use, and ecoregions) were prepared to identify river reaches that would satisfy the primary criteria of the study. In addition, Google Earth (web-based; version 9.165.0.1) was used to investigate land uses and upstream anthropogenic impacts (e.g. weirs, dams, agricultural fields, sand mining, and informal settlements) in the catchment and to identify potential sites in the selected reaches and catchment

sections. In a next step, the potential sites identified during the desktop study were visited by a core team during a ground-truth survey on 6 October 2020. Where necessary, alternative sites were identified and assessed. During the final step, the selected sites were verified during a field visit on 25 February 2021, in preparation for the first sampling visit in April 2021.

3.3 DESCRIPTION OF STUDY AREA

3.3.1 Physical environment

QwaQwa is situated in the north-east of the Free State province, bordering on KwaZulu-Natal to the south-east and Lesotho on the south-west. The area is located in a relatively narrow valley in a section of the Maloti-Drakensberg mountain range, at an elevation of between 1 675 to more than 2 200 metres above sea level (m.a.s.l.) (see Figure 3.3). The mountainous area serves as the headwaters for several streams, including the Mphukojwane, Namahadi, and Metsi-Matso rivers, which join to form the Elands River, and later the Wilge River. The Wilge River is an important source of water for the Vaal Dam complex, contributing 36% of surface water to the Upper Vaal River catchment (mean annual runoff [MAR] = 868 m³/a; Department of Water Affairs and Forestry [DWAF], 2004).

3.3.1.1 Climate

Annual temperatures in QwaQwa range from -5°C in mid-winter to 35°C in mid-summer (Department of Water Affairs, 2011). The mean annual temperature is lower in the higher-lying mountainous areas (13°C) and higher in the lower-lying Phuthaditjhaba area (14°C) (see Figure 3.4). Frost occurs in winter, with occasional snowfall in the higher-lying areas (DWAF, 2004).

QwaQwa falls within the summer rainfall region of South Africa. Rainfall is strongly seasonal, with most of the rain occurring between October and March (DWAF, 2004). Rainfall is the highest in the highaltitude headwaters of the streams (mean annual precipitation [MAP] >950 mm), and decreases gradually towards Phuthaditjhaba (MAP <750 mm) (see Figure 3.4). Longer-term meteorological data (1986-2016) show that Phuthaditjhaba received a mean annual rainfall of 629 mm over the last three decades (Agricultural Research Council Institute for Soil Climate and Water, 2016).



Figure 3.3. Topography of the study area

3.3.1.2 Geology, soil, and erodibility

The study area is mainly underlain by the Karoo Supergroup (Hobbs *et al.*, 2013). The Mphukojwane, Namahadi, and Metsi-Matso rivers have predominantly basalt at the origin of the rivers, with arenite and mudstone in the downstream areas (CEM, 2003). The soils comprise mainly of shallow lithosols, representative of the Glenrosa and Mispah soil forms (CEM, 2003).

The topography of the QwaQwa area is mostly steep due to the mountainous slopes of the Maloti-Drakensberg mountains. This makes the area vulnerable to erosion, which can be aggravated by some land uses and overgrazing. Using an Erodibility Index that considers a combination of factors such as the slope, soil type, rainfall intensity, and land use, the sensitivity of an area to erosion can be expressed (Rooseboom *et al.*, 1992). The headwaters of the study rivers have moderate sediment potential (Erodibility Index value of 10), while the downstream areas have a high sediment delivery potential (Erodibility Index value of 8) (CEM, 2003).



Figure 3.4. Mean annual temperature (top) and mean annual precipitation (MAP) (bottom) of the study area

3.3.1.3 Vegetation and land use

QwaQwa falls within the Drakensberg Afromontane region of Southern Africa. The vegetation comprises predominantly mountain grassland types interspersed with patches of Afromontane forest. The Namahadi River originates in uKhahlamba Basalt Grassland vegetation type, whereafter it flows through the Northern Drakensberg Highland Grassland, Basotho Mountain Shrubland, and Eastern Free State Sandy Grassland vegetation types (see Figure 3.5). The Metsi-Matso and Mphuhokjwane rivers originate in Northern Drakensberg Highland Grassland vegetation type and run through the Eastern Free State Sandy Grassland type in their lower reaches before joining the Elands River.

Land use in the area is predominantly urban (both formal and informal) and agricultural (cattle and goats) (see Figure 3.5). The rural areas around Phuthaditjhaba are extensively used for grazing (Taylor, 2023). Environmental degradation due to overgrazing, soil erosion, veld fires, and soil mining for building is widespread in these areas (Melore, 2017; Melore and Nel, 2020).

The local rivers play an important role in the water supply to domestic, agricultural, and industrial users in the area. The water quality in the river is impacted by diffuse urban runoff (informal and formal settlements), point source discharges (municipal wastewater), and industrial effluent from Phuthaditjhaba (CEM, 2003).



Figure 3.5. Vegetation (top) and land cover types (bottom) in the study area

3.3.2 Socio-economic environment

3.3.2.1 Brief history of QwaQwa

QwaQwa became a self-governing homeland for South Sotho people in 1974 (Slater, 2002). The population of QwaQwa grew from 24 000 in 1970 to approximately 300 000 in 1908, and approximately 450 000 in 1989 (Slater, 2002). The rapid population growth was mostly due to immigrants from whiteowned farms. Initially, these immigrants could choose, with the permission of the local chiefs, where they wanted to settle. However, as the area became increasingly more crowded, immigrants settled in the smaller villages surrounding Phuthaditihaba. These immigrants received a plot on which an informal dwelling could be erected, with no access to agricultural land or basic services. The majority of immigrants in the villages had no access to piped water, sanitation, or electricity, and depended on the rivers and streams for their drinking water. Phuthaditihaba provided better access to improved living conditions and employment opportunities than the more rural villages, and as a result three industrial areas were established in the 1980s, which provided 27 000 jobs by 1992 (Pickles and Woods, 1992; Slater, 2002). QwaQwa was deregulated as a homeland during the transition from the former regime to the new democratic dispensation in South Africa. The withdrawal of government subsidies after 1994 caused many industries to close down or to relocate to other urban centres. This resulted in widespread job losses, with high levels of poverty, unemployment, and income inequality prevailing in QwaQwa (StatsSA, 2011).

3.3.2.2 Present situation

QwaQwa is one of the four towns in the Maluti-a-Phofung Local Municipality; the other three are Harrismith (53 km north-east of QwaQwa), Kestell (36 km north-west of QwaQwa), and Tshiame (41 km north-east of QwaQwa) (Mocwagae, 2020). QwaQwa serves as the administrative headquarters of both the Maluti-a-Phofung Local Municipality and Thabo Mofutsanyana District Municipality. The area has a large number of informal households located at the edge of the urban sections of Phuthaditjhaba and sprawled across the slopes to an altitude of 1 646 m.a.s.l. (Melore, 2017). Land use in the area comprises mostly urban activities (both formal and informal) and agriculture (cattle and goat rearing) (CEM, 2003).

Phuthaditjhaba is the business centre of QwaQwa (see Figure 3.6), and one of the most densely populated areas in South Africa with approximately 22 294 people per km² (Mukwada and Mutana, 2023). The population is growing at a rate of 0.87%, with the present population of 56 251 projected to increase to 78 361 by 2050 (StatsSA, 2011). The majority of the people are black African (99%), with Asian, coloured, and white groups contributing 0.5%, 0.2%, and 0.1% to the population respectively (Melore, 2017). Illiteracy is relatively high (22% of the population), with 8.9% of people older than 20 years being unschooled (Melore, 2017).



Figure 3.6. Residential areas in the study area

The dependency ratio in QwaQwa is high and continues to grow (Melore, 2017). The economically active population (15-64 years) increased from 4.9% in 2001 to 5.3% in 2011, while the proportion of the population older than 64 years increased from 60.6% in 2001 to 62% in 2011 (StatsSA, 2012). According to Noble and Wright (2013), 36.8% of the population of QwaQwa live with material deprivation, 56% with employment deprivation, 22.8% with education deprivation, and 61.4% with environmental deprivation. Around 74.9% of the population of QwaQwa live on the lower poverty line (R604) and 83.4% in the upper poverty line (R1 113) (Melore and Nel, 2020). Approximately 52% of income is from social grants, while 34% comes from daily labour wages (Melore, 2017).

Unfortunately, all of these socio-demographic indicators highlight the high vulnerability of communities in QwaQwa to climate-related hazards.

3.3.2.3 Water crisis in QwaQwa

QwaQwa has been experiencing a chronic water crisis since 2015 (Sekhele and Voua Otomo, 2023). According to Mocwagae (2020), the water crisis has been long in the making, which started with insufficient water planning since the founding of QwaQwa as a homeland in 1974. Factors such as poor infrastructure, poor maintenance and neglect, poor service delivery and gross mismanagement by the local authority, and climate variability all contributed to the present situation where the Maluti-a-Phofung Local Municipality has not been able to sufficiently provide water of good quality to the local community (Macupe, 2020; Mdlalane, 2020; Mocwagae, 2020; Kleynhans, 2021; Sekhele and Voua Otomo, 2023). The water crisis in QwaQwa made national headlines during the 2014 to 2016 drought. The poor state of water service delivery was compounded by a severe drought that resulted in very low dam levels (the Fika-Patso Dam, which supplies 80% of water in QwaQwa, was at 10% capacity level) (Mocwagae, 2020). The Maluti-a-Phofung Local Municipality responded to the severe water shortages by allocating the available water to critical sectors, such as health facilities, and hiring water tankers to deliver water to communities (Kings, 2017; Masuabi, 2020; Karrim, 2020). According to Motaung (2022), wards received water on a weekly basis, with residents having to fetch water from the tankers. Other strategies included the activation of boreholes and distribution of plastic water tanks. The Free State government responded to the crisis by allocating additional funding (of R1.6 billion) to repair pipelines between the Fika-Patso Dam and water reservoirs supplying households (Mocwagae, 2020), while the national government pledged emergency relief funds (R200 million) and 5 000 water tanks to be strategically placed in the catchment. At the community level, residents with the means to buy water did so, while those with access to cars fetched water from neighbouring towns (Macupe, 2020). Residents who could not afford to buy water resorted to fetching water from the rivers for drinking and domestic uses. However, these strategies offered only short-term solutions and are not sustainable on the long term. The 2014 to 2016 drought exposed the high vulnerability of communities in QwaQwa to climate-related hazards.

3.3.2.4 Water infrastructure and services provision

The general state of infrastructure and service provision in the area are relatively poor, with approximately 68% of households not having piped water in their homes, 70% are without access to waterborne toilets, and 11% are without access to electricity (StatsSA, 2011). The road networks and drainage systems are poor, or largely lacking, especially in the informal settlements (Melore, 2017). Furthermore, the mountainous landscape makes the cost of infrastructure provision and maintenance high, with the local authorities struggling to cover these costs due to limited financial capacity (Melore, 2017; Melore and Nel, 2020).

The main source of domestic water is the Fika-Patso Dam (full supply capacity: 29.43 million m³) on the Namahadi River (Department of Water and Sanitation [DWS], 2017). The dam, which came into operation in 1986, supplies 85% of water in QwaQwa. The older Metsi-Matso Dam (full supply capacity: 4.5 million m³), operational since 1976, and the Sterkfontein Dam (full supply capacity: 2 616.95 million m³), operational since 1974, supply the remaining 15% water to QwaQwa (Mocwagae,

2020). The Sterkfontein Dam was established in 1968 to augment the water supply to Johannesburg through the Tugela-Vaal transfer scheme (Orange-Senqu River Commission, 2013).

The three bulk water schemes that provide water to the QwaQwa area are:

- the Fika-Patso scheme, which provides drinking water to the western, central, and northern areas of QwaQwa;
- the Metsi-Matso scheme, which provides drinking water to the south-eastern areas of QwaQwa; and
- the Sterkfontein scheme, which is being developed to augment bulk water to the Fika-Patso scheme. The first phase, commissioned in 2013, supplies bulk water to Tshiame, Mokgholokweng, and the special economic zone, while the bulk water supplies to the northern areas of QwaQwa commenced in 2014 (Free State Department of Cooperative Governance and Traditional Affairs [CoGTA] and Chell Engineering, 2018). A pipeline from the Sterkfontein Dam to the area is planned at a cost of R2.1 billion (Khalane, 2020).

According to the DWS (2017), the demands on the Fika-Patso and Metsi-Matso dams are limited by the capacities of the two water treatment works (WTW) receiving water from the dams. The Fika-Patso WTW has a capacity of 41 million litres per day (ML/d), while the Metsi-Matso (Makwana) WTW has a capacity of 10 ML/d. Although the capacity of the Makwana WTW was increased from 5.08 ML/d to 10 ML/d in 2016, there is a strong possibility that the WTWs would not be able to meet the average annual daily demand due to increased demand, especially during winter and drought situations (DWS, 2017). This implies that any future growth in demand needs to be supplied by the Sterkfontein scheme (DWS, 2017). As a further step to relieve the pressure on surface water resources, groundwater resources are being explored with several boreholes being drilled in the area (CoGTA, 2020).

The water crisis in the area is creating a huge financial burden on the Maluti-a-Phofung Local Municipality, which has spent close to R30 million between 2016 and 2019 on tankers to provide water to residents (Khalane, 2020). According to members of the Maluti-a-Phofung mayoral committee, the main problems include old and decaying infrastructure, the failure to clean blocked waterways, as well as poor revenue collection and usage by the municipality.

Serious problems with water purification and distribution have been reported, especially in 2016 and 2017 when acute water shortages were experienced (*Bloemfontein Courant*, 2016; Kings, 2017; Mocwagae, 2020). The critical state of the WTW in QwaQwa was confirmed by the 2022 Blue Drop Report (DWS, 2022a), which found that no water quality monitoring was conducted for drinking water by the water services authority. This non-compliance presents a serious health risk to the residents who receive water from these supply systems. This is very concerning, considering the poor water quality of the rivers in the catchment (Sekhele and Voua Otomo, 2023). The WWTPs in QwaQwa are also in a poor state, with dysfunctional processes, improper sewage sludge disposal, and frequent infrastructural failures (Moloi *et al.*, 2020; DWS, 2022b; Sekhele and Voua Otomo, 2023). The poor quality of wastewater effluents poses a serious risk, not only to public health, but also to the ecological integrity of natural water resources (Edokpayi *et al.*, 2017; Steffen *et al.*, 2022).

3.3.2.5 Vulnerability and the potential impact of climate change on services

The top three long-term risks identified by the 2023 Global Risks Report (World Economic Forum, 2023) are related to climate change: (1) failure to mitigate climate change, (2) failure to adapt to climate change, and (3) natural disasters and extreme weather events. The fourth risk, biodiversity loss and ecosystem collapse, is strongly linked to climate change. The fact that weak governance could further increase the exposure to these risks was also raised by the report. Studies have shown that both the water management sector and local municipalities in South Africa are poorly prepared for expected medium- and long-term changes associated with climate change (Mukheibir, 2008; Ziervogel *et al.*, 2010). For example, neither the Thabo Mofutsanyane District Municipality nor the Maluti-a-Phofung Local Municipality have conducted any scientific risk assessment, nor do they have functional disaster management plans in place (Maluti-a-Phofung Local Municipality, 2018). A district disaster risk assessment was, however, initiated by the District Disaster Management Centre in 2021.

CHAPTER 4: DEVELOPMENT OF A DOWNSCALED CLIMATE AND HYDROLOGICAL MODEL FOR QWAQWA

4.1 INTRODUCTION

This component of the study investigated the risk factors associated with the projected impacts of expected future changes on the occurrences of hydro-climatically driven events, especially the more extreme ones, such as extremes of temperatures, soil water conditions, extreme rainfall events (including sequential extremes), floods (including multiple day floods), meteorological, as well as hydrological, droughts of varying severities and durations, and soil loss (in both tonnes and tonnes per hectare [t/ha]). The focus was on terrestrial and river ecosystems in the QwaQwa area of the Upper Vaal River catchment in order to develop relevant and appropriate risk reduction strategies useful to the local authorities, to the water and agriculture sectors, as well as to local communities in the study area.

The hydrological component was undertaken in light of projected climate change, which is expected to have a marked impact not only on extremes, but also on the flow regimes of QwaQwa's rivers, such as changes to annual flows, low flows, the variability of flows, the seasonality of flows, and periods of intermittence, which could affect river discharges and affect the vulnerability of aquatic biota and river ecosystems. Ultimately, it was important to interpret the results from this component by considering the potential effects of the projected changes on the ecosystem services delivered by these rivers, and on communities that rely on these water sources.

This chapter presents only an extract from a more comprehensive report on the development of a downscaled climate and hydrological model for QwaQwa, including detailed results and projections for the sub-catchments of the study area (see Volume 2 of this report).

4.2 METHODOLOGY FOLLOWED FOR DEVELOPING THE DOWNSCALED CLIMATE AND HYDROLOGICAL MODEL

4.2.1 Historical databases used

The University of KwaZulu-Natal's Centre for Water Resources Research's Quinary Catchments Database (QnCDB), which contains 50 years (1950-1999) of data on daily rainfall, daily maximum and minimum temperature, daily reference potential evaporation, daily solar radiation, and daily maximum and minimum relative humidity, was used as the core database for the baseline (i.e. historical) studies. Additionally, the outputs from a one arc minute (i.e. ~1.7 x 1.7 km) hydro-climate database, developed in WRC Project K5/1490, were used for the more localised studies. These outputs consist of rainfall information (Lynch, 2004), as well as temperature and temperature-derived information (Schulze and Maharaj, 2004).

4.2.2 Climate change databases used

Downscaled and bias-corrected daily hydro-climatic outputs were available at quinary catchments spatial resolution from six Coupled Model Intercomparison Project Phase 5 (CMIP5) Global Climate Models (GCMs) from WRC Project K5/2833 (Wolski *et al.*, 2022, in Schütte *et al.*, 2022). The six downscaled and bias-corrected GCMs used to produce daily outputs were the following:

- Australian Community Climate and Earth System Simulator;
- Geophysical Fluid Dynamics Laboratory Coupled Model;
- National Centre for Meteorological Research Coupled Global Climate Model, version 5;
- Max Planck Institute Coupled Earth System Model;
- Norwegian Earth System Model; and the
- Community Climate System Model.

This ensemble of very high-resolution climate model simulations of present-day climate and projections of future climate changes over South Africa were produced by the CSIR (Engelbrecht *et al.*, 2019; Engelbrecht *et al.*, 2020), and are termed Conformal Cubic Atmospheric Model (CCAM) projections. As in previous work at the CSIR, the CCAM regional climate model was used to develop the downscaled projections. The six GCM simulations from the CMIP5 archive, based on the emission scenarios described by representative concentration pathways (RCPs) 4.5 and 8.5, were first downscaled to a 50 km spatial resolution globally. The selection of these six GCMs was based on their ability to provide a reasonable representation of the El Niño-Southern Oscillation phenomenon for Southern Africa. The simulations span the period 1961 to 2100, with this study utilising the RCP 8.5 low mitigation scenario for three time periods, namely:

- the *Present*, represented by the 30-year period from 1961 to 1990;
- the *Immediate Future*, represented by 2016 to 2045; and
- the Distant Future, represented by the period 2071 to 2100.

Daily rainfall and minimum, as well as maximum, air temperature projections were bias corrected to the resolution of the QnCDB with the methods described in Wolski *et al.* (2022) for all 5 838 quinaries covering South Africa, Eswatini, and Lesotho. The datasets relevant to the QwaQwa study area were isolated, with the data then analysed for the selected hydro-climatic variables selected for this study, and with the data then used as climate input into the Agricultural Catchments Research Unit (ACRU) model to determine climate change impacts on hydrological responses in QwaQwa.

4.2.3 The hydrological model used: Agricultural Catchments Research Unit (ACRU)

The daily time-step process-based and multi-purpose ACRU agro-hydrological model (Schulze, 1995; Smithers and Schulze, 1995; updates) was applied in this component of the study.

4.2.3.1 ACRU model attributes

The ACRU agro-hydrological modelling system (Schulze, 1995; Schulze and Smithers, 2004; continual updates), which has been, and is currently being, used extensively in water resources and climate change studies in Southern Africa, is centred around the following objectives and attributes (see Figures 4.1 and 4.2):

- It is a daily time-step and conceptual-physical model;
- with variables (rather than optimised parameters values) estimated from physically based characteristics of the catchment; and
- with the model revolving around daily multi-layer soil water budgeting.

As such, the model has been developed essentially into a versatile simulation model of the hydrological and related system (see Figure 4.2), structured to be highly sensitive to climate drivers and to land cover, land use, and management changes on the soil water and runoff regimes, and with its water budget being responsive to supplementary watering by irrigation, to changes in tillage practices, to enhanced atmospheric carbon dioxide concentrations associated with climate change, or to the onset and degree of plant stress, which may change with global warming.



Figure 4.1. General structure and multi-purposeness of the ACRU model Source: Schulze (1995)

ACRU is a *multi-purpose* model that integrates the various water budgeting and runoff production components of the terrestrial hydrological system (see Figure 4.1). It can be applied as a versatile model for design hydrology (including flow routing through channels and dams), crop yield estimation, reservoir yield simulation, ecological requirements, wetlands hydrological responses, riparian zone processes, irrigation water demand and supply, water resources assessment, planning optimum water resource utilisation/allocation, conflict management in water resources, and land use impacts – in each case with associated risk analyses – and all of which can respond differently with climate change.

ACRU can operate at multiple scales as a *point* model or as a *lumped* small catchments model, on large catchments or at national scale as a *distributed* cell-type model with flows taking place from "exterior" through "interior" cells according to a predetermined scheme, as is the case in the QwaQwa, simulations, with the facility to generate individually requested outputs at each sub-catchment's exit.

The model includes a *dynamic input option* to facilitate modelling of hydrological responses to climate or land use or management changes in a time series, be they long-term/gradual changes (e.g. urbanisation changes over time, or climate trends over time), abrupt changes (e.g. construction of a dam), or changes of an intra-annual nature (e.g. crops with non-annual cycles).

The ACRU model has been linked to the QnCDB (Schulze and Horan, 2010) for applications at a range of scales in South Africa, Lesotho, and Eswatini for climate change impacts and other studies.



Figure 4.2. Schematic of major processes represented in the ACRU model Source: Based on Schulze (1995)

4.2.3.2 General structure of the ACRU model

Multi-layer soil water budgeting by partitioning and redistribution of soil water is depicted in a highly simplified schematic in Figure 4.2. Rainfall and/or irrigation application that is not abstracted as interception or converted to stormflow (either rapid response or delayed) first enters through the surface layer and "resides" in the topsoil horizon. When that is "filled" to beyond its drained upper limit (field capacity), the "excess" water percolates into the subsoil horizon as saturated drainage at a rate dependent on respective horizon soil textural characteristics, wetness, and other drainage-related properties. Should the soil water content of the bottom subsoil horizon of the plant root zone exceed its drained upper limit, saturated vertical drainage/recharge into the intermediate and eventually groundwater stores occurs, from which baseflow may be generated at an exponential decay rate that is dependent on geological or aquifer characteristics, as well as the groundwater store.

Unsaturated soil water redistribution, both upwards and downwards, also occurs, but at a rate considerably slower than the water movement under saturated conditions, and is dependent on, inter alia, the relative wetness of adjacent soil horizons in the root zone. Evaporation takes place from water

previously intercepted by the crop's or vegetation's canopy, as well as simultaneously from the various soil horizons, in which case it is either split into separate components of soil water evaporation (from the topsoil horizon only) and plant transpiration (from all horizons in the root zone), or combined, as total evaporation.

Evaporative demand on the plant is estimated, inter alia, according to atmospheric demand (through a reference potential evaporation) and the plant's stage of growth. The roots absorb soil water in proportion to the distributions of root mass density within the respective horizons, except when conditions of low soil water content prevail, in which case the relatively wetter horizons provide higher proportions of soil water to the plant in order to obviate plant stress as long as possible.

It is vital in agro-hydrological modelling to determine at which point in the depletion of the plant-available water reservoir plant stress actually sets in, since stress implies soil water extraction below optimum, the necessity to irrigate, a reduction in crop yield, and lower runoff potential. In modelling terms, this problem may be expressed as the critical soil water content at which total evaporation, *E*, is reduced to below the vegetation's maximum evaporation, *E*_m (formerly termed "potential evapotranspiration"). *E* equals E_m until a certain fraction of maximum (profile) available soil water to the plant, *PAW*, is exhausted (see Figure 4.3). The critical soil water fraction at which stress commences varies according to atmospheric demand (the hotter it is, the sooner stress commences) and the critical leaf water potential of the respective vegetation; the latter being an index of the resilience of the vegetation to stress situations. Plant stress, and a reduction in evaporative losses, however, also occur when the soil is too wet; i.e. soil water content exceeds *PAW*. Furthermore, plant stress, when the soil dries out, can be either mild or severe, as illustrated in Figure 4.3. The various levels of stress are defined as follows:

• Excess *soil water stress* occurs when actual soil water content θ exceeds that at the drained upper limit θ_{DUL} ; i.e. $\theta > \theta_{DUL}$

and total evaporation, E, drops to below its maximum, E_m .

• No soil water stress occurs when the plant can transpire at its maximum rate (i.e. $E = E_m$) with the soil water content then below that of the *DUL*, but exceeding the soil water content at a specified fraction of *PAW* at which plant stress commences, namely θ_{fs} , which in the case where it has been set at 0.4 *PAW*, $\theta_{DUL} > \theta > \theta_{fs}$

or
$$\theta_{DUL} > \theta > 0.4(\theta_{DUL} - \theta_{PWP}) + \theta_{PWP}$$

where θ_{PWP} = soil water content at the permanent wilting point.

- Mild *soil water stress* is experienced when soil water content is below the stress fraction, θ_{fs}, but the plant is still transpiring at more than 20% of its maximum evaporation; i.e.
 θ_{fs} > θ > 0.2 E/E_m
 θ_{fs} > θ > 0.6 (θ_{fs} θ_{PWP}) + θ_{PWP}
- Severe *soil water stress* is defined as the soil water content at which total evaporation has been reduced to below 20% of maximum evaporation; i.e. θ < 0.2 *E*/E_m
 which in this case equates to θ < 0.6 (θ_{fs} – θ_{PWP}) + θ_{PWP}



Figure 4.3. Interrelationships used in ACRU between soil water content and the ratio of *E* : *E*_{tm} Source: Schulze (1995; 2008)

The ratio of E : E_{tm} expresses the level of plant water stress (right) and different levels of stress experienced by plants (left).

4.2.3.3 Generation of stormflows with the ACRU model

Stormflow *Q* is defined as the water that is generated from a specific rainfall event, either at or near the surface in a catchment or sub-catchment, and contributes to flows of streams within that catchment/sub-catchment (see Figure 4.4). It is largely from stormflow events that, for example, reservoirs are filled and design runoffs for selected return periods are computed. Furthermore, the soil detachment process in the production of sediment yield from a catchment is highly correlated with the volume of stormflow from an event. Important statistics on stormflows include annual means, inter-annual variabilities, magnitudes in wet and dry years, and the number of stormflow events per annum that exceeds critical thresholds.

Stormflow can be generated from both the impervious parts of the catchment connected directly to a stream (e.g. paved surfaces, roofs, and permanently saturated areas directly adjacent to a stream – ACRU variable name ADJIMP in Figure 4.4) and from the pervious portions of a catchment. The amount of the stormflow that is generated from the pervious areas (expressed either as a depth equivalent in mm, or as a volume in m^3) in essence depends on the magnitude of the rainfall event (*P* in Figure 4.4) and how wet the catchment is just prior to the rainfall event.

Stormflow, Qs, is computed in the ACRU model (Schulze, 1995; updates) in mm equivalents as

 $Q_s = (P_n - I_a)^2 / (P + I_a + S)$ for $P_n > I_a$

Where:

 P_n = net rainfall (mm); i.e. gross (measured) rainfall minus canopy interception losses

- *I_a* = initial abstractions (mm) before stormflow commences, consisting mainly of the infiltration that occurs between the beginning of the rainfall event and the beginning of storm runoff, plus any depression storage
- S = the soil's potential maximum retention (mm), which is equated to the soil water deficit and is an expression of the wetness or dryness of the soil



Figure 4.4. Schematic of runoff-generating mechanisms in the ACRU model

In ACRU, the soil water deficit, *S*, is calculated by the daily multi-layer soil water budget, and for computations of stormflow, a critical soil depth, D_{sc} (SMDDEP, in m, in Figure 4.4), is defined from which *S* is determined. The depth of D_{sc} accounts for the different dominant runoff-producing mechanisms, which may vary in different climates, as well as with catchment land uses, tillage practices, litter/mulch cover, and soil conditions. This depth is therefore generally shallow in more arid areas characterised by eutrophic (i.e. poorly leached and drained) soils and high-intensity storms, which would produce predominantly surface runoff, but is generally deeper in high-rainfall areas with dystrophic (highly leached, well-drained) soils where interflow and "push-through" runoff-generating mechanisms predominate. For all hydrological simulations in this report, D_{sc} was defined as the thickness of the topsoil.

A major determinant of initial abstractions is soil water content. In order to eliminate estimations of both I_a and S in the equation above, I_a is expressed as a coefficient, c, of S, where c is an index of infiltrability into the soil and varies with rainfall intensity (in the thunderstorm season: smaller c), tillage practice, and surface cover / litter / mulch (Schulze, 1995). For all simulations of baseline hydrological responses in this document, the c of I_a was input as the value assigned on a month-by-month basis (ACRU variable COIAM in Figure 4.4) by Schulze (2004) for the over 70 baseline land cover types found in South Africa that were defined by Mucina and Rutherford (2006). For simulations with other land uses (including fire and degradation / rehabilitation regimes), the monthly values of the c of I_a were taken from Schulze (2008), in which all assumptions are explained.

Not all stormflow generated from a rainfall event exits the catchment on the same day as the rainfall occurs, and the fraction that does depends on the size of the catchment, the catchment's slope, and

other factors (Schulze, 1995). This necessitates a stormflow response coefficient, F_{sr} , to be input, which controls the "lag" of stormflows and is effectively an index of interflow (ACRU variable name QFRESP in Figure 4.4). In all simulations on all sub-catchments in this report, F_{sr} was set at 0.3 – a value which has been found experimentally to be typical in South Africa for use at the spatial scale of quaternary and quinary catchments (e.g. Warburton *et al.*, 2010) when the ACRU model's flow routing option is not used, as in this case.

4.2.3.4 Generation of baseflows with the ACRU model

Baseflows consist of contributions to runoff from the intermediate/groundwater store that had previously been recharged. These contributions are made up of slow and delayed flows to the catchment's streams. In the ACRU model it is assumed that the groundwater store is always "connected" to the stream system. Unlike many other models that compute baseflow indirectly from total runoff hydrographs with an empirically derived "separation curve", ACRU computes baseflow explicitly from recharged soil water stored in the intermediate/groundwater zone (Schulze, 1995).

The stored water is derived from rainfall of previous events that has been redistributed through the various soil horizons and has drained into the intermediate/groundwater store when the deepest soil horizon's water content exceeds its drained upper limit (field capacity). The *rate of drainage* of this "excess" water out of the deepest soil horizon *into the groundwater store* depends on that horizon's soil texture class, which in this report has been input to vary from catchment to catchment according to soil attributes.

The rate of release of water from the groundwater store into the stream is determined by a release coefficient, F_{bff} , which is dependent, inter alia, on the geology, area, and slope of the catchment. F_{bff} operates as a "decay" function, which is input for a catchment as a single value (COFRU in Figure 4.4), but based on experiences with ACRU in many catchment studies, F_{bff} is not a constant decay function, but is enhanced or decreased internally in ACRU, dependent on the magnitude of the previous day's groundwater store, S_{gwp} , such that empirically

$$F_{bff} = F_{bfi} [[[(S_{gwp})^2 - S_{gwp}] / 1000 + 1.3] / 11]$$

Where:

F_{bff}	=	final baseflow release coefficient
F _{bfi}	=	input baseflow release coefficient
S_{gwp}	=	magnitude of previous day's intermediate/groundwater store (mm)

For all simulations in this report, an experimentally determined typical value of F_{bff} of 0.009 (Kienzle *et al.*, 1997) was applied in all quinary catchments in the study area.

4.2.3.5 Generation of peak discharge

The peak discharge is the highest flow rate of a hydrograph (see Figure 4.4). In the ACRU model an estimate of the peak discharge associated with each day's stormflow volume generated for the selected simulation period can be made by assuming a single triangular unit hydrograph. For these simulations,

the Soil Conservation Service peak discharge equation (United States Department of Agriculture Soil Conservation Service, 1972), modified significantly by Schulze and Schmidt (1995), is used. In its modified version

$$q_p = 0.2083 Q_s A / 1.83 L$$

Where:

 q_p = peak discharge (m³.s⁻¹)

Q_s = stormflow depth (mm) from an individual catchment

A = catchment area (km²)

$$= \frac{A^{0.35}MAP^{1.1}}{41.67\gamma^{0.3}\bar{I}30^{0.87}}$$

1.83 = a multiplier that was computed assuming high-intensity rainfall to be associated with annual maximum one-day storms over relatively small catchments

with the lag equation having been developed by Schmidt and Schulze (1984) using several hundred hydrographs from over 20 research catchments at seven hydro-climatically divergent regions in the United States of America and South Africa, and in which

$$A = \text{catchment area (km2)}$$

- *MAP* = mean annual precipitation (in mm)
- Y = mean catchment slope (%), determined in the case of this report from a 200-m digital elevation model
- \bar{I}_{30} = magnitude of the two-year return period 30-minute rainfall intensity (mm.h⁻¹)

As is evident from the above equations, Schmidt and Schulze (1984) found that climatic attributes play a major role in determining a catchment's runoff response, or lag, time. For example, they found that a rainfall event's intensity, best represented by the most intense 30-minute period of that event, significantly affects catchment lag time (Schmidt and Schulze, 1984), as did the MAP, which was used as a surrogate variable to describe the retardation of stormflow as affected by a catchment's vegetative cover. Therefore, by using the lag equation above (i.e. L =), the potential effects of climate change on catchment lag, and hence peak discharge, can be estimated.

4.2.3.6 Generation of sediment yields with the ACRU model

Complex deterministic models are available to estimate erosion processes and sediment transport. However, these models are limited in their application owing to their reliance on calibration. The Universal Soil Loss Equation (USLE) (Wischmeier and Smith, 1978) is an equation that has received recognition as an empirical method that is useful for planning and design purposes. This method is the foundation for other empirical equations that are then applied at the catchment scale to estimate sediment yield, such as the daily stormflow event-based Modified Universal Soil Loss Equation (MUSLE) (Williams, 1975), which has been widely verified worldwide and in South Africa (Kienzle *et al.*, 1997). Sediment yield at any quinary outlet (or that of any other spatial unit) may be estimated using the ACRU model, with the MUSLE embedded and expressed as:

$$Y_{sd} = \alpha_{sy} (Q_v x q_p)^{\beta_{sy}} K x LS x C x P$$

Where:

 Y_{sd} = sediment yield (t) from an individual stormflow event

 Q_v = stormflow volume for the event (m³)

 q_p = peak discharge for the event (m³/s)

K = soil erodibility factor (t h/N.ha)

LS = slope length and gradient factor (-)

C = cover and management factor (-)

P = support practice factor (-)

While the MUSLE coefficients, α_{sy} and β_{sy} , are location specific (Simons and Sentürk, 1992) and are determined for specific climatic zones, default values set at 8.734 for α_{sy} and 0.56 for β_{sy} were, however, used in sediment yield simulations for this research.

Information needed for each quinary catchment when estimating sediment yield thus includes:

- the stormflow volume for each event (using the equations provided earlier in this section, but with the mm equivalent Q being converted to a volume Q_v in m³ by multiplying out for area);
- the peak discharge (m³) for each event (using the equations provided earlier in this section);
- the 30-minute rainfall intensity (mm/h) for the two-year return period, *f*₃₀, used in the peak discharge equation and computed for historical data, as outlined in Schulze (2012), and for climate change studies in South Africa by techniques developed by Knoesen (2011);
- the soil erodibility factor, *K*, determined from the Institute for Soil Climate and Water's soil land types and mapped in detail for South Africa by Schulze and Horan (2010);
- the slope length factor, calculated from each quinary catchment's average slope gradient determined from, for example, a 200-m resolution digital elevation model and an equation developed by Schulze (1979) that relates slope gradient to the slope length factor;
- the cover and management factor, *C*, as determined by Schulze (2004);
- the support practice factor, *P*, not applicable for these simulations under baseline land cover conditions and thus set to 1; and
- a factor proportioning the amount of the sediment generated from a stormflow event and which
 reaches the outlet to the respective quinary catchment on the day of the event, in order to
 account for sediment eroded at one location and which may be stored temporarily, only to be
 subsequently remobilised several times before reaching the catchment outlet, and generally
 defaulted to 0.45 in South African studies.

4.2.3.7 Verification studies on the ACRU model's output

The ACRU model is arguably the most comprehensively verified (as against calibrated) model in Southern Africa (Schulze, 2022), and in addition to verifications of end-product outputs such as streamflow, its components of baseflow and stormflow or sediment yield and internal state variables such as soil water content have been verified against observed data. In addition to verification studies on South African observations, such studies have also been undertaken on observed catchment data from the United States of America, Canada, New Zealand, Germany, Eswatini, Eritrea, and Zimbabwe.

4.2.3.8 Model links to databases

As previously alluded to, the ACRU model has been linked to historical daily climate databases for the 5 838 quinary catchments covering South Africa (Schulze *et al.*, 2010), as well as to daily climate output for present and future scenarios from GCMs, downscaled to quinaries to accomplish the analyses of climate change impacts on climate and water-related variables in QwaQwa.

4.2.3.9 Baseline assessments and in situ assessments

To assess the baseline impacts of climate change, model runs were:

- first undertaken with natural vegetation hydrological input and the most recent soils data derived from terrain unit resolution soils parameters, with inputs from historical (baseline) climate data;
- thereafter, simulations using historical climate data were run using hydrological attributes of actual land uses, which were derived from the 2018 National Land Cover, and with key land uses in regard to hydrological responses being:
 - natural vegetation;
 - degraded areas;
 - wetlands;
 - urban areas with different levels of adjunct and disjunct impervious areas;
 - agricultural areas with different crops and levels of management; and
 - dams and irrigation.
- followed by the same runs, but using climate inputs from multiple GCMs and for different future scenarios.

4.2.4 The spatial resolution of the hydrological assessments undertaken in the QwaQwa study area

4.2.4.1 The quinary catchment: The primary spatial unit of this study

The primary, and coarsest, spatial unit in this study is the quinary catchment, with three altitudinally delineated quinaries making up a DWS quaternary catchment. In total, 5 838 quinaries cover South Africa, Lesotho, and Eswatini. These quinaries are designated the "upper", "middle", and "lower" quinaries of a quaternary catchment, and within the QwaQwa study area three quinaries of the same quaternary are located, namely Quinaries 1219, 1220, and 1221, as shown in Figure 4.5. Altitude was

selected as the key criterion for sub-delineation since it is altitude that largely influences rainfall, as well as temperature and potential evaporation patterns.



Figure 4.5. Water quality sampling points (green rings), as well as additional quinary-related points identified as sub-catchment outlets for hydrological modelling purposes (red squares; left map), and (right map) the quinary catchments with the sub-catchment numbering system

4.2.4.2 Sub-catchments within the quinaries used for modelling in this study

A key focus, however, of the broader research project into which this report makes inputs revolves around the risks associated with sediments, nutrients, and pesticides that result in the deterioration of, inter alia, drinking water quality and public health. For this reason, seven water quality sampling points were identified within the QwaQwa study area, with these located along three key rivers of interest, namely Namahadi, Metsi-Matso, and the Mphukojwane, all of which originate in the Maloti-Drakensberg mountains near Golden Gate National Park in the Great Escarpment Mountains and Highveld ecoregions, and on the outskirts of Phuthaditjhaba merge to form the Elands River (see Figure 4.6).

The names, coordinates, and altitudes of the seven sampling points, used in the climate and hydrological modelling, are listed in Table 4.1.

No.	Site name	Latitude	Longitude	Altitude (m)
1	Namahadi Upper	-28.6495667	28.6976719	1 783
2	Namahadi Middle	-28.5729722	28.6575556	1 655
3	Metsi-Matso Upper	-28.5886667	28.7379722	1 886
4	Metsi-Matso Middle	-28.5643421	28.6972297	1 766
5	Mphukojwane Upper	-28.5938608	28.7609721	1 875
6	Mphukojwane Middle	-28.5523177	28.7990148	1 680
7	Elands Lower	-28.4970001	28.6283333	1 638

Table 4.1. Names, coordinates, and altitudes of the seven sampling points as focus points inthe climate and hydrological modelling



Figure 4.6. The main rivers in the study area in relation to residential areas of Phuthaditjhaba (left), with the location of the seven water quality sampling points located on the Namahadi and Metsi-Matso rivers flowing north-westwards from the south and east, and the Mphukojwane flowing north-eastwards, merging to form the Elands River in the north

With water flows under current and projected future climatic conditions to be modelled with the ACRU model at the seven sampling points, these seven points became logical locations of sub-catchments. However, with much of the hydrological information (such as soils and natural vegetation type) available only at quinary catchment resolution, an overlay of the quinaries with these seven sampling locations yielded a total of 19 modelling sub-catchments, each of which was considered relatively homogeneous with regard to their hydrological responses. The seven sampling and the 12 additional points are shown in Figure 4.6 (on the left), with the sub-catchment numbering system shown in Figure 4.6 (on the right). The flow routing within the study area among the 19 sub-catchments is shown in Figure 4.7, with the locations of the water quality sampling points highlighted in red at the outlets of Sub-catchments 5, 7, 8, 9, 15, 16, and 19.

The mean altitudes and areas of the 19 sub-catchments are provided in Table 4.2, with mean altitudes ranging from 1 677 to 2 521 m.a.s.l. and sub-catchment areas ranging from 1.1 to 72.8 km², totalling 501.3 km². With only seven of the 19 sub-catchments thus delineated and falling entirely within a specific quinary and the remaining 12 sub-catchments still crossing more than one quinary, the percentages of each sub-catchment within each quinary were calculated, so as to provide area-weight specific model input variables. These percentages per sub-catchment are listed in Table 4.3.



Figure 4.7. The flow routing among the 19 sub-catchments within the study area, with the locations of the water quality sampling points highlighted in red

Sub-catchment	Mean altitude (m)	Area (km²)
1	2 373	64.4
2	2 221	19.9
3	2 246	17.5
4	2 278	18.2
5	2 004	9.2
6	1 817	33.9
7	1 857	72.8
8	1 949	14.5
9	1 931	29.6
10	1 786	36.2
11	1 680	2.3
12	1 756	8.1
13	2 521	28.2
14	1 908	39.1
15	2 000	1.1
16	1 802	12.1
17	1 677	17.3
18	1 921	41.1
19	1 815	35.8

Table 4.2. Mean altitudes and areas of the 19 sub-catchments identified for QwaQwa

Sub-	Upper quinary	0/	Middle quinary	0/	Lower quinary	0/
catchment	catchment	70	catchment	70	catchment	70
1	1219	25%	1220	75%		
2	1219	25%	1220	75%		
3	1219	25%	1220	75%		
4	1219	25%	1220	75%		
5	1219	25%	1220	45%	1221	30%
6					1221	100%
7			1220	80%	1221	20%
8			1220	80%	1221	20%
9			1220	50%	1221	50%
10					1221	100%
11					1221	100%
12					1221	100%
13	1219	50%	1220	50%		
14			1220	50%	1221	50%
15			1220	100%		
16					1221	100%
17					1221	100%
18			1220	50%	1221	50%
19					1221	100%

Table 4.3. Percentages of specific quinaries per sub-catchment

4.2.4.3 Further delineation of sub-catchments within the quinaries into hydrological response units (HRUs) for more detailed modelling in this study

In the final modelling configuration, each of the 19 sub-catchments was further sub-delineated into six interlinked and water-related land use-based HRUs to enable the individual assessment of hydrological responses from, inter alia, natural vegetation, degradation, urban areas, wetlands, dams, dryland agricultural systems, and/or irrigation within a sub-catchment, as well as to enable the assessment of the impacts of land uses of an entire sub-catchment and eventually of the entire QwaQwa system. This configuration facilitated modelling the hydrological responses of individual land uses within each sub-catchment under present and projected future climatic conditions.

4.2.5 Location of the sampling sites

The majority of the sampling sites were located in the lower quinary (1221). Only two sites, Metsi-Matso Upper and Mphukojwane Upper, were situated in the middle quinary (1220) (see Table 4.4 and Figure 4.5). None of the sampling sites were situated in the upper quinary (1219).

Table 4.4. I	Location of th	e sampling site	s in relation to the	quinaries and	sub-catchments
					•••••

Sampling sites	Upper quinary (1219)	Middle quinary (1220)	Lower quinary (1221)	Sub-catchment			
Metsi-Matso	Metsi-Matso						
Upstream of MU		Х		8			
MU		Х		8			
MM			Х	9/10			
Namahadi							
Upstream of NU	Х			5			
NU			Х	6			
NM			Х	7			
NL			Х	10			
Mpukojwane							
MPU		Х		15			
MPM			Х	16			
Elands River							
ELU			X	11			
ELL			X	19			

Site codes: MU = Metsi-Matso Upper, MM = Metsi-Matso Middle, NU = Namahadi Upper, NM = Namahadi Middle, NL = Namahadi Lower, MPU = Mphukojwane Upper, MPM = Mphukojwane Middle, ELU = Elands Upper, ELL = Elands Lower

4.3 RESULTS

A summary of the most important findings and trends is presented in the main report. Detailed results for the climate and hydrological modelling are presented in Volume 2 of this report.

4.3.1 Temperature

4.3.1.1 Historical conditions

There is a clear relationship between temperature (both maximum and minimum temperatures) and elevation in the catchment, which implies that the higher-lying areas are cooler on average than the lower-lying areas. This is evident from the maps in Figure 4.8, which show a distinct north-south trend of temperatures in the catchment, with the north-eastern part having the highest temperatures. In summer, mean monthly maximum temperatures range from less than 19°C in the south-west, to more than 25°C in the north-east of the catchment. In winter, mean monthly minimum temperatures could range between less than -3°C in the higher-lying areas in the south (where snow is common in winter), to more than 3°C in the lower-lying areas to the north (see Figure 4.8). Analysis of the historical data also indicated that the annual maximum temperatures are higher in the lower-altitude lower quinary (where the majority of sampling sites are located), and that differences between the cooler upper and middle quinaries are greater than between the middle and lower quinaries. During very hot years (90th percentile), the annual maximum temperature could be 0.66°C warmer than the median, and during very cold years (10th percentile), the annual maximum could be 0.79°C lower than the median.



Figure 4.8. Monthly means for historical maximum temperatures for January and historical minimum temperatures for July for QwaQwa

4.3.1.2 Projections for the future

Both day-time maximum and night-time minimum temperatures are projected to increase markedly in the catchment, but not at the same rate, nor with the same spatial patterns. In the immediate future (mid-2030s), the average January maximum temperature could increase by ~1.4°C in the catchment from the defined present, and by more than 4°C in the more distant future (2080s). This means that in summer (January) the mean daily maximum temperatures in the upper quinary could increase from 24.11°C (present) to 25.47°C in the near future (2016-2045) and 28.18°C in the distant future (2071-2100), while in the lower guinary it could increase from 28.27°C to 29.63°C in the near future, and 32.34°C in the distant future. In winter (July), the mean daily maximum temperature in the upper quinary could increase from 11.06°C (present) to 12.69°C in the near future and 15.63°C in the distant future, while in the lower guinary it may increase from 16.84°C (present) to 18.47°C in the near future and 21.41°C in the distant future. In terms of minimum temperatures, the mean daily minimum temperature in winter (July) is expected to increase by 1.39°C (from -2.83°C at present) in the near future and by 3.84°C in the distant future in the upper quinary, and by 1.41°C (from 2.92°C at present) in the near future and by 3.84°C in the distant future in the lower quinary. In summer, the mean daily minimum temperatures are projected to increase by 1.54°C in the near future and by 4.18°C in the distant future in all the guinaries, which indicates that daily minimum temperatures in summer could increase slightly more than the daily maximum temperatures.

The projections also indicate that the number of moderately cold (Tmin<2°C), cold (Tmin<0°C), and very cold (Tmin \leq 2°C) days per year could decrease markedly in the future. In the high-altitude upper quinary, the number of very cold mornings could decrease from 94.6 days (historically) to only 52.9 days in future, which is a reduction of 57.4%. The reduction could be even higher in the middle and lower quinaries, where very cold mornings could decrease by 72.4% and 81.8% respectively. Days with frost are expected to decrease by 41.2%, 65%, and 70.8% in the upper, middle, and lower quinaries respectively. This is a concerning result considering the importance of frost as a natural disturbance in grassland ecosystems.

4.3.2 Potential evaporation

4.3.2.1 Historical conditions

Potential evaporation in QwaQwa is very high, with annual averages increasing northwards in the catchment from less than 1 200 mm in the cool, mountainous south-west to more than 1 800 mm over the north (see Figure 4.9). As for temperature, a clear relationship exists between elevation and mean annual reference potential evaporation, with evaporation being significantly higher in the lower-lying parts of the catchment than in the higher-lying areas. Historical data show that annual A-pan equivalent reference potential evaporation in the upper quinary varies between 1 491 mm in a "cool" year (10th percentile) and 1 638 mm in a "hot" year (90th percentile), while it varies between 1 759 mm and 1 946 between cool and hot years in the lower quinary.



Figure 4.9. Mean annual A-pan equivalent reference potential evaporation (mm) across QwaQwa under historical climatic conditions Source: Schulze (2008)

On a monthly basis, mean monthly reference evaporation is the highest in December, with 199 mm in the upper quinary, which is equivalent to 6.4 mm/day, and 214 mm; i.e. 6.9 mm/day, in the warmer lower quinary. The lowest average historical reference potential evaporation is experienced in June, when the difference between the cooler upper quinary at 78 mm (~2.6 mm/day) and the hotter lower quinary at 108 mm (~3.6 mm/day) is relatively high.

4.3.2.2 Projections for the future

Mean annual reference potential evaporation in QwaQwa is projected to increase from ~112 mm/a in the upper quinary and ~134 mm per year in the lower quinary in the near future, which represents an increase of approximately 7%. In the more distant future, the increases may range from >350 mm in the cooler upper quinary to ~424 mm in the lower quinary, which represents increases of approximately 22% per year. These projected increases are significant, especially when considering the impact these may have on evaporation from open water bodies, greater irrigation demands, and more rapid drying of soil.

4.3.3 Rainfall

4.3.3.1 Historical conditions

The mountainous areas in the catchment receive higher rainfall than the lower-lying areas, with MAP decreasing from more than 1 100 mm in the highlands (south) to approximately 700 mm in the lowlands (north) (see Figure 4.10). These amounts may, however, vary markedly within and between years. The historical data show that the mean annual rainfall could be approximately 32% lower during dry years (10th percentile) and approximately 37% higher during wet years (90th percentile). For example, the mean annual rainfall in the upper quinary could vary between 699 mm in dry years and 1 423 mm in wet years, whereas it may vary between 528 mm and 1 084 mm between dry and wet years respectively in the lower quinary. The historical data also indicate that years that receive below-average rainfall are more abundant than years with above-average rainfall, and that rainfall variability is higher during the dry season.

The catchment receives its rainfall predominantly in summer, with January receiving the highest rainfall, and June and July the lowest (see Figure 4.11). However, rainfall may vary drastically between years, with an inter-annual coefficient of variation (CV) of between 40% (December) and 170% (July). For example, the mean monthly rainfall may vary between 63 mm in a dry year, compared to more than 330 mm in a wet year (upper quinary).



Figure 4.10. Historical mean annual rainfall in QwaQwa



Figure 4.11. Historical median monthly rainfall in the upper, middle, and lower quinaries

In terms of the sub-catchments, the Metsi-Matso River receives the highest mean annual rainfall (>800 mm) and the Mphukojwane the lowest (~700 mm). In the Namahadi, the mean annual rainfall decreases from approximately 785 mm in the upper reaches to 718 mm in the lower reaches.

4.3.3.2 Projections for the future

Rainfall in QwaQwa is projected to increase in the future (see Figure 4.12). Although the differences in mean annual rainfall appear to be small, they become quite significant in dry and wet years. For example, in a dry year, the mean annual rainfall in the upper quinary is projected to increase by approximately 1.7% (12.7 mm) in the near future and by 17.3% (129.6 mm) in the distant future, compared to an average year when rainfall is expected to increase by 4.1% (40.2 mm) in the near future and by 18.5% (185.8 mm) in the distant future. For wet years, projections indicate increases of 2.7% (36.1 mm) in the near future and 17.7% (229.6 mm) in the distant future. The projections also show that bigger changes could occur at the higher altitude in the higher runoff-producing upper quinary (see Figure 4.13). In the wettest year in 10, it also appears, that in terms of rainfall, the GCMs tend to overestimate the rainfall for dry years and underestimate rainfall for wet years. This highlights the fact that the inter-annual rainfall variability may be underestimated by the GCMs. (A detailed discussion of the verification of the results are presented in Volume 2 of the report.)

Figure 4.14 presents the projected changes in the mean annual rainfall for the 19 sub-catchments identified in the study area. Increases within the sub-catchments range between 21 mm (Sub-catchment 18) in the lowlands to 35 mm (Sub-catchment 1) in the mountainous areas.



Figure 4.12. MAP (mm) across the 19 QwaQwa sub-catchments for the present period, derived from multiple Coupled Model Intercomparison Project Phase 5 (CMIP5) Global Climate Models (GCMs), and the projected increase from the present to the immediate future



Figure 4.13. Annual precipitation (mm) in the wettest year in 10 across the 19 QwaQwa subcatchments for the present period, derived from multiple CMIP5 GCMs, and the projected increase from the present to the immediate future



Figure 4.14. Comparison of mean annual rainfall (mm) across the 19 sub-catchments for the present and immediate future periods as derived from multiple CMIP5 GCMs

The wetter higher-lying areas appear to be more sensitive to dry and wet spells than the drier lowerlying areas. The projections indicate that the number of dry spells in the upper parts of the catchment could increase in the immediate future, with the greatest changes being expected for the two- and threemonth-long dry spells. In contrast, the number of wet spells are projected to decrease by 15% for twomonth, 25% for three-month, and 40% for four-month wet spells in the immediate future respectively. It therefore seems as if more dry spells and fewer wet spells are expected in the near future. This may have severe consequences for agricultural production and water resource management in QwaQwa.

4.3.4 Runoff

4.3.4.1 Historical conditions

The bulk of the runoff in the catchment is generated in the upper quinary (Mean Annual Runoff, MAR, = 505 mm), which delivers nearly three times the runoff generated in the lower quinary (MAR = 177 mm). Interesting to note is that this ratio is higher (6.43 times) during dry years, and lower (2.32 times) during wet years. The historical data show that in dry years, the MAR in the upper quinary may be reduced to 238 mm, compared to 829 mm in wet years. In the lower quinary, the MAR of 117 mm may be as low as 37 mm in dry years and as high as 357 mm in wet years.

The highest runoff generally occurs in summer, with February being the month with the highest median monthly runoff (see Figure 4.15). Considering that median monthly rainfall peaks in January, it is clear that there is a lag of approximately one month between rainfall and runoff in the catchment.


Figure 4.15. Median monthly runoff from baseline land cover under historical climatic conditions for the upper, middle, and lower quinaries in QwaQwa

Intra-annual variability in runoff is high in the catchment, but is lower in the upper quinary (where runoff is higher) and higher in the lower quinary (where runoff is lower).

4.3.4.2 Projections for the future

The projections show that MAR in the upper quinary may increase by nearly 7% from the present day (1961-1990) to the near future (2016-2045), and by up to 35% in the distant future (2071-2100), which indicates an acceleration of climate effects into the future. However, the biggest change is expected to occur in the drier lower quinary, where the acceleration effect of climate change is expected to be even higher than in the upper and middle quinaries. In the lowlands, the MAR is projected to increase by nearly 16% (19.2 mm) in the near future, and by 68% (84.7 mm) in the distant future from present-day conditions (122.9 mm). In dry years (1:10 year low-flow conditions), the MAR in the lowlands could increase by 12% in the near future, and by 65% in the distant future, while in wet years (1:10 year high-flow conditions), it could be 11% and 61% higher in the near and distant futures respectively. These increases are markedly higher than the increases in the mean annual rainfall projected for the lower quinary. These projected changes are seen more clearly when expressed as ratio changes in Figure 4.16, where a ratio change of, for example, 1.6 represents a 60% change. The acceleration of projected climate change as time moves on is clearly evident in Figure 4.17, with the changes from the present to the distant future ratios being several times those of present to the near future.



Figure 4.16. Runoff ratios from baseline land cover conditions for the near and distant futures to the present for the upper, middle, and lower quinaries in the QwaQwa study area

The maps in Figure 4.17 present the MAR (derived from multiple GCMs) for the present period spatially (top), as well as the difference between the present and the near future (bottom). The spatial range in MAR under the present climate is from 311 mm in the south-western mountains to 59 mm in the drier north-west. This represents a 5.3 times difference from the highest sub-catchment to the lowest, compared with only a 1.6 times difference in the case of rainfall (979 mm compared to 604 mm), which illustrates the sensitivity of runoff to rainfall. Changes in MAR from the present to the immediate future increase from 29.4 mm in the mountains of the south-west to 14.4 mm in the lower-rainfall lowlands (see Figure 4.17). These changes represent an average increase in MAR of 17.9% in the near future, with the percentage increase being only 9.5% in the mountains, while they go up to 25.9% in the hydrologically more sensitive lowlands, compared with average increases of only 3.5% for mean annual rainfall (range: 3.4-3.6%) and 4.4% (range: 4.3-4.6%) in the wettest year in 10. These statistics once again illustrate the high sensitivity of runoff to changes in rainfall, as well as that the sensitivities are not the same spatially.

It should be noted that although the runoff derived from the GCMs matched the runoff derived from the historical climate data nearly perfectly, the GCM-derived runoff overestimated historically derived runoffs, when averaged over the 19 sub-catchments and expressed as percentages, in the high-flow summer months from December to March, but underestimated runoff in the critical spring period from August to November. A detailed discussion of the validation of the runoff derived from the GCMs are presented in Volume 2 of the report.



Figure 4.17. Average annual local runoff (streamflow on map, mm) across the 19 QwaQwa subcatchments for the present period, derived from multiple CMIP5 GCMs, and the projected increase from the present into the immediate future, all under baseline land cover

4.3.5 Stormflows

4.3.5.1 Historical conditions

In QwaQwa, stormflows, defined as the component of runoff generated at or near the surface within a catchment from a specific rainfall event, occur predominantly in summer, particularly in the steeper and higher-lying upper quinary. For example, in January and February stormflow may contribute up to 70% to the total runoff in the upper quinary, compared to less than 30% in winter. In the lower quinary, stormflow contributes approximately 60% to the runoff in summer, and 20% or less in winter.

4.3.5.2 Projections for the future

The average annual local stormflow (mm) across the 19 sub-catchments in QwaQwa for the present period (derived from multiple CMIP5 GCMs) is projected to increase in the immediate future (see Figure 4.18). The average present GCM-derived stormflow simulations range from 202 mm in the mountains to only 45 mm in the lowlands. This represents a decrease by a factor of 4.5 compared with a decrease by a factor of only 1.6 in the case of mean annual rainfall. The projected changes in the mean annual stormflows from the present, represented by the mid-1970s, to the immediate future, represented by the 2030s, are presented in Figure 4.18. The map shows increases into the future, ranging from 28 mm equivalent in the mountains to only 8 mm in the lowlands. However, when expressed as percentage increases, the increase in the mountains is only ~13.6%, compared to an increase of up to 21.8% in the lowlands. However, it should be noted that stormflows seem to be overestimated for the mid-summer months and underestimated from August to November.

4.3.6 Baseflows

4.3.6.1 Historical conditions

Baseflows, defined as the contribution to runoff from earlier rainfall events where rainfall has percolated through the soil horizon into the vadose and groundwater zones, contribute to streamflow when the river channel is connected to groundwater. Baseflows are therefore more evenly distributed over the year and are important for sustaining streamflow during the dry, or non-rainy, season. In QwaQwa, baseflows are higher in wet years than in median and dry years. The historical data also show that baseflows are higher in the upper quinary than in the middle and lower quinaries. It also indicates that baseflows are the main contributor to runoff in the dry season (April to August), whereas stormflows are more dominant during the wet season.

4.3.6.2 Projections for the future

Mean annual local baseflows (mm) are projected to increase across the 19 QwaQwa sub-catchments from the present period (derived from multiple CMIP5 GCMs) to the immediate future for baseline land cover conditions (see Figure 4.18). The average present GCM-derived baseflow simulations of 44.1 mm equivalent (compared to 97.6 mm for stormflows) range from 109.4 mm (compared to 202 mm for stormflows) in the highlands to only 9.8 mm in the lowlands. This represents a decrease by a factor of

11.2 (compared to 4.5 for stormflows and only 1.6 in the case of mean annual rainfall). Baseflows are thus firstly considerably lower than stormflows and, secondly, spatially far more sensitive than stormflows.

The projected changes in mean annual baseflows from the present, represented by the mid-1970s, to the immediate future of the 2030s are shown Figure 4.18. This shows that baseflows are likely to increase in the future, with increases ranging from 1.8 mm in the mountains to 6.1 mm in the lowlands. However, when expressed as percentage increases, it is clear that the increase in the mountains is only \sim 1.6%, compared with percentage increases of up to 46.9% in the lowlands (see Figure 4.19).



Figure 4.18. Projected changes in mean annual baseflows from the present to the immediate future in mm and as a percentage

The projections also indicate that the recharge to groundwater is likely to increase in the near and distant future, especially in the lower-lying part of the study catchment. Recharge in the lowlands could increase by nearly 7% in the near future and by 38% in the distant future. In the upper quinary, recharge is projected to decrease slightly in the near future and increase by only 1.7% in the distant future.



Figure 4.19. Average annual local stormflow (mm), under baseline land cover conditions, across the 19 sub-catchments for the present period, derived from multiple CMIP5 GCMs, and the projected increase from the present into the immediate future

4.3.7 Accumulated streamflow

In streamflow analysis, it is important to remember that the runoff from the upper quinary joins that of the middle quinary to converge downstream with the local runoff from the lower quinary. The accumulated flow from the entire upstream area is referred to as accumulated streamflow.

It is clear from Figure 4.20 that the mean annual accumulated streamflows for the present (as derived from the GCMs) follow the main river system in the catchment, the Namahadi, very closely. The accumulated streamflows vary greatly between the 19 sub-catchments, from 0.17 million m³/a to 73.47 million m³, depending primarily on the size and rainfall of the sub-catchments. These GCM-derived statistics compare very favourably with the average accumulated streamflows derived from historical climate data for the same 1961-1990 period, which have a range from 0.18 million m³ to 73.55 million m³.

The projections show that the mean annual accumulated streamflows are likely to increase from present conditions into the immediate future (see Figure 4.20), with increases ranging from 0.027 million m^3/a in the small Sub-catchment 15, to 10.53 million m^3/a in the downstream exiting Sub-catchment 19, with an average increase of 2.93 million m^3/a (see Figure 4.21).



Figure 4.20. Projected changes in mean annual accumulated streamflows from the present to the immediate future in m³/a and as a percentage, all under baseline land cover conditions



Figure 4.21. Average annual accumulated streamflows, under baseline land cover conditions, across the 19 QwaQwa sub-catchments for the present period, derived from multiple CMIP5 GCMs, and projected increases from the present to the immediate future

4.3.8 Peak discharges

Peak discharges refer to the highest flow of a hydrograph, and is particularly relevant for considering sediment yield in a catchment. Peak discharge is not only a function of stormflow volume in a catchment, but is significantly correlated with catchment area, main channel length, mean slope gradient, and vegetation cover (Fu *et al.*, 2020). Due to these factors being different for the various sub-catchments in the study area, a Peak Discharge Index (PDI) per unit area was used to compare peak discharge between the various sub-catchments.

Figure 4.22 shows firstly that peaks per unit area are considerably higher in the steeper and higher runoff-producing upper quinary than in the middle and especially the flatter lower quinary. Secondly, it shows that the high PDI is particularly a late summer phenomenon, most likely associated with thunderstorm activity, with the highest unit peaks in February, and with a gradual build-up from spring though early summer to late summer.



Figure 4.22. Monthly means of the Peak Discharge Index (PDI) under baseline land cover conditions in the upper, middle, and lower quinaries of the QwaQwa study area

It is clearly evident from Figure 4.23 that the accumulated average annual peak discharge under present climatic conditions becomes progressively larger from the various sources in the east, south, and west of the QwaQwa study area to the river exit in the north, where it exceeds 2 200 m³/s. From the present to the immediate future, Figure 4.23 shows increases in peak discharges in a downstream direction, ranging from <5 m³/s to >250 m³/s at its exit from the study area, which indicates that this area might be more prone to floods in the future.



Figure 4.23. Mean annual accumulated peak discharge across the 19 QwaQwa sub-catchments for the present period as derived from multiple CMIP5 GCMs, and projected increases from the present to the immediate future, all under baseline land cover conditions

4.3.9 Sediment yields

4.3.9.1 Historical conditions

The historical data show that there are large differences between the amount of sediment delivered to the river channels in the mountainous, wetter upper quinary and the flatter, drier lower quinary. Under historical climatic conditions, ~2 100 t/km².a of sediment are generated in the upper quinary of QwaQwa on average, compared with ~700 t/km².a in the middle quinary and only ~100 t/km².a in the drier lower quinary (see Figure 4.24). However, in a 1:10 wet year, these values increase from ~2 100 to 4 700 t/km².a in the upper quinary, from 700 to 3 200 t/km².a in the middle quinary, and from approximately 100 to <400 t/km².a in the lower quinary (see Figure 4.24). These increases represent amplifications of 2.2 times in the upper, 4.6 times in the middle, and ~4 times in the lower quinaries. Conversely, in a 1:10 dry year, the tonnages of sediment loss decrease to 1 000, 150, and to around zero for the upper, middle, and lower quinaries respectively.



Figure 4.24. Annual sediment yields in t/km².a under baseline land cover conditions in the driest year in 10, a median sediment production year, and a 1:10 year high for the upper, middle, and lower quinaries in the study area

On a month-by-month basis, Figure 4.25 shows that the peak sediment-producing month is February, with a rapid decline thereafter. It increases again from August onwards when rainfall events are infrequent and, because of the more frontal type rainfall, of low intensity. The production of sediment is, however, highly variable, with the inter-annual CV in excess of 200% for most months, and up to 500% in April and May. The results indicate that sediment yields are considerably more sensitive to climate than, for example, runoff.



Figure 4.25. Monthly means of sediment yields under historical climatic and baseline land cover conditions from the upper, middle, and lower quinaries in the study area

4.3.9.2 Projections for the future

The average annual accumulated sediment yield is projected to increase from present values across the 19 sub-catchments (Figure 4.26). The top maps in Figure 4.27 represent the present sediment yields, as derived from the GCMs, in tonnes and in t/ha, while the projected increases from the present into the immediate future are shown in the bottom maps in tonnes and in t/ha. The accumulation of sediment yields is clearly visible in the top left map, while the top right map shows which sub-catchments are sensitive to sediment yield production. It is clear that the mountainous areas, with their steep slopes, are particularly vulnerable, as well as some downstream sub-catchments where soil type plays a major role.

The projected changes in mean annual sediment yields from the present into the immediate future (see Figure 4.26), in t/ha.a (bottom right) and as a percentage (bottom left), range from 0.05 to 2.29 t/ha.a with an average of 0.42 t/ha.a, which converts to percentage changes of 7.9% to 25.3% with an average of 20.5%.



Figure 4.26. Projected changes in mean annual sediment yields from the present to the immediate future in t/ha.a and as a percentage, all under baseline land cover



Figure 4.27. Mean annual accumulated baseline sediment yields in tonnes and in t/ha across the 19 sub-catchments for the present period, derived from multiple GCMs, and projected increases from the present to the immediate future in tonnes and in t/ha

4.4 KEY FINDINGS AND TRENDS

The QwaQwa study area, despite being just over 500 km² in extent, covers a wide range of altitudes and an associated range of climates across a small space, and with that an amplified range of hydrological responses such as stormflows, baseflows, accumulated streamflows, peak discharges, and sediment yields. This physiographic diversity is rendered even more complex by the overlay of people living in different densities and levels of urban formality across the 19 sub-catchments into which the area was delineated.

Given the above overall perspective, the following observations from hydrologically related modelling were made in the QwaQwa; firstly under baseline conditions of natural vegetation, and secondly considering actual land uses.

4.4.1 Rainfall

The relationships between rainfall and altitude in QwaQwa is complex and non-linear, with the highest rainfall occurring before the highest altitudes.

The highest altitude sub-catchment, Sub-catchment 1, receives a mid-summer monthly January average rainfall of ~170 mm, while the sub-catchment with the lowest altitude, Sub-catchment 17, receives only ~110 mm during January, on average.

Monthly rainfall totals are highly variable from one year to the next, with the CV (%) of monthly rainfall being the lowest in high-rainfall summer months at ~40-50% and the highest in the low-rainfall winter months at ~150%.

Climate change projections of annual rainfall in this area are for an increase of ~8% in the immediate future and ~12.5% in the more distant future. Furthermore, in the future relatively more rainfall is projected in mid-summer and relatively less in the winter months. The highest absolute (i.e. mm) projected change in rainfall in the immediate future in the high-rainfall mountainous sub-catchments is ~30-35 mm per month.

4.4.2 Potential evaporation

Historically, potential evaporation in QwaQwa is the lowest in the cooler higher-altitude upper quinary at 1 550 mm/a, increasing to 1 860 mm in the hotter lower-altitude lower quinary, with the range in the 1:10 hot year from 1 650 to 1 950 mm.

Under climate change conditions, annual potential evaporation is projected to increase by ~7% in the immediate future and by a very marked ~22% in the more distant future.

4.4.3 Runoff

The median monthly runoff is the highest in the high-rainfall upper quinary of the study area, peaking at 80 mm equivalent in February, and the lowest in the drier lower quinary at <20 mm, with February again the month of the highest runoff.

However, the month-month (CV%) of runoff is very high throughout the year – in the range of 60% to 150% – with the highest values at 150% in spring (owing to uncertainty of when the rainy season sets in) and the lowest values in mid-summer with its more predictable rains.

Spatially, the CVs of runoff are the highest in the lower-altitude quinaries with their lower and less predictable runoffs, with values of \sim 100% in mid-summer months, while, conversely, they are the lowest in the higher-runoff upper quinaries with values of \sim 70% in mid-summer months.

Projected changes in runoff in the immediate future are expressed in absolute (i.e. mm) terms and/or as relative (%) changes, and while mm (and hence m³) changes are the highest in the high-runoff mountainous areas, those are also the areas with the lowest percentage increases (~7% in a median year vs ~16% in the lower quinaries).

4.4.4 Baseflows

Baseflows as a contributor to total runoff are relatively (%) higher in the winter months from April to August than in the summer months.

While baseflows derive from drainage of soil water into the groundwater zone, they lag behind drainage, and baseflows are also steadier throughout the year compared to drainage.

Projected changes in baseflows with climate change, as a percentage, are the lowest in the high-rainfall upper sub-catchments of QwaQwa and, conversely, are the highest in the lower-rainfall sub-catchments.

4.4.5 Stormflows

The sediment-laden stormflow component of runoff is the highest in spring and summer and the lowest in winter.

There is a projected annual increase of between 13% and 21% in stormflows in the QwaQwa area in the immediate future, with absolute changes in mm equivalents the highest in the high-altitude mountains at 27 mm, and down to <10 mm in the drier lowlands.

4.4.6 Peak discharge

With peak flows depending not only on rainfall intensity and the catchment's slope, but also on the area of a catchment/sub-catchment, for certain comparative purposes peak discharge was expressed as a PDI in m³/s.km². PDIs were found to be approximately three times higher in the upper quinary than in the middle quinary and approximately eight times higher in the upper than in the lower quinary of QwaQwa.

In the immediate future, the biggest changes in PDIs will be in the lower-altitude/lower-rainfall subcatchments, with the percentage increases in PDIs in the immediate future ranging from 12% to 17%, with significant potential impacts on sediment yields.

4.4.7 Sediment yield

Under historical climatic conditions, on average ~2 100 t/km².a of sediment are generated in the upper quinary of QwaQwa compared with ~700 t/km².a in the middle quinary to only ~100 t/km².a in the drier lower quinary of QwaQwa; i.e. major differences occur between the hilly high-rainfall upper quinary and the flatter, drier lower quinary.

However, in the 1:10 wet year, these values increase from ~2 100 to 4 700 t/km².a in the upper quinary to 3 200 t/km².a (from 700) in the middle to <400 t/km².a (from ~100) in the lower quinary of QwaQwa, with these representing amplifications of 2.2 times in the upper, 4.6 times in the middle, and approximately four times in the lower quinary, while in the 1:10 dry year the tonnages of sediment loss decrease to 1 000 vs 150 to around zero for the upper, middle, and lower quinaries respectively.

Together with the inter-annual CVs (%) of sediment yields being ~500% in the lower and ~350% in the upper quinary, a major conclusion is that sediment yields are considerably more sensitive to climate than, for example, runoff.

In the immediate future, sediment yields in QwaQwa are projected to increase by 15% to 25%.

CHAPTER 5: ECOLOGICAL RISK ASSESSMENT OF THE HEADWATER STREAMS IN QWAQWA

5.1 INTRODUCTION

The bulk of potable water supply sources in South Africa rely on freshwater systems, and particularly rivers (providing 85% of the country's total water consumption) (World Wildlife Fund, 2016). The water quality of a river system is indicative of its ecological health. Rivers are important ecosystems for humans and the general wellbeing of the earth and all its inhabitants. Some communities also depend on river water for their domestic supply. It is particularly important that the water quality of such rivers is regularly monitored and assessed by local communities. Informed policy decisions for sustainable water resources management also require the acquisition of scientific data on water quality (Revermann *et al.*, 2018). This section therefore aims to assess the monitored physico-chemical characteristics of water samples and chemical element analyses of the rivers under study. The ecological health of the rivers was also evaluated using a battery of toxicity tests.

5.2 DATA COLLECTION AND ANALYSES

5.2.1 Components included in the ecological risk assessment

The ecological risk assessment included a range of components under three broad categories: physical and chemical water quality, ecotoxicological tests, and river health concerning the hydromorphological status (see Table 5.1). The aim of these assessments was to identify potential hazards to human and environmental health associated with extreme weather events in the study area.

5.2.2 Field sampling

Two field trips, made by the ecological team, were undertaken to collect data at the selected sampling sites in the catchment in April 2021 and March 2022. During the first sampling visit, data were collected at seven sites. Upon reflection on the success of the first sampling visit, two more sites were added to better account for the individual pollution loads from the various streams in order to improve the characterisation of the identified hazards: Elands Upper was added downstream of where the Mphukojwane joins the Elands River, and Namahadi Middle was added upstream of where the Metsi-Matso flows into the Namahadi River (see Figure 5.1). In addition to the combined field trips, individual data-collection visits were made as required by the respective researchers.

Table 5.1. Components included in the ecological risk assessment of the streams

Physical, microbial, and chemical water quality	Ecotoxicological tests	River health	
System variables Water temperature, pH, DO, total suspended solids (TSS), total dissolved solids (TDS), EC, chemical oxygen demand (COD), biochemical oxygen demand (BOD), redox potential	River sediment Reproduction test (<i>Folsomia</i> <i>candida</i>) Metal analysis	Aquatic habitat Instream and riparian habitat assessment	
Anions* Bromide, chloride, carbonate, fluoride, nitrate, nitrite	River water Algal growth inhibition test (<i>Raphidocelis subcapitata</i>) Acute mortality test (<i>Daphnia</i> <i>magna</i>) Protozoan toxicity test (<i>Tetrahymena thermophila</i>)	Aquatic biota Fish Response Assessment Index (FRAI)	
Metals* Al, As, Ca, Cu, Mg, Mn, Fe, Zn	In vitro tests* Yeast reporter gene assays Ames test Micronucleus assay		
Organic micropollutants (OMPs)* 35 OMPs were included in the analyses Microbial* Faecal coliform bacteria <i>Escherichia coli</i>			

in the study area

*Note: These components were not included in the original scope of the study and were funded from other non-WRC projects/sources.

5.2.3 Description of sampling sites

The geospatial characteristics of the nine sampling sites are presented in Table 5.2.

The source areas of the Metsi-Matso, Namahadi, and Mphukojwane rivers fall in the Eastern Escarpment Mountains 15.03 Level 2 ecoregion (see Figure 5.2). This ecoregion, which occurs in highaltitude mountainous areas, is characterised by high rainfall and moderate to low mean annual temperatures (Kleynhans *et al.*, 2005). Vegetation in these areas consists predominantly of grassland types, including Afro-montane and Alti-mountain vegetation types. The upper sampling points, as well as two of the middle sampling points, are all situated in the Eastern Escarpment Mountains 15.03 Level 2 ecoregion, while the sites in the lower Namahadi and the Elands rivers occur in the Eastern Escarpment Mountains 15.01 Level 2 ecoregion. The geomorphological characteristics of most of the sites resemble the Upper Foothills geomorphological zones (see Figure 5.2).

Even though the Metsi-Matso River is considered to be relatively natural, the Namahadi and Mphukojwane rivers are described as moderately modified and not intact (see Figure 5.3).

No Freshwater Ecosystem Priority Areas were identified in the QwaQwa area (Quaternary Catchment C81F) (Nel *et al.*, 2011).



Figure 5.1. Sampling sites on the Metsi-Matso, Namahadi, Mphukojwane, and Elands rivers for ecological sampling

Site	ELL	ELU	MPM	MPU	NL	NM	NU	MM	MU
River	Elands	Elands	Mphukojwane	Mphukojwane	Namahadi	Namahadi	Namahadi	Metsi-Matso	Metsi-Matso
Latitude:	-28.47425	-28.497	-28.5523177	-28.5938608	-28.5431111	-28.5729722	-28.6495667	-28.5643421	-28.5886667
Longitude	28.834	28.8283333	28.7990148	28.7609721	28.8493333	28.8575556	28.8976719	28.8972297	28.9379722
Altitude (m)	1 670	1 638	1 680	1 875	1 651	1 655	1 783	1 766	1 886
Ecoregion Level 1	Eastern	Eastern	Eastern	Eastern	Eastern	Eastern	Eastern	Eastern	Eastern
	Escarpment	Escarpment	Escarpment	Escarpment	Escarpment	Escarpment	Escarpment	Escarpment	Escarpment
	Mountains	Mountains	Mountains	Mountains	Mountains	Mountains	Mountains	Mountains	Mountains
Ecoregion Level 2	15.01	15.01	15.03	15.03	15.01	15.01	15.03	15.03	15.03
Geomorphological	Lower	Lower	Upper	Mountain	Lower	Upper	Upper	Upper	Upper
zone	Foothills	Foothills	Foothills	Stream	Foothills	Foothills	Foothills	Foothills	Foothills

Table 5.2. Summary of the sampling sites on the Namahadi, Metsi-Matso, Mphukojwane, and Elands rivers



Figure 5.2. Level 2 ecoregions and geomorphological zones at the study sites



Figure 5.3. National Freshwater Ecosystem Priority Areas river and wetland conditions in the study area

5.2.3.1 Namahadi River

The Namahadi River is a perennial stream originating at an elevation of more than 2 000 m.a.s.l., and the main source of freshwater in QwaQwa. The Fika-Patso Dam, built on the river, has a full supply capacity of 26.3 million m³, and supplies the western, central, and northern areas of QwaQwa (DWS,

2017). The dam wall, 65 m high and 300 m wide, represents a severe barrier to the natural flow regime and continuity of ecological processes in the river.

(a) Namahadi Upper

Namahadi Upper is situated in the upper reaches of the river, approximately 3.29 km from its origin. The geomorphic features of the river channel resemble an Upper Foothills zone with a moderately steep slope and the river bed dominated by bedrock, boulders, large and small cobbles, and pebbles (see Appendix A: Plate 1). The river channel is between 3 m and 5 m wide, with shallow pools and riffle-rapid sections alternating in this section of the river.

The lower riparian vegetation comprises predominantly of grass with exotic trees (black wattle [*Acacia mearnsii*]) encroaching into the marginal zone of the river. The upper riparian zone is predominantly covered by grass interspersed by small shrubs, with clumps of oldwood (*Leucosidea sericea*). The upland areas are relatively steep with signs of erosion and occasional dongas forming.

This section of the Namahadi River flows past the village of Pitseng. Pitseng is a small village with a population of 131 persons (28 households) (StatsSA, 2011). The nearest dwellings are approximately 130 m away from the sampling point. The villagers use groundwater, stored in reservoirs, for drinking, but use the river for doing laundry.

(b) Namahadi Middle

Namahadi Middle is situated on the Namahadi River approximately 1.38 km upstream of where it joins the Metsi-Matso River (see Appendix A: Plate 2). In this section, the river channel is classified as an Upper Foothills geomorphic zone. The channel has a low gradient and resembles a mixed bed alluvial channel with gravel, sand, and silt dominating the substrate. The channel is between 15 m and 30 m wide, with pool habitat being more abundant than rapid-riffle sections.

The riparian vegetation is dominated by grass, with clutches of reeds, sedges, and shrubs (e.g. *Lavandula* spp.) also present. The upper riparian zone and flood benches are highly disturbed, for example, by the presence of foot paths, gravel roads, and solid waste disposal. The riverbed is highly modified due to several instream structures (upstream weir and low water crossing; downstream road bridge) present at the site.

This section of the Namahadi River flows through urban areas with the Thaba Bosiu residential area on the escarpment above the right river bank, and the Ha-Sethunya residential area on the left bank. The density of people in this part of the catchment is markedly higher than at the Namahadi Upper: Thaba Bosiu has a population of approximately 2 935 people (885 households) and a density of 8 06.51 people per km², whereas Ha-Sethunya has a population of 6 678 (1 977 households) and a density of 1 516.2 people per km² (StatsSA, 2011). The urban impacts on the river are visible, and include sedimentation of the riverbed, as well as flow and riverbed modification due to instream structures.

(c) Namahadi Lower

Namahadi Lower is situated in the lower reaches of the river, approximately 4.6 km downstream of the Namahadi Middle site and 5.9 km upstream of where the Mphukojwane River joins the Namahadi. This section of the river falls within the Lower Foothills geomorphic zone. The river resembles a mixed bed alluvial channel with gravel, sand, and silt dominating the substrate. The channel is between 12 m and 45 m wide, with pools dominating the aquatic habitat. The riverbed is highly modified due to several instream structures (upstream weir and low water crossing; downstream road bridge) present at the site (see Appendix A: Plate 2, Photos 2-3).

At this point, the Namahadi River flows through very densely populated areas (2 495.9 people per km²) (StatsSA, 2011), with widespread urban impacts. The river banks are highly impacted by overgrazing, trampling by cattle and small stock, sand mining, and brick making.

5.2.3.2 Metsi-Matso

The perennial Metsi-Matso River originates at an elevation of more than 2 000 m.a.s.l. The river is considered to be largely natural from its origin to upstream of its confluence with the Namahadi River, despite the presence of a relatively large dam in its upper reaches. The Metsi-Matso Dam, built on the Metsi-Matso River approximately 6 983 m from the river's source in the Drakensberg Mountains, contributes to the water provisioning of the QwaQwa area through the Metsi-Matso Dam scheme that supplies drinking water the south-eastern areas of QwaQwa (Free State CoGTA and Chell Engineering, 2018). The Metsi-Matso Dam falls within the Thaba Bosiu and Dinkweng Tribal Councils (DWAF, 2007). Extensive wetlands occur at the dam's upstream end (DWAF, 2007).

(a) Metsi-Matso Upper

The Metsi-Matso Upper site is situated in the river reach immediately downstream of the Metsi-Matso Dam (see Appendix A: Plate 3). The geomorphological characteristics of the river channel represent the Upper Foothills zone. The channel slope is moderately steep, with a mixed bedrock-cobble bed. The flow regime of the river is, however, severely impacted by the presence of the dam (see Appendix A: Plate 3, Photo 1). During the site visit in October 2020, there was no surface flow, which left sections of the river dry (see Appendix A: Plate 3, Photo 5). A number of pools, varying in size, were present in the river channel. When the river flows, these pools should be connected by equally long rapid-riffle sections.

The riparian vegetation is dominated by grass, with trees and shrubs occurring along the river channel. The exotic and highly invasive black wattle (*A. mearnsii*) is present at the dam wall, as well as along the left bank. The vegetation along the edge of dam, upstream of the site, comprises predominantly of wetland species, as well as *Helichrysum* spp.

(b) Metsi-Matso Middle

Metsi-Matso Middle is situated in the middle reaches, approximately 5 km upstream of where the Namahadi River joins the Metsi-Matso River. The river channel resembles the Upper Foothills

geomorphic zone, with a moderately steep slope and a mixed bedrock-cobble substrate (see Appendix A: Plate 3). This section of the river consists of pools connected by rapid-riffle sections. At the site, the pool is mainly underlain by silt, while the rapid-riffle section is underlain by bedrock, boulders, and cobbles (see Appendix A: Plate 3, Photo 4).

The riparian vegetation in the river reach comprises predominantly of grass and sedges. At the site, the pool is fringed by sedges (*Cyperus* and *Juncus* spp.) and bulrushes (*Typha capensis*). Downstream of this pool, a rapid-riffle connects it to another pool with a steep right bank covered by various species of trees.

Metsi-Matso Middle is situated on the outskirts of the Metsi-Matso residential area (see Appendix A: Plate 3, Photo 1). Upstream of the site the main impacts on the river include flow and bed modification due to instream weirs and road crossings, as well as faecal contamination due to the watering of stock (Motholo, 2014).

5.2.3.3 Mphukojwane River

The Mphukojwane is a perennial stream that originates on an escarpment (at an elevation of more than 1 900 m) south-west of Phuthaditjhaba. The stream drops over the escarpment to flow northwards for approximately 13 km to join the Metsi-Matso River from the south. Although pristine at its origin, the Mphukojwane River is highly impacted by urban runoff and pollution in the middle and lower reaches (Motholo, 2014).

(a) Mphukojwane Upper

Mphukojwane Upper is situated approximately 1 km downstream from the river's source on the escarpment (see Appendix A: Plate 4). The stream channel is steep, with clear, fast-flowing water. The channel is narrow (1 m to 3 m wide), and resembles a Mountain Stream geomorphic zone type. The streambed is dominated by bedrock and boulders, with step pools and cascades occurring. The riparian zones are dominated by grass, with dense woody vegetation growing next to the stream.

Even though the site appears to be relatively pristine, serious disturbance and erosion occur within the riparian zone approximately 300 m downstream of the site. Here the river flows past the Tseki residential area (right bank) and the upper section of the Poelong residential area (left bank). The density in these residential areas is high at 2 214.4 people per km² in Tseki, and 1 269.8 people per km² in Poelong (StatsSA, 2011).

(b) Mphukojwane Middle

Mphukojwane Middle is situated approximately 13.2 km downstream of the upstream site. Here the river flows through densely populated areas. Phamong (on the right bank of the river) has a population of 8 010 people (2 288 households) and a density of 2 792.1 people per km², and Makhalaneng (on the left bank) has a population of 5 085 people (1 552 households) and a density of 1 542.6 people per km² (StatsSA, 2011).

At Mphukojwane Middle, the stream flows through the Eastern Escarpment Mountain 15.03 Level 2 ecoregion. In this section, the stream channel has a lower gradient (than the upstream site) and resembles an alluvial channel with sand and gravel dominating the streambed. At the reach level, pools are more abundant than rapid-riffle series.

The riparian zone is highly disturbed, with deep erosion dongas occurring on the left bank of the river and encroachment by the highly invasive and exotic *A. mearnsii*. Other impacts include bed modification due to sedimentation and instream structures, flow modification due to instream structures, and pollution from urban sources.

5.2.3.4 Elands River

The Elands River has a catchment size of 1 405 km² (Republic of South Africa [RSA], 2016). Several streams originating in the mountains above QwaQwa, including the Metsi-Matso, Namahadi, Kollatshwene, and Mphukojwane rivers, join to form the Elands River. The Elands River in turn joins the Wilge River, which is the largest tributary of the Vaal River.

Biomonitoring results from 2002 to 2003 indicated that the river's ecological integrity is "fair" to "poor" (River Health Programme, 2003). These results are supported by Motholo (2014), who found the water in the river to be turbid and highly polluted. Being situated at the downstream end of the rivers that drain the QwaQwa area, the Elands River receives runoff impacted by industrial processes and return flows from water treatment plants (WTPs), as well as from general urban and rural activities. Motholo (2014) also found the aquatic macroinvertebrate community in the Elands River to be dominated by highly tolerant taxa, which reflected the poor water quality of the river.

A gauging station (C8H005) in the Elands River is situated approximately 15 km downstream of the Elands River site.

(a) Elands Upper

Elands Upper is situated 2.3 km downstream of where the Mphukojwane joins the main stem to become known as the Elands River. At this point, the river flows through the Eastern Escarpment Mountain 15.01 Level 2 ecoregion. The river channel resembles a Lower Foothills geomorphological zone with a series of runs, pools, and riffles (see Appendix A: Plate 5).

The riparian zone is highly disturbed, with eroded and unstable river banks and encroachment of alien vegetation into the lower riparian zone. Other impacts include bed modification due to algal growth and instream structures, flow modification due to instream structures, and pollution from urban sources.

(b) Elands Lower

The most downstream site on the Elands River is situated approximately 4.7 km downstream of where the Metsi-Matso and Mphukojwane rivers join to form the Elands River. The site is located at a road bridge where the R712 crosses the Elands River (see Appendix A: Plate 5).

Elands Lower is situated in the Eastern Escarpment Mountains 15.01 Level 2 ecoregion. Here, the geomorphological features of the channel resemble the Lower Foothills geomorphic zone. The channel has a lower gradient and is more slow-flowing than upstream. The riverbed is predominantly alluvial, with sand and gravel being the most abundant substrate types. The channel is markedly wider than at the upstream sites, with aquatic habitat comprising mainly of pools, separated by shorter rapid-riffle sections.

The lower riparian and marginal zones at the site are invaded by exotic trees, including the weeping willow (*Salix babylonica*), poplars (*Populus canescens*), black locust tree (*Robina pseudoacacia*), and black wattle (*A. mearnsii*). The habitat integrity (HI) of the site is also affected by instream modifications such as sedimentation and instream structures (road bridge) that influence the natural riverbed and flow regime of the river.

5.2.4 Description of sampling and analytical methods applied in the study

5.2.4.1 Physical and chemical water quality

(a) System variables

Physico-chemical parameters were measured in situ during field sampling in April 2021 and March 2022. A portable multi-parameter reader (SensoDirect 150 – Lovibond) was used for on-site measurement of pH, DO, EC, TDS, redox potential, and temperature. Prior to sampling, the portable parameter was calibrated according to the manufacturer's specifications. Samples for COD and other water quality parameters that could not be measured on-site were collected in glass bottles, protected from light to avoid photodegradation, and transported to the laboratory in ice chests. The bottles were labelled appropriately on-site. All field readings were taken in triplicates and so were all the analyses.

The COD and BOD analyses of the samples were conducted by the accredited Institute for Groundwater Studies' Laboratory Services (UFS campus).

The main anions fluoride, chloride, nitrite, bromide, nitrate, sulphate, and phosphate were analysed by ion chromatography (Dionex Integrion HPIC, Dionex/Thermo Fisher Scientific). The determination of 21 elements, dissolved in water (see Appendix C), was carried out by means of an Inductively Coupled Plasma Optical Emission Spectrometer (ICP-OES) (AVIO 200, Perkin Elmer). The DOC was determined as non-purgeable organic carbon with a TOC-5000 (Shimadzu) according to the standard DIN EN 1484:2019-04.

(b) Metals

Water samples were collected in 500-ml plastic bottles that were pre-treated with 10% nitric acid. Unfiltered samples were kept on ice during transportation to the laboratory and stored at 4°C before they were sent for analysis. Analysis was conducted by an accredited commercial laboratory (ISO/IEC 17025:2005), WaterLab (Pty) Ltd, based in Pretoria, South Africa. The samples were analysed, following standard techniques, to determine the concentrations of 66 elements, including heavy metals,

and micro and macro nutrients using an ICP-OES (Perkin Elmer, Optima 2100 DV). Analytical accuracy was determined using certified standards (De Bruyn Spectroscopic Solutions 500MUL20-50 STD2), and recoveries were within 10% of certified values.

(c) Organic micropollutants (OMPs)

The selection of target substances for OMP analysis was based on typical representatives of pharmaceuticals, personal care products, industrial chemicals, and pesticides and biocides, which are often still detected in the effluent of WWTPs or are diffusely discharged from agriculture and thus occur globally in polluted rivers. Other environmentally problematic substances such as ethinylestradiol, estradiol or beta-lactam, and tetracycline antibiotics were not included because their relatively high instability leads to their degradation during the transportation of the samples to the analytical laboratory, which takes several days.

Water samples were filtrated immediately after sampling using a 6-ml syringe with attached syringe filter (CHROMAFIL Xtra, Regenerated cellulose, 25 mm, 0.2 μ m) that was pre-rinsed five times with sample water. Thereafter, 6 ml of the sample water was filtered in a 40-ml glass vessel (rinsed five times in advance with sample water) for further use. Exactly 5 ml of the filtered sample water was then transferred into a 12-ml glass vessel (pre-flushed with sample water). After the addition of 100 μ L of internal standard mixture and intensive shaking, 5 ml of the water sample was filtered through the pre-loaded filter. The first 4 ml was discarded, while the last 1 ml was placed into a designated 2-ml high-performance liquid chromatography vial.

The analytical determination of 35 selected OMPs was carried out via an ultra-high-performance liquid chromatography system (Nexera X2, Shimadzu) with the chromatographic column Luna Omega Polar C18 (100A, 1.6 µm, 100 x 2.1 mm) and the corresponding precolumn SecurityGuard™ ULTRA Cartridge (Fully Porous Polar C18 2.1 mm) in combination with a tandem mass spectrometer (Triple Quadrupol-MS QTRAP 6500+ detector, Sciex) according to a method developed by the Institute of Water Chemistry (Technical University of Dresden, Germany). The injection volume was always 50 µL. The compound separation was performed at a constant flow rate of 0.3 ml/min using a water/acetonitrile gradient. Each eluent was supplemented with acetic acid (0.04% v/v). An analytical standard mix of the OMPs (10 µg/L) was prepared from stock solutions in water/methanol (50% v/v) and diluted for quality control standards (0.1 µg/L) for measurement. An internal standard mix (IS) of 27 isotopically labelled OMPs was prepared equally and all solutions were stored at -18°C. The calibration was carried out for all substances in the range from 10 to 2 500 ng/L. Detection of micropollutants was performed in ESI(-) or ESI(+) mode according to optimised ionisation conditions for each analyte using the following source conditions: IS -4500 V (ESI(-)) or IS +5500 V (ESI(+)), temperature 480°C, curtain gas 40 psi, collision gas medium, and Gas 1 and Gas 2 at 50 psi. The quantitative determination of masses was carried out in multiple reaction monitoring measurements.

5.2.4.2 Ecotoxicological tests

A range of ecotoxicological tests were used to evaluate the effect of river sediment and water samples on exposed organisms by quantifying the effects on their growth inhibition, reproduction, and mortality. For the river sediment, a reproduction test with the springtail *F. candida* was used in combination with metal analysis. For the river water, three aquatic organisms (each representing a respective trophic level) were used, namely *D. magna*, a primary consumer that feeds on algae, *R. subcapitata* (a primary producer), and *T. thermophila* a protozoan (decomposer). Lastly, a range of in vitro assays was used to detect estrogenic and androgenic activity (Yeast-based Estrogenic Screen [YES], Yeast-based Antiestrogenic Screen [YAES], Yeast-based Androgenic Screen [YAS], and Yeast-based Anti-androgenic Screen [YAAS] tests after ISO 19040-1; 2018), the presence of dioxins (Yeast-based Dioxin-like Screen [YDS]), the mutagenicity (Ames fluctuation test after OECD 471; ISO 11350:2012), and the genotoxicity (micronucleus assay after OECD 487; ISO 21427-2:2006) in the water samples.

(a) River sediment

River sediment samples were collected from the Namahadi, Metsi-Matso, and the Elands rivers. Approximately 5 kg of sediment were collected per site and transported to the laboratory in dark refuse bags (950 x 750 mm). The samples were dried in a Kelvinator incubator at 25°C and passed through a 2-mm sieve to remove coarse debris.

Following the ISO 11267 protocol (International Organization for Standardization, 2014), the springtail *F. candida* was exposed to 0 (control), 50%, 75%, and 100% of the collected sediment samples at 20°C for 28 days. The OECD artificial soil (Organisation for Economic Co-operation and Development [OECD], 2004) was used as the clean (negative) control. Ten live *F. candida* were exposed to 20 g of the different concentrations of the sediment (re)moistened to 50% of its water-holding capacity. The exposures were carried out in four replicates. The experimental organisms were fed 5 mg baker's yeast (Superbake Instant Yeast) per treatment per week. At the end of the exposure period, survival and reproduction were assessed by counting the surviving adults and new offspring.

Metal analysis (water extraction) of the sediment samples was conducted using inductively coupled plasma mass spectrometry. A total of 71 elements including cadmium, lead, iron, mercury, and zinc were assessed. The method used extracts from the metals present in the water phase of the soil. These metals are the ones supposedly bioavailable to the organisms that come in contact with the substrate (Voua Otomo *et al.*, 2011). After metal analysis, elements with sediment concentrations lower than 0.05 mg/kg were considered minor contributors to the potential ecotoxicological effects of the sediment and were not included in subsequent analyses.

A one-way analysis of variance, followed by Tukey's post-hoc test, was performed to compare juvenile numbers per site (GraphPad Prism version 5.00, www.graphpad.com). Whenever possible, the online median effective concentration (EC50) (AAT Bioquest, https://www.aatbio.com/tools/ec50-calculator) was used to determine the concentration of sediment (in percentage) necessary to inhibit reproduction by 50%.

(b) River water

The determination of river water toxicity was based on three bioassays using aquatic organisms representing different trophic levels.

• Algal growth inhibition test

The OECD 201 algae growth inhibition test method was used to determine the toxicity of samples on a freshwater algal species, *R. subcapitata*. The experiments had five treatments and a control, and each treatment had three replicates.

The statistical design is based on hypothesis testing (hypothesis-based no effect concentration) and regression (egression-based x% effect concentration). Measurements were taken at 24-hour intervals for 72 hours and the results were analysed statistically using ToxRat Professional 3.2 Software.

Algaltoxkit F^{TM} supplied by MicroBiotests Inc. (Belgium) was used for the experiments. The algal beads were de-immobilised according to the manufacturer's instructions. An algal density of 1×10⁶ cell/ml was prepared from the concentrated algal inoculum by measurement of the optical density of the inoculum on a spectrophotometer (Jenway 6300) at a wavelength of 670 nm. Dilution series of the samples were prepared, and each flask was inoculated with 1×10⁴ cells/ml as the test start concentration. The inoculated samples were incubated at 23°C with a sideway illumination of 10 000 Lux for 72 hours. Optical density measurements of the test cells were taken at 24-hour intervals for 72 hours. The obtained data were used to determine yield and growth inhibition of *R. subcapitata* after exposure to the water samples. Data analysis was performed using ToxRat® Professional Software 3.3 for the determination of critical concentrations.

• Daphnia magna acute mortality test

D. magna were exposed to the water samples collected for toxicity testing using the ISO 6341 method. Hatching of the ephippia was achieved according to the supplier's (Daphtoxkit F MagnaTM, MicroBiotests Inc., Belgium) instructions. The young daphnids were pre-fed two hours prior to the commencement of experiments to prevent starvation to death. Dilution series of the samples were prepared according to standard procedures. There were five treatments and a control, and each treatment had four replicates. A hatching petri dish was placed on a light table and five actively swimming neonates were transferred into each of the test wells. The multiwall plate was covered and incubated in darkness at 20°C. After 24-and 48-hour incubation, the test plates were scored to determine the number of dead or immobilised daphnids. Neonates that could not swim after gentle agitation of the liquid for 15 seconds were considered immobilised even if they could move their antennae. Experimental data were analysed using ToxRAT® Professional Software 3.3 for the determination of mortality, statistical significance, and critical concentrations.

• Protozoan toxicity test

This test measured the growth inhibition of the ciliate protozoan *T. thermophila* after 24 hours. In this period, normal growing cultures completed at least five generation cycles. The test was performed in disposable 1-cm polystyrene spectrophotometric cuvettes. The tests were performed according to the

standard operational procedure manual of the Protoxkit F[™] provided by the supplier (Microbiotests Inc., Belgium). The spectrophotometer was zero-calibrated at 440 nm using distilled water. The optical density of each test cell was measured and recorded at 440 nm. At the end of the reading at the initial time (T0), cells were incubated (in darkness) at 30°C for 24 hours. After a 24-hour incubation, the optical density readings were taken again. The data regression analysis was conducted using Microsoft Excel Analysis ToolPak.

(c) In vitro tests

• Preparation of samples

Solid phase extraction was used to concentrate the micropollutants in the water samples for transportation to Germany, where they were analysed. For the solid phase extraction, an aliquot of 1 000 ml surface water was filtered through glass fibre filters (<0.7 μ m, 90 mm, Sartorius, Germany) at a low under pressure (-200 mbar) using a vacuum manifolder. A hydrophilic-lipophilic balanced cartridge (200 mg 6 cc, Waters Oasis) was selected. Hydrophilic-lipophilic balanced cartridges were conditioned with 5 ml acetone (analytical grade, purity >99.8%, Carl Roth GmbH, Germany), 5 ml methanol (analytical grade, purity >99.8%, Carl Roth GmbH, Germany) and 5 ml ultrapure water (Millipore® filter systems). After conditioning, the filtered water samples were loaded onto the hydrophilic-lipophilic balanced cartridges with a flow rate of 4.0 to 5.5 ml/min and subsequently washed with 5 ml ultrapure water. Loaded cartridges washed with ultrapure water were gently dried under a nitrogen stream (purity 99.5%, Afrox, South Africa) and afterwards stored at -20°C until elution.

After transportation to the laboratories at Technical University of Dresden, the elution of cartridges was performed in the following way: eluted by 2 ml acetone and 2 ml methanol, completely dried under a gentle nitrogen stream, diluted using 100 µL dimethyl sulfoxide (DMSO) (purity >99.5%, Carl Roth GmbH, Germany), and stored in 1.5-ml glass vials at -20°C until testing in the enrichment factor of 10 000.

• Yeast-based reporter gene assays for endocrine activities

The following yeast-based reporter gene assays were used: YES (agonistic activity of human estrogen receptor [hER] α), YAES (antagonistic activity of hER α), YAS (agonistic activity of human androgen receptor [hAR]), YAAS (antagonistic activity of hAR), and YDS (agonistic activity of human aryl-hydrocarbon receptor [AhR]). The research group of J. Oehlmann (Department Aquatic Ecotoxicology, Goethe University Frankfurt am Main, Germany) kindly provided the applied yeast strains of recombinant *Saccharomyces cerevisiae*. Yeast-based reporter gene assays were conducted as previously described (Abbas *et al.*, 2019; Völker *et al.*, 2016; Wagner *et al.*, 2013; ISO 19040-1). For the various yeast bioassays, the following substances were used as positive control: 17ß-Estradiol (E2, Chemical Abstracts Service [CAS] 50-28-2, purity \geq 99%; Sigma-Aldrich, Germany) for estrogenic screen, 4-Hydroxytamoxifen (4-OHT; CAS 68392-35-8; purity >70%, Z-Isomer; Sigma-Aldrich, Germany) for anti-estrogenic screen, testosterone (T, CAS 58-22-0, purity >99%; Merck, Darmstadt, Germany) for androgenic screen, flutamide (Flu, CAS 13311-84-7; Sigma-Aldrich F 9397, Germany) for anti-androgenic screen, and β -Naphthoflavone (β -NF, CAS 6051-87-2, purity \geq 98%; purum, Germany)

for screening dioxin-like compounds. After exposure for over 20 hours, 4-Methylumbelliferyl β-Dgalactopyranoside (CAS 6160-78-7, Carl Roth GmbH, Germany) was admixed and yeast tests were further incubated in the dark to measure enzyme activity by fluorescence after one hour as described by Völker *et al.* (2016). Hormonal equivalence (e.g. 17ß-Estradiol equivalence in ng/L) was calculated as described by previous studies (Völker *et al.*, 2016; Könemann *et al.*, 2018; Kunz *et al.*, 2017).

• Ames fluctuation test

The mutagenicity of native and extracted wastewater samples was assessed using the Ames fluctuation test (ISO 11350; OECD 471) with the *Salmonella typhimurium* strains TA 98 (detection of frameshift mutation) and TA 100 (detection of basepair substitution). The research group of J. Oehlmann (Department Aquatic Ecotoxicology, Goethe University Frankfurt am Main) kindly provided the strains used. Ames fluctuation tests were conducted as described by Magdeburg *et al.* (2014). For testing native samples or extracted samples, overnight cultures were diluted to 1 800 or 180 (TA 98) and 450 or 45 (TA 100) formazine attenuation units in the assay medium. The positive control was carried out using 4-Nitro-Phenylenediamine (CAS 99-56-9, Sigma-Aldrich, Germany) for TA 98 and nitrofurantoin (NF, CAS 67-20-9, Sigma-Aldrich, Germany) for TA 100. DMSO was used as solvent control for extracted samples.

Micronucleus in vitro (MNvit) assay

The genotoxicity of the water samples was tested with the in vitro mammalian cell micronucleus test MNvit (ISO 21427-2, OECD TG 487) (Sommaggio et al., 2018; Reifferscheid et al., 2007) was conducted with human hepatoma cells (HepG2). HepG2 cells were obtained from the Leibniz Institute Deutsche Sammlung von Mikroorganismen und Zellkulturen (German collection of microorganism and cell culture, cell no. ACC-180, Braunschweig, Germany) and cultured in RPMI1640 medium (with Iglutamine; ROTI®Cell RPMI-1640 CELLPURE®, Carl Roth GmbH, Germany) with 10% new-born calf serum (heat activated; Sigma-Aldrich, Germany) and 1% penicillin/streptomycin (WFI solution; Biochrom, Germany) at 37°C and 5% carbon dioxide. For the MNvit test, HepG2 cells were used between passages 6 to 15 and seeded (6 x 104 cells/ml) onto adhesive microscope slides (Superfrost® UltraPlus by Menzel, Carl Roth GmbH, Germany) in four-well microtiter plates for an incubation period of 24 hours. Cells were then treated with three different concentrations of surface water and in replicates for 24 hours. The negative (solvent) control was carried out in replicates using 5 ml culture medium only (native sample testing), or with addition of 50 µL DMSO (extracted sample testing). The positive (solvent) control was carried out in replicates using 4 ml culture medium and 1 ml ethyl methanesulfonate (EMS, CAS 62-50-0, Sigma-Aldrich, Germany) at 1.75 mg EMS/ml in culture medium only (native samples), or with the addition of 50 µL DMSO (extracted samples). After a 24-hour exposure, treatment solutions were removed from each well, and 1.5% tri-natrium-citrate solution was added. Subsequently, cells were fixed by twice using an iced solution of methanol: acetic acid: formaldehyde at 37% (3: 1: 0.05, v/v/v). After drying, slides were stained twice with 5% Giemsa solution for 15 and 20 minutes respectively. The analysis of MNvit was performed microscopically and mono-, di-, tri- to tetra-, and multi-nucleated cells, as well as micronucleus per dinucleated cell, were counted. Approximately 500 cells per slide, two slides per sample and control, as well as three different concentrations per sample were counted; thus, approximately 1 000 cells per control and 3 000 cells per sample were analysed.

5.2.4.3 River health

Although more components, or index models, are generally used to assess the health, or EcoStatus, of a river, only two of these were applied in this study due to a narrower scope and limited funding.

(a) Instream and riparian habitat integrity (HI)

The assessment of the habitats available to biota is essential during any assessment of biological integrity and provides an indication of the degree to which a river has been modified from its natural state (Barbour *et al.*, 1999). The HI of the study area was assessed according to the methodology described by the DWAF (1999). This rapid HI assessment is a simplified version of the HI approach developed by Kleynhans (1996; 2007). It involves a qualitative assessment of the number and severity of anthropogenic disturbances, and the potential negative impact it may have on the river ecosystem. Disturbances include both abiotic and biotic factors, which are regarded as the primary causes of degradation of the instream and riparian habitat (Belcher *et al.*, 2018). The severity of each impact is ranked using a six-point scale between 0 (no impact) and 25 (critical impact). The assessment is ground-based and focuses on a single site within a local reach. It considers mainly impacts visible at the site, or immediately upstream or downstream of the sampling site and is therefore of a low confidence (DWAF, 1999). The HI assessment for the sites was based on observations made during site visits in February 2021, April 2021, and December 2021, together with Google Earth images and limited literature on the catchment.

The HI assessment assesses both the impacts on the instream and the riparian zones of the river. The total scores for each of these components are then calculated and used to place the HI in a specific HI category (see Table 5.3).

Category	Description	Score of total (%)
A	Unmodified, natural.	90-100
В	Largely natural with few modifications. A small change in natural	80-90
	habitats and biota may have taken place but the ecosystem	
	functions are essentially unchanged.	
С	Moderately modified. A loss and change of natural habitat and	60-79
	biota have occurred but the basic ecosystem functions are still	
	predominantly unchanged.	
D	Largely modified. Large loss of natural habitat; biota and basic	40-59
	ecosystem functions has occurred.	
E	The loss of natural habitat; biota and basic ecosystem functions	20-39
	are extensive.	
F	Modifications have reached a critical level and the lotic system	0-19
	has been modified completely with an almost complete loss of	
	natural habitat and biota. In worst instances, basic ecosystem	
	functions have been destroyed and changes are irreversible.	

Table 5.3. River health categories

Source: DWAF (1999)

(b) Fish community health

• Instream fish habitat assessment

An assessment of fish habitat potential was conducted at each of the sampling sites as indicated by the FRAI method (Kleynhans, 2007). Fish habitat potential refers to the potential of a specific habitat to provide suitable conditions for fish species to occur there (Kleynhans, 2007). This assessment considers the diversity of velocity-depth classes present (e.g. slow deep, slow shallow, fast deep, and fast shallow), as well as the various types of fish cover available in each of the velocity-depth classes (e.g. substrate, marginal vegetation, water column, etc.). In addition, qualitative and quantitative information was gathered on habitat condition, flow conditions, water depths, substrate, etc., which are reported for each of the sampling sites.

• Fish community assessment

The FRAI developed by Kleynhans (2007) was used to determine the present ecological status of the fish communities for river reaches represented by the selected sampling sites. The FRAI is based on the environmental intolerances and preferences of the reference fish assemblage believed to have occurred in a reach under natural conditions with the response of the constituent species of the assemblage to particular groups of environmental determinants or drivers (Kleynhans, 2007).

Reference conditions and data availability

As part of the FRAI, a reference fish assemblage is determined for each sampling site or river reach to be assessed. This list of reference species is generally based on existing or historical fish records and reports, information from other sites in similar ecoregions, and expert knowledge and experience. This information is also used to determine the frequency of occurrence for each reference species under natural conditions. Reference conditions broadly refer to "expectations on the state of aquatic biological communities in the absence of human disturbance and pollution" (Kleynhans, 2007). In the context of this report, it refers specifically to the fish species present in a particular river reach and their frequency of occurrence under reference habitat conditions (Kleynhans *et al.*, 2007).

Fish sampling

Fish sampling at the study sites was done by electro-narcosis (conducting an electric current into the water, which immobilises the fish momentarily). This method is advised by the River Health Programme (see Kleynhans, 2003) for sampling fast-shallow, fast-deep, and slow-shallow habitats; e.g. riffles, rapids, glides, runs, and shallow pools. A SAMUS 725G backpack-electroshocker, powered by a 12-volt battery, was used to apply the electric current to the water with one sampler with a dip-net to collect the stunned fish.

Fish sampled in the different biotopes at the sampling site were kept in separate plastic buckets until identification and measurements were completed. The distance of the river stretch sampled was recorded at each site. Sampling effort differed between the various sites, depending on the availability of fish habitat at each site. Each fish sampled was identified to species level, weighed (g), and the fork

length (mm) noted. Observations were also made on the general health of the collected fish and anomalies were noted.

In addition to the ichthyological data, in situ information was also collected for various physico-chemical variables, including temperature, pH, oxygen saturation, EC, and turbidity. The diversity of fish habitat available at each of the sampling sites was assessed and noted and information on the state of local catchment conditions was assessed as guided by the HI Index (Kleynhans, 1996). Additional information on fish microhabitat that was collected for each biotope sampled included a minimum of 10 random water depths (cm), substrate, fish cover (available at that point), and surface flow (when applicable).

5.2.5 Identification of potential risks to human health and the delivery of ecosystem services

The results from each of the components that formed part of the ecological risk assessment (see Table 5.1) were used to identify potential risks for human and environmental health for each of the sampling sites studied. These risks are presented in Table 5.7. The results from the various components were evaluated according to the recommended thresholds for each specific component.

5.2.5.1 Physical and chemical water quality

(a) System variables

The Target Water Quality Ranges (TWQRs) used by the DWS (DWAF, 1996a; 1996b) were used to evaluate and interpret the general quality of the water samples. If the concentration for a chemical-specific parameter, e.g. EC, was below the TWQR value, it indicates that the parameter was not likely to affect aquatic life or domestic use negatively, and was coded with the colour green (no immediate concern). However, when a value exceeded the TWQR threshold for aquatic ecosystems, but not for domestic use, it was coded yellow (slight concern). In cases where the value exceeded the threshold for domestic use, it was considered a human risk and coded red (serious concern).

(b) Metals

The TWQRs for selected elements were used to evaluate the metal content of the water samples (see DWAF, 1996a; 1996b). If the metal content in the sample was below the TWQR values indicated for aquatic ecosystems (and therefore also domestic use), it was not considered a risk to either aquatic or human life and coded green (no immediate concern). However, when the content was at a concentration equal to TWQR but below the Acute Effect Value, it implied that the pollutants could occasionally affect aquatic organisms and was coded yellow (slight concern). In cases where the metal content exceeded the TWQR or Acute Effect Value concentrations for aquatic ecosystems and domestic use, it was considered to pose a risk to human and environmental health and coded red (serious concern).

(c) OMPs

South Africa's water quality guidelines, published in 1996 (DWAF, 1996a; 1996b), are generally considered to be outdated and do not provide criteria for emerging pollutants and toxic substances that may be present in water bodies. Although the guidelines for domestic use have more recently been refined into the South African National Standard 241 (SANS 241, 2015) for drinking water, it is currently being revised again (Kruger *et al.*, 2022). TWQR limits do not yet exist for most OMPs. For this reason, Health-oriented Guidance Values (HGVs), which were developed by the German Federal Environment Agency, were used for an initial assessment of the trace substance levels detected. The basis of this concept is the general precautionary value of 100 ng/L for previously unevaluated or partially evaluated substances in drinking water, which is derived from studies by Dieter (2014). For those substances for which more extensive toxicological data are available, the HGVs have been updated; in the case of the investigated active pharmaceutical ingredients carbamazepine ($0.3 \mu g/L$), diclofenac ($0.3 \mu g/L$), and ibuprofen ($1 \mu g/L$), an upward adjustment was made.

When concentrations were below the limit of quantification (LOQ), which is generally considered as 10 ng/L, it was coded green, when the concentration was higher than the LOQ but below the HGV, it was coded yellow, and when the concentration exceeded the HGV, it was coded red.

5.2.5.2 Ecotoxicological tests

(a) River sediment

Due to the lack of river sediment quality guidelines in South Africa, the National Action List can provide sediment metal concentration thresholds that are acceptable for disposal in marine environments.

The National Action List can be accessed online at <u>https://www.dffe.gov.za/sites/default/files/docs/</u> marinedisposal actionlist technicalreport.pdf. The revised National Action List was approved and standardised by the Department of Environmental Affairs and Tourism. In this National Action List, metals are categorised as Annex I metals (cadmium and mercury) and Annex II metals (arsenic, chromium, copper, lead, nickel, and zinc), based on their toxicity and their concentrations in sediments. Based on their concentrations in the sediment, the metals of interest are classified as Level I (or action level) and Level II (or prohibition level) (see Table 5.4). The sediment quality is judged based on three concentration ranges; i.e. concentrations that fall below the action level, concentrations that fall between the action level and the prohibition level, and concentrations that exceed the prohibition level. These guidelines consider the sediment moderately contaminated if metal concentrations fall between the action level and the prohibition level. The sediment is considered highly contaminated if the concentrations of the metals exceed the prohibition level (see Table 5.4). It is stated in the document that the presence of contaminants in the sediment does not imply that adverse biological effects are exerted on aquatic ecosystems. In this study, the pollutant class was determined based on the total metal concentrations of arsenic, cadmium, chromium, lead, mercury, nickel, and zinc. The metal concentrations in the water phase of the sediments were useful to help flag those metals that might have biological effects on sediment-dwelling organisms.

 Table 5.4. Level I and Level II acceptable concentration ranges and thresholds used for the

 screening of dredged sediment proposed for marine disposal, in mg/kg dry weight, in South

Metal	Action level (mg/kg)	Prohibition level (mg/kg)
Cd	1.5-10.0	>10.0
Hg	0.5-5.0	>5.0
or for a combined level of these metals	1.0-5.0	>5.0
As	30-150	>150
Cr	50-500	>500
Cu	50-500	>500
Pb	100-500	>500
Ni	50-500	>500
Zn	150-750	>750
or a combined level of these substances	50-500	>500

Africa

(b) River water

To determine the toxicity of the water samples, the endpoints of the three tests were measured with regard to growth inhibition / yield and mortality. Toxicities and lethal concentration / effective concentration values are inversely related; toxicity units (TUs) are thus used to describe concentration-based toxicity measurements. Acute TU is used to express the acute toxicity of concentration-based toxicities. The different classes of toxicity and toxicity values of the samples tested are presented in Tables 5.5 and 5.6. Toxicity values are usually calculated by dividing 100 by the LC50 values of the bioassays. Class weight scores would be evaluated by the allocation of a test score for the effect results of each test of the battery using Equation 1:

Σ all test scores / n

Equation 1

Where: n = number of tests performed and % class score were calculated using Equation 2:

% Class weight score = Class score / maximum class weight score × 100..... Equation 2

A sample is considered "highly acutely toxic" if the acute TU values, for at least one of the tests, fall between 10 and 100 TU as presented in hazard classification and class weight tables.

TU		Symbol	
PE<20%	Class I	No acute hazard	0
20% <pe<50< td=""><td>Class II</td><td>Slight acute hazard</td><td>8</td></pe<50<>	Class II	Slight acute hazard	8
50 <pe<100< td=""><td>Class III</td><td>Acute hazard</td><td>8</td></pe<100<>	Class III	Acute hazard	8
PE=100 in at least one test	Class IV	High acute hazard	** *
PE=100 in all tests	Class V	Very high acute hazard	

 Table 5.5. Hazard classification system for natural waters

Source: Adapted from Persoone et al. (2003); Kaza et al. (2007); Szklarek et al. (2021)
No significant toxic effect	Score 0
Significant toxic effect but <lc (≤1="" ec50="" td="" tu)<=""><td>Score 1</td></lc>	Score 1
1-10 TU	Score 2
10-100 TU	Score 3
>100 TU	Score 4

Table 5.6. Calculation of the class weight scores

Source: Adapted from Persoone et al. (2003); Kaza et al. (2007); Szklarek et al. (2021)

(c) In vitro tests

The limit of quantification (LOQ) was calculated based on the variability of the corrected fluorescence values of all negative controls. The LOQ was derived by inserting the mean response of all negative control values plus (YES, YAS, YDS) respectively minus (YAES, YAAS) three-fold the standard deviation of the mean response of all negative control values in the inverse of the four-parametric logistic function of the respective reference compound. The derived LOQ value was also divided by the final enrichment factor of 20.834. All calculations were performed with the statistical software R (version 4.2.1, 2022-06-23).

For a better overview of the results, a green-yellow-red colour scheme was applied to the results. Equivalence concentration values below the LOQ were assessed as low (green), equivalence concentration values above LOQ but below three times the LOQ, were assessed as moderate (yellow), and equivalence concentration values above three times the LOQ were assessed as high (red).

5.2.5.3 River health

Both the HI Index and FRAI were used to assess the present state of the aquatic habitat and fish communities at the study sites to calculate a total score that is related to an ecological or river health category (see Table 5.3). These categories range from A, which indicates an unmodified or natural condition, to F, which indicates a critical situation where ecosystems have been transformed to the extent that the natural habitat and biota have been almost completely lost and basic ecosystem functions are destroyed (DWAF, 1999).

For the purposes of this study, Category C (moderately modified) was coded as yellow (slight concern), Category D (largely modified) was coded orange (concern), Category E (seriously modified) was coded red (serious concern), and Category F (critically modified) was coded purple (grave concern).

5.3 RESULTS AND DISCUSSION

This section provides a brief summary of the main findings for each of the ecological components. A table with numerical results for the various sampling sites is presented in Appendix C.

5.3.1 Physical and chemical water quality

5.3.1.1 System variables

The values for all eight parameters measured varied across the sampling sites. The values were highly significant for TDS, redox potential, and COD, in that order. The pH values ranged between 7.42 and 10, and the 75th centile value was 9.7. The implication of this result is that the water at the sites was generally high in salts and other contaminants. The sites had also hard water due to the alkalinity values observed.

A significant positive relationship was observed between COD and temperature. This means that warmer freshwater systems will have significantly higher COD. Other strong, but not statistically significant, relationships were revealed in the study. These include COD and TDS (0.72), SAC₂₅₄ and DOC (0.95), DO and EC, as well as a negative correlation between temperature and redox potential. This confirms the importance of temperature as a key driver of freshwater ecosystem dynamics and health.

5.3.1.2 Metals

The concentrations of three of the eight metals included in the assessment exceeded the TWQR limits for domestic use. Aluminium and iron exceeded the limit at both sites in the Mphukojwane, as well as in the lower Namahadi River, while manganese and iron exceeded the limit in the Elands River (see Table 5.7). Aluminium exceeded the TWQR for aquatic systems in the Elands River and the middle Namahadi River, whereas zinc exceeded these limits at all the sampling sites. Arsenic, calcium, and magnesium concentrations were below the TWQRs for aquatic ecosystems and domestic use.

Potential sources of the high metal concentrations include natural sources (e.g. iron and copper could occur naturally in the catchment), agriculture, industrial waste, and WWTP effluent (Galvin, 1996; DWAF, 1996a). Of particular concern is the poor performance of the WWTPs in the catchment that release improperly treated sewage and industrial effluent into the streams (DWS, 2022b). It has been reported that effluent from the Phuthaditjhaba WWTP contained unacceptably high concentrations of magnesium, iron, manganese, zinc, arsenic, cobalt, and copper (Moloi *et al.*, 2020). See Appendix B for potential sources and risks associated with the metal elements recorded in the catchment.

5.3.1.3 OMPs

Fifteen of the 35 OMPs tested for were detected in the water of the catchment at concentrations above the level of quantification (10 ng/L). Of these, Bisphenol-4 (BP-4), caffeine, and 8-OH-efavirenz had the highest concentrations with maximum values of 262, 217, and 195 ng/L respectively. However, concentrations for the majority of the OMPs tested for were in the low to medium range, even at sites in the Mphukojwane and Elands rivers where they were clearly detected. The two sites on the Elands River (lower and upper), together with the Mphukojwane Middle site, had the highest concentrations of OMPs. At all three these sites, the concentrations for BP-4, acetochlor, efavirenz, and 8-OH-efavirenz exceeded the HGV. Caffeine has no relevance to health, but serves as a tracer of human influences.

In the context of the investigations in QwaQwa, increasing urban contaminations in the flow direction of the rivers were clearly indicated by increasing caffeine concentrations. In addition, more basic parameters such as chloride, EC, DOC, or SAC₂₅₄ correlate well with the occurrence of OMPs. It was clear from the results that the water from the sites that were exposed to lower levels of anthropogenic impacts in the upper part of the catchment was less polluted than the water at the sites in the middle and lower parts of the catchment.

5.3.2 Ecotoxicological tests

5.3.2.1 River sediment

After exposing F. candida to the sediment samples from the Metsi-Matso, Namahadi, and Elands rivers, all the examined sediments were found to be variably contaminated with chromium. The Namahadi Middle site had the highest concentration of chromium (569 mg/kg), followed by the shared lower site on the Elands River (529 mg/kg). The concentrations of chromium at both these sites were above the recommended prohibition level of 500 mg Cr/kg. The sediment from the upper reaches of both the Metsi-Matso and the Namahadi rivers and the middle reach of the Namahadi River had lower chromium concentrations (416, 400, and 472 mg/kg respectively) but were still above the recommended threshold for the action level (50 mg Cr/kg). Zinc was the second highest concentrated metal in the sediment, measured at 178 mg/kg in the joint lower site on the Elands River. This concentration was also higher than the recommended threshold for the action level (150 mg Zn/kg). Nickel was the third highest concentrated metal in the sediment with a concentration of 68 mg/kg measure at the upper site of the Metsi-Matso River, which was also the only site with a nickel concentration above the recommended threshold for the action level of 50 mg Ni/kg. The bioassay using F. candida revealed, using effective concentrations (EC50), that the upper reach of the Metsi-Matso River (EC50=24% sediment) was the most deleterious site for the reproduction of F. candida. This site was more than twice more toxic than the Namahadi Upper site (EC50= 60% sediment). Sediments from the middle reaches of both the Metsi-Matso and Namahadi rivers had the same EC50 value of 54%, denoting the same type of pressure exerted by the surrounding peri-urban centre of Phuthaditjhaba. Surprisingly, the shared lowest site (the Elands River) had an EC50 >100% and was the least toxic site in terms of the reproduction of F. candida. These findings are discussed in light of the environmental and social dynamics of the QwaQwa region.

The heavy metal contamination observed in this study originated mainly from point pollution sources associated with each river of interest. The Metsi-Matso Upper site is located below the dam. Dam impoundments are known to be a huge disturbance to riverine systems. For instance, a high concentration of chromium observed in this site could be linked to the dam impoundment. The high metal concentrations in sediment from this site were only observed in total metal content and not in bioavailable fractions measured from water extracts. This provides more insight into the historical degradation of the river that was caused by the construction of the dam. In the case of the Metsi-Matso River, the dam was a point source for the sediment degradation in the upper reaches of this river system. The metals may have sunk and persisted in the sediments over a long period, which acted as

their repository. The Namahadi Upper site is situated just after the Malemahole tributary running along the Witsieshoek Mountain Lodge. The mountain lodge would be a point pollution source for the Namahadi River since Malemahole is a repository for waste generated by the lodge, which at the time of data collection was said to lack adequate wastewater management facilities. Nevertheless, the heavy metals in the sediments of the upper reaches of the Namahadi River were only detectable in the total metal content. The Elands River was the only site with several known point sources of pollution such as the local WWTP with poor waste management; textile, brick, and cement factories; agricultural waste; as well as urban and municipal waste (Moloi *et al.*, 2020; Mosolloane *et al.*, 2018). Based on the metal analysis, the Metsi-Matso Upper site was more contaminated than the Namahadi Upper site. The middle sites in both rivers were fairly contaminated. The lower site on the Elands River was the most contaminated site. Overall, based on the total metal concentrations, the Metsi-Matso River seemed to have received more anthropogenic influence than the Namahadi River.

Regarding sediment effect on survival and reproduction, the samples did not cause death to *F. candida*. On the contrary, Vezzone *et al.* (2019) observed mortality of collembola (*F. candida*) and earthworm (*Eisenia andrei*) in dredged sediments collected from the Rodrigo de Freitas Iagoon, Brazil.

The findings from this study showed that in all sediment samples collected from the rivers of interest, F. candida was able to reproduce successfully after 28 days of exposure at 20°C. The EC50 values revealed that the sediment samples from the Metsi-Matso and Namahadi rivers did have an effect on the reproduction of *F. candida* during the exposure period. The EC50 values varied among sites in each river. The Elands River, which was the lowest common point between the Metsi-Matso and Namahadi rivers, recorded an EC50 value greater than 100% (which means that the sediment collected from this shared lowest point did not cause any hindering effect on the reproduction of *F. candida*). The Elands River runs along the industrial area of Phuthaditjhaba, where it receives industrial effluents and untreated sewage waste from the town and its suburbs. The Phuthaditihaba WWTP is located close to the Elands River. Yet, surprisingly, the EC50 results showed that sediments from the Elands River did not hamper the reproduction of F. candida, in contrast to all sediments collected from the upper and middle sites of all rivers of interest. In contrast to this finding, Cloete et al. (2017) recorded 76% inhibition of the growth of ostracods after exposure to the downstream site sediment of the Mpumalanga River in the Mpumalanga province, South Africa. The middle parts of the Metsi-Matso and Namahadi rivers recorded the same EC50 values of 54%. The common factor that caused the deterioration of sediments from these sites is the presence of human settlements that contribute to the degradation of river sediments through anthropogenic activities such as dumping of garbage in the waterways, human and livestock faecal contamination, and unmanaged sewage effluent from both municipal waste and WWTPs (Moloi et al., 2020). Based on the bioassays used in this study, the Metsi-Matso Upper site sediment sample was the most deleterious. This did not support the hypothesis that the upper site of a river would be supposedly pristine and have better quality than the downstream sites. Since the Metsi-Matso Upper site was located just below the Metsi-Matso Dam, the degradation of the sediment quality of the Metsi-Matso Upper site could be attributed to dam impoundment effects on rivers and other anthropogenic activities, such as recreation. It was previously reported that man-made structures, such

as dams and reservoirs, degrade surface water quality and quantity by altering the natural state of a river and disturbing its physiochemical parameters (pH, salinity, DO, etc.), which cause some pollutants bounded in sediments to be released (Chu *et al.*, 2016; Chen *et al.*, 2018; Liu *et al.*, 2018) and eventually threaten aquatic ecosystems. In terms of sediment quality, the sites along the Metsi-Matso River can be ranked as follows: upper site < middle site < lower site. In the Namahadi River, the sediment from the upper site (EC50 = 60%) was less degraded than the sediment from the middle site (EC50 = 54%), and almost twice as degraded as the sediment from the lower site (EC50 > 100%). In terms of sediment quality, the sites along the reproduction results, the sediment quality of both the Metsi-Matso and Namahadi rivers improved from the upper sites to the lower ones, which means that neither the Metsi-Matso nor the Namahadi River followed the normal river trend where river quality decreases from the upper to the lower reaches of the system.

5.3.2.2 River water

The water from the Namahadi Middle sampling site was considered to be highly acutely toxic and presenting a potential health hazard. Alarmingly, the water at all the sites in the upper sections of the catchment was acutely toxic, except for the upper Namahadi, where it was slightly acutely toxic.

5.3.2.3 In vitro tests

The in vitro biotests found that the water in the catchment had the potential for endocrine disruption and genotoxicity, as well as to cause dioxin-like impacts. The results for the Mphukojwane and the Elands rivers were particularly concerning. The results indicated increased estrogenic activity at the Elands Upper and Mphukojwane Middle sites, as well as highly increased activity of the aryl-hydrocarbon receptor for the Elands River sample. Elevated levels of the aryl-hydrocarbon receptor, which point towards the presence of dioxin-like substances, were indicated for the samples from the middle Mphukojwane, middle and lower Namahadi, and the middle Metsi-Matso rivers.

The increased estrogenic potential indicates pollution with estrogenic substances. Estrogen is an endocrine disruptor that originates from natural endogenous oestrogens or synthetic oestrogens. The compounds have a diverse spectrum of natural and anthropogenic origin and many of these chemicals exhibit estrogenic activity; for instance, industrial chemicals, pesticides, or human medicine, such as oral contraceptives, could be the origin. There are indications that other drugs, such as antiretrovirals used for HIV therapy, may also have an estrogenic effect (Sikora *et al.*, 2010). Dioxins, which may enter the river through industrial effluents and leachate from landfills and storm runoff, are generally highly toxic with negative impacts on human and environmental health impacts.

No anti-estrogenic (YAES), androgenic (YAS), or anti-androgenic activity could be detected at any of the sampling sites.

5.3.3 River health

In a desktop assessment of the Ecological Importance and Sensitivity Categories of quaternary catchments in South Africa, Kleynhans (2002) indicated that the Elands River is of moderate importance (Class C) and moderately sensitive (Class C). The ecological state of the Elands River was estimated to be largely natural (Kleynhans, 2000). No mention, however, was made of the smaller rivers and streams that feed into the Elands River. There is very little ecological information on these streams, and there are, for example, no official biomonitoring sites in the QwaQwa area. Due to these uncertainties with regard to species composition and distribution in the upper catchment, it was not possible to complete the Ecological Importance and Sensitivity Categories assessments for the respective sampling sites with confidence.

5.3.3.1 Instream and riparian HI

The instream habitats at the sampling sites have been moderately to seriously modified from their natural state due to anthropogenic impacts, with the HI scores varying between 35% (Category E; Metsi-Matso Upper and Namahadi Lower) and 64% (Category C; Namahadi Middle). In the Namahadi and Mphukojwane rivers, the level of degradation increased from upstream to downstream, with the upper reaches being moderately modified and the lower reaches being severely impacted. In the Metsi-Matso River, the presence of the Metsi-Matso Dam wall dominates the upper reaches of the river, which severely changed the downstream flow regime and caused increased inundation upstream of the dam wall. The dam has furthermore been stocked with exotic angling fish species such as rainbow trout (*Oncorhynchus mykiss*), which are known to be a very effective predator on the young of indigenous fish species in the tributaries of the Caledon River (Avenant and Kotze, 2004). Even though the Mphukojwane Upper site is situated very close to its source, the human impacts are clearly visible. Of special concern is the dumping of disposable diapers in the stream, both in the upper reaches and downstream. This poses a large risk in terms of the water quality, especially since the stream flows through residential areas.

The scores for the integrity of the riparian zones varied between 28% (Category E; Namahadi Lower) and 65% (Category C; Namahadi Upper). The most important impacts on the riparian zones of the streams are the encroachment of exotic vegetation (notably black wattle, willows, and poplars), extensive erosion, removal of vegetation (for cultivation and livestock grazing), and sand mining (Namahadi Lower). These impacts are of concern, since the degraded river banks could have a severe impact on the instream habitat in the case of increased rainfall and floods under a changing climate.

It is important to note that the Resource Quality Objectives (RQOs) for the instream and riparian habitats in the catchment were set at Category C (\geq 62%) (RSA, 2016). This implies that the RQOs were attained at six of the eight sampling sites.

5.3.3.2 Fish community health

The present ecological state of the fish communities in the river reaches, represented by the selected sampling sites, varied from moderately modified (Category C; Namahadi Upper and Middle) to critically

modified (Category F; Elands Upper). In the Namahadi River, there was a slight decrease in ecological integrity from upstream to downstream. Despite the large impacts on the instream and riparian habitats on the river, especially in the middle and lower reaches, the fish communities persisted. There were signs of recruitment, and the fish specimens sampled showed good external health, even at the Namahadi Lower site where the instream and riparian habitats have been seriously modified. Very few fish specimens were sampled in the Metsi-Matso River. This could be related to the loss of natural flow in the upper reaches and the loss of habitat diversity due to sedimentation in the middle reaches. However, the water of the Metsi-Matso Dam, and possibly also the upper reaches of the river, is oligotrophic with low biological productivity (Barkhuizen, 2015). It could be that the frequency of the occurrence of the expected species was lower than anticipated due to the low productivity, and not only due to external impacts on the river. The sampling sites on the Mphukojwane River showed signs of serious habitat degradation, even in the upper reaches. However, there were signs of recruitment at the Mphukojwane Middle site.

The results from the Elands Middle site, located downstream of the outlets of the WWTPs and inflow from the other tributaries, were very concerning. No fish was found in the reach, and the fish community is considered to be critically modified from natural conditions. This is three categories below the RQOs set for the river (RSA, 2016).

5.4 KEY FINDINGS AND POTENTIAL RISKS IDENTIFIED

The ecological risk assessment clearly showed that the upper reaches of the rivers, situated predominantly in rural and less densely populated areas, were less affected by anthropogenic impacts than the middle and lower reaches of the rivers that flow through urban and densely populated parts of the study area.

A clear deteriorating trend from upstream to downstream was observed for most of the ecological components included in the study:

- Water quality, in general, deteriorated from upstream to downstream, with the water at the most downstream sites on the **Elands River** being hypertrophic, acutely toxic, and having high levels of manganese and iron and a range of emergent pollutants such as dioxins, caffeine, pharmaceuticals, and herbicides.
- The concentration of faecal coliform bacteria, including *E. coli*, widely used as an indicator of faecal pollution in surface waters, increased from upstream to downstream in the catchment.
 Faecal coliform bacteria concentrations were unacceptably high in the lower Namahadi, middle Mphukojwane, and the Elands rivers, where concentrations exceed the recommended TWQR by 1 700 times.
- The cumulative effects of upstream anthropogenic impacts on the Elands River were also reflected by the degraded state of the instream and riparian habitats that were largely modified from their natural condition, as well as the absence of freshwater fish in the upper Elands River site.

It was also clear that the pattern of pollution correlated with the respective land uses in the catchment area:

- This study was the first to detect and confirm the presence of OMPs such as sulfamethoxazole, BP-4, ibuprofen, carbamazepine, atenolol, nevirapine, efavirenz (or its metabolite, 8-OHefavirenz), and the classical indicator of human impact, caffeine, in the catchment. The data showed that the **lower Mphukojwane** and the **Elands** rivers were the most affected, with the substances being present in the low to medium range.
- This study was also the first to conduct a series of in vitro biotests in the catchment. These tests
 indicated elevated estrogenic activity in the middle Mphukojwane and Elands rivers, as well
 as highly increased activity of the aryl-hydrocarbon in the Elands River, which can induce
 inflammatory responses and may lead to chronic inflammatory diseases, including asthma,
 cardiovascular diseases, and increased cancer risk upon exposure.

A series of ecotoxicological assays and tests confirmed the toxicity of the river sediment and water at some sites in the catchment:

(a) River sediment

- The ecotoxicological investigation into the metal content of sediment in the Metsi-Matso, Namahadi, and Elands rivers found that all the sediment samples were contaminated with chromium above the recommended prohibition level. The **Elands River** was the most polluted and showed high concentrations (above the recommended thresholds) of both chromium and zinc.
- Bioassays, using the sediment samples, found the **upper site of the Metsi-Matso** to be the most deleterious to the reproduction of *F. candida*, while the upper site on the Elands River was the least toxic. This was a surprising result, but could be due to the presence of upstream weirs in the lower reaches of the Namahadi that interrupt the downstream movement of sediment in the catchment. Toxicity in the Namahadi River typically increased from upstream to downstream.

(b) River water

- Five metal elements were detected at elevated concentrations in the water samples, namely aluminium, iron, manganese, copper, and zinc. The Mphukojwane, lower Namahadi, and Elands rivers were the most impacted, with aluminium and iron concentrations exceeding the TWQRs for both domestic use and aquatic ecosystems at these sites. Copper and zinc concentrations exceeded the TWQRs for aquatic ecosystems in the Mphukojwane and lower Namahadi, while manganese and iron concentrations exceeded the TWQRs for domestic use and aquatic ecosystems in the TWQRs for domestic use and aquatic ecosystems in the Upper Elands River. Zinc exceeded the TWQR for aquatic ecosystem at all sites in the catchment.
- Acute toxicity tests with *D. magna* and growth inhibition tests with *T. thermophilia*, using river water samples, found that toxicity increased from upstream to downstream, with the sites in the middle and lower reaches being more toxic than those in the upper reaches.

- Elevated levels of dioxins / dioxin-like substances were found in the **Metsi-Matso**, **Namahadi**, and **Elands** rivers. Considering the negative health impacts of dioxins on human health and aquatic biota, the high levels in the Elands River are particularly concerning.
- There is concern about the water of the **Mphukojwane River**, with the upper site classified as having **high acute toxicity** (Class IV) and the middle site as having **very high acute toxicity** (Class V).

In terms of general water quality and river health, the following were noted:

- TDS, redox potential, and COD were high at most of the sampling sites.
- A significant relationship between water temperature and COD was identified in the catchment, which indicates that an increase in water temperature could result in higher COD. This is an important consideration given the strong warming trend projected for the catchment in the immediate to the distant future.
- The strong correlations between water quality parameters such as EC, major ions like chloride and sulphate, and DOC and OMPs, imply that elevated values for the first-mentioned parameters may serve as an indication that micropollutants are present in the river water.
- The modified structure and condition of instream and riparian habitats at the sites in the upper, supposedly more "pristine" reaches, are concerning. Of particular concern is the severe flow modification in the upper Metsi-Matso due to the lack of environmental flows being released from the Metsi-Matso Dam, severe infestation of black wattle trees in the upper Namahadi, and severe erosion and solid waste disposal in the upper Mphukojwane. Also concerning is the apparent rapid expansion of unlicensed sand mining (mostly for building and brick making) in the instream and riparian zones of streams and rivers in the catchment.

Ecological risk component	ELL	ELU	МРМ	MPU	NL	NM	NU	ММ	MU
Physical and chemical water quality									
System variables and nutrients	Nitrates	Sulphates, nitrates, chlorides	EC, sulphates, nitrates, chlorides	EC, phos- phates, nitrates	EC	EC			
OMPs	BP, ibuprofen, bisphenol A (BPA), diclofenac, carbamazepine, sulfamethoxazole, atenolol, nevirapine BP-4, caffeine, acetochlor	BP, ibuprofen, diclofenac, fipronil, DEET, carbamazepine, sulfamethoxazole, atenolol BP-4, caffeine, efavirenz, 8-OH-	BP, ibuprofen, diclofenac, sulfamethoxazole, atenolol, caffeine, nevirapine BP-4, caffeine,	BP, diclofenac, DEET, caffeine,	BP, diclofenac, caffeine, sulfamethoxazole	BP, BP-4, diclofenac, caffeine, sulfamethoxazole	BP, diclofenac, DEET, caffeine	BP, diclofenac, caffeine	BP, diclofenac, caffeine, atrazine
	efavirenz, 8-OH- efavirenz	efavirenz	efavirenz						
Ecotoxicological	tests								
Metal elements	-			-	-	_	-	-	
Sediment	Cr	Cr, Zn	Cr	Cr	Cr	Cr	Cr	Cr	Cr, Zn, Ni
Water		Mn, Fe	Al, Fe	Al, Fe	Al, Fe	Al, Cu, Zn	Cu, Zn	Fe	
I oxicity class									
	Acute hazard	Acute hazard	Acute hazard	Acute hazard	Acute hazard	High acute hazard	Slight acute hazard	Acute hazard	Acute hazard
In vitro tests				•					•
Ames test (TA 98, TA 100)									
Yeast-based tests (YES, YAES, YAS, YAAS, YDS)		YES, YDS	YES		YDS	YDS		YDS	YDS
Microbial pollution									
Faecal coliform bacteria and <i>E.</i> <i>coli</i>		Significant risk	Significant risk	Significant risk	Significant risk	Significant risk	Slight risk	Significant risk	
River health									
Instream habitat		Largely modified	Largely modified	Moderately modified	Seriously modified	Moderately modified	Moderately modified	Largely modified	Seriously modified
Riparian habitat		Largely modified	Largely modified	Moderately modified	Seriously modified	Largely modified	Moderately modified	Largely modified	Largely modified
Fish community		Critically modified	Largely modified	Largely modified	Moderately modified	Moderately modified	Moderately modified	Moderately modified	

Table 5.7. Summary of the ecological risks identified in the QwaQwa study area

(Grey: No data; Green: No immediate concern; Yellow: Slight concern; Orange: Concern; Red: Serious concern; Purple: Grave concern)

CHAPTER 6: VULNERABILITY ASSESSMENT / DISASTER RISK ASSESSMENT OF QWAQWA COMMUNITIES

6.1 INTRODUCTION

South Africa is vulnerable to climate change given its water and food insecurity, with potential impacts on health, human settlements, infrastructure, and critical ecosystem services (World Bank, 2021). South Africa is ranked 92 out of 181 countries on the 2020 Notre Dame Global Adaptation Initiative Index, due to a combination of political, geographic, and social factors. This index ranks 181 countries using a score that calculates a country's vulnerability to climate change and other global challenges, as well as their readiness to improve their resilience. The higher the score, the more readily a country is making progress to improve its resilience to climate change (World Bank, 2021).

South Africa submitted its Nationally Determined Contribution to the UN Framework Convention on Climate Change in 2016 and published its Third National Communication in 2018 in support of its efforts to realise the national development goals and to increase resilience to climate change by improving mitigation and adaptation efforts.

The current global trend represents a shift from a strong emphasis on DRR and coping capacities to building household and community resilience using a systemic risk thinking approach. Resilience is a broad and dynamic concept, with no universally accepted definition of the term, which should be context specific and scalable. The Community Capitals Framework (CCF) is increasingly used by researchers to conduct community vulnerability analyses and to develop resilience and development strategies from a systems perspective (Mattos, 2015; Peters, 2016). First developed by Flora *et al.* (2004), the CCF examines seven community capitals, or assets, namely natural, financial, social, political, cultural, human, and built capital (Mattos, 2015). Jordaan (2017) added institutional capital to this list. The term "capital" can be described as "human created assets that are invested to create new resources without consuming the entire asset" (Peters, 2016). Using the CCF approach as a baseline from where communities can improve their responses to shocks is more empowering and inspiring, unlike the vulnerability assessment, which is a more negative approach to community assessment. The increase in the frequency and intensity of hazards and disasters calls for more holistic and dynamic approaches to be deal with disaster risks, including those exacerbated by climate change. It is therefore very important to build community resilience to multi-hazards using a systems thinking approach.

In line with international best practices, it is important to focus on reducing disaster risk and building community resilience, which will also support South Africa's National Development Plan 2030, the SFDRR, the Paris Agreement on Climate Change, and the Agenda 2030 for Sustainable Development.

6.2 RISK ASSESSMENT METHODOLOGY

6.2.1 Understanding risk

The United Nations Office for Disaster Risk Reduction (UNDRR, 2016) defines disaster risk assessment as "a qualitative or quantitative approach to determine the nature and extent of disaster risk by analysing potential hazards and evaluating existing conditions of exposure and vulnerability that together could harm people, property, services, livelihoods and the environment on which they depend". The disaster risk assessment process, as supported by the Disaster Management Act, No. 57 of 2002 (RSA, 2002), as amended Act No. 16 of 2015, is depicted in Figure 6.1. The disaster risk assessment process involves four stages, namely (1) identification of risk factors, (2) estimation of the level of the risk, (3) evaluation of the risk, and (4) monitoring the risk reduction strategies, as shown in Figure 6.1. Risk analysis is a sub-process of risk assessment that involves the identification of risk factors and the estimation of the level of disaster risk. For the purposes of this study, a risk analysis was conducted.





Source: RSA (2002)

6.2.2 Risk equation

Risk assessment methodology differs extensively among scholars and professionals (National Disaster Management Centre [NDMC], 2016) due to the different theoretical interpretations of disaster risk. The formula most commonly applied to assess risk is $\mathbf{R} = \mathbf{H} \times \mathbf{E} \times \mathbf{V} / \mathbf{C}$, where:

R = Risk

H = Hazard

E = Exposure V = Vulnerability C = Coping Capacity

For a comprehensive risk assessment, each component of the risk equation is assessed in detail, using various indicators to arrive at a final risk matrix, risk prioritisation, scenario mapping, and risk reduction strategies accompanied by cost-benefit analyses for the various DRR options. This study deviated slightly from the general approach. It adopted a different approach to calculating risk, guided by the CCF, which is more empowering on the premise that any vulnerable community has some available assets on which to improve and build resilience; as opposed to looking at the weaknesses of the community in terms of their vulnerabilities assessment (Mattos, 2015).

6.2.3 The Community Capitals Framework (CCF) and other relevant frameworks

The CCF is increasingly being used by researchers to conduct community vulnerability analyses and developing resilience and development strategies from a systems perspective (Mattos, 2015; Peters, 2016). As stated earlier, the term "capital" refers to an asset created by humans in order to invest and create new resources without consuming the entire asset (Peters, 2016). Although the term "community capital" is used, it could also refer to community assets, as some capitals, such as natural capital, are not often created but are available to communities as an endowment (Belle *et al.*, 2017). The various capitals are interrelated and their availability, in terms of quantity and quality, increases community resilience, while their absence, or inadequacy, increases community vulnerability.

Another relevant framework to assess vulnerability by evaluating community assets is the Community Based Resilience Analysis (CoBRA).



Figure 6.2. Community capitals used to analyse resilience Source: Peters (2016)

6.2.4 Community Based Resilience Analysis (CoBRA)

The CoBRA conceptual framework was first used to assess the impacts of community-based DRR interventions to build local resilience in drought-stricken areas in the Horn of Africa in order to establish if the humanitarian intervention led to community resilience building (see Figure 6.3; United Nations Development Programme [UNDP] Drylands Development Centre, 2014). The CoBRA identifies both contextual and universal characteristics of resilience, using mostly qualitative and process-orientated tools to identify the key building blocks of community resilience (see Figure 6.3). The framework also analyses community- and household-level characteristics of resilience, which can then be used to develop indicators for quantitative impact assessments. In this research, the outcomes of the CoBRA were used to describe the outcomes and the level of community resilience in QwaQwa.

The UNDP Drylands Development Centre's (2014) CoBRA framework is very similar to the Department for International Development's (2011) Tango Resilience Assessment Framework, which is not expanded upon in this report.



Figure 6.3. The CoBRA framework

Source: UNDP Drylands Development Centre (2014)



Figure 6.4. Steps in undertaking a CoBRA

Source: Fitzgibbon et al. (2014)

More capitals are generated when invested, but they are depleted when they are not sustainably used, which may lead to the weakening of a community's resilience (Mattos 2015; Emery and Flora, 2006). According to the CCF, risk results from an interaction among hazard, exposure, and capitals (resilience), which is expressed as follows:

$$Risk = \frac{Hazard \times Exposure}{Capitals}$$

6.3 DATA COLLECTION AND ANALYSES

Field data collection was divided into four phases. The first phase was the identification and collection of information from relevant key informants (KIs). The KIs included disaster risk management institutions, government departments, non-governmental organisations (NGOs), schools, churches, etc., as guided by the NDMC (2016) guidelines on relevant key stakeholders to include in a disaster risk management assessment. In the second phase, data were collected via household surveys in the QwaQwa area. The third phase was field observation by the researchers, which was integrated with the data collected during the first two phases. The fourth phase comprised a workshop that was conducted with key stakeholders in QwaQwa to validate the data collected during the earlier phases and to obtain more inputs, which were built into the final research report findings.

Quantitative and qualitative data were collected concurrently during the study. The quantitative data were analysed using the Statistical Package for Social Sciences, while the qualitative data were analysed using thematic analysis.

6.3.1 Data collection

6.3.1.1 Key informant (KI) interviews

KIs were identified and contact persons and their email addresses and phone numbers were obtained. A detailed interview guide, containing the questions and instructions, was prepared. The interviews were conducted during a fieldtrip to the study area by Prof. Belle, Ms Muyambo (PhD student), and a research assistant from 13 to 15 December 2021. During this field visit, rapport was established with the identified KIs to allow for follow-up, and observations were noted. Many challenges were experienced during the field visit; for example, many officials were still working from home and were therefore unavailable, heavy rain was experienced during the visit, KIs' reluctance and apprehension to share information with the researchers due to the fear of being victimised, social restrictions due to fears of contracting COVID-19, as well as the festive mood due to the proximity to the Christmas holiday season. Some KIs were unavailable due to the fact that the local elections of November 2021 brought many changes to government offices and new portfolios were still to be filled. The schools were also about to close; there was thus a challenge to meet the principals of selected schools. Prof. Belle was invited to attend the Thabo Mofutsanyane District Disaster Advisory Forum on 12 January 2022 and used this opportunity to speak and to make follow-ups with the relevant stakeholders. Despite all these efforts, the KI response rate was low.

6.3.1.2 Household surveys

QwaQwa comprises 30 wards that were targeted for the household survey in early 2022. The November 2021 local elections made it difficult to obtain the contact details of the newly elected ward councillors, who were the first target for the household survey. Due to the elections, some ward councillors were changed and the municipality and Election Commission Office were still busy updating the list of newly elected and returning ward councillors. Despite these challenges, 246 households, representing all 30 wards, were sampled. These also included the various villages in close proximity to the ecological sampling sites identified (see Section 5.2.3). Simple random and snowballing sampling were used.

6.3.1.3 Field observations

The researchers made three visits to the study area in total: the first to make contact and conduct interviews with the identified KIs, and two to conduct the household surveys. During each of these visits, field observations related to climate and water issues were made by the researchers.

6.3.1.4 Stakeholder validation workshop

A multi-stakeholder validation workshop was held on 24 November 2022 at Letsheng Lodge in QwaQwa. This workshop was attended by more than 25 participants from diverse institutions, such as different government departments, the local municipality, NGOs, civil society organisations, and traditional authorities. The aims of the workshop were to give the research team an opportunity to present their findings on the research project (the results from all three components, namely climate

and hydrological modelling, ecological risk, and disaster risk analysis, were presented), to solicit final inputs from stakeholders, and to validate the key findings.

6.3.2 Data analyses and results

6.3.2.1 Hazard analysis

The aim of this section is to present the results of the hazard analysis in the study area. Hazard analysis is part of risk analysis, as depicted in Figure 6.1. The NDMC (2016) defines hazard analysis as "outlining the nature of each hazard in terms of its defined characteristics". It includes the identification, ranking, analysis, and prioritisation of hazards.

(a) Hazard identification

The selection of hazards was guided by literature, field observation, their apparent connection with climate change, and consultation with experts. The 17 hazards that are most likely to occur in the area are presented in Table 6.1.

Hazards					
Floods	Pests	Air pollution			
Wildfires	Lightning	Land pollution			
Drought	Erosion	Strong winds			
Heat waves	Water pollution	Hail			
Animal diseases	Water scarcity	Snow			
Human diseases	Earthquakes				

(b) Hazard ranking

During hazard ranking, the identified hazards were graded according to their prevalence. The key stakeholders, who were the KIs who participated in this portion of the study, ranked the identified hazards from 1 to 5, where 1 indicates a hazard that was least common, 2 - not so common, 3 - average, 4 - common, and 5 - most common. The rating was based on their expert knowledge, experience, and perception. As shown in Figure 6.5, water scarcity topped the list with a rating of 4.4. This was followed by drought (3.1), floods (2.6), and water pollution (2.4). Wildfires received an overall rating of 1.8.

The multi-hazard analysis focused on these 17 hazards, using commonly shared indicators, probability, area covered, magnitude, frequency, intensity, and predictability. Table 6.2 presents the Hazard Index and rating for each hazard.

The study then focused on climate-related hazards. Figure 6.5 presents the four hazards that were included in the overall risk analysis.



Figure 6.5. Hazard ranking

Count	Hazard	Average score / Hazard Index	Hazard rating
1	Drought	3.7	High
2	Extreme cold	3.7	High
3	Floods	3.6	High
4	Strong winds	3.6	High
5	Water scarcity	3.5	High
6	Hail	3.5	High
7	Lightning	3.4	High
8	Frost/snow	3.4	High
9	Heatwaves	3.3	High
10	Wildfires	3.2	High
11	Erosion	2.6	Medium
12	Alien plants	0.9	Very low
13	Land pollution	0.6	Very low
14	Fog	0.6	Very low
15	Human diseases	0.5	Very low
16	Air pollution	0.4	Very low
17	Dust	0.4	Very low

Table 6.2. Hazard Index and rating

Key: Very low: 0-0.9; Low: 1-1.9; Medium: 2-2.9; High: 3-3.9; Very high: 4-5



Figure 6.6. Selected climate-related hazards scoring

The overall average score of the selected climate hazards was calculated as follows:

6.3.2.2 Exposure analysis

This section presents the results of the exposure analysis of the critical elements of the four hazards that were prioritised, namely drought, floods, wildfires, and water scarcity. These hazards were included in the risk analysis. Exposure analysis, like hazard analysis, is part of risk analysis, and is defined as the identification of elements that are open to damage, injury, or death from climate change-related events (OECD, 2012; UNDP, 2010). The analysis included the identification of critical elements open to harm from weather events and water scarcity, as guided by literature, field observation, and expert consultations. It was necessary to select elements generic to the hazards in question; hence population, structures, land, and water sources were selected for analysis. The overall exposure per element and for each hazard is presented in Table 6.3.

Most of the elements analysed showed high exposure to all the hazards, while the population and water sources showed very high exposure to water scarcity, and water sources showed very high exposure to wildfires. Only water sources showed low exposure to water scarcity, most probably because of existing water scarcity in the catchment.

Element/hazard	Drought	Floods	Wildfires	Water scarcity	Average per element
Population	3.7	3.2	2.6	4	3.38
Structures	3.1	3.3	2.6	3.6	3.15
Land	3.7	3.5	3.1	4	3.58
Water sources	3.3	2.8	1.6		2.57
Average per	3.4	3.2	3.07	3.3	Overall exposure
hazard					= 3.17

Table 6.3. Overall exposure scores and rating

Key: Very low: 0-0.9; Low: 1-1.9; Medium: 2-2.9; High: 3-3.9; Very high: 4-5

The next step was to analyse community resilience to common climate-related hazards of drought, floods, wildfires, and water scarcity using the CCF.

6.3.2.3 Resilience analysis

(a) General socio-economic and demographic characteristics of the participants

Table 6.4 summarises the distribution of the study participants according to socio-economic attributes. Socio-economic and demographic characteristics are useful in resilience studies because they provide valuable information that can be used in policy and strategic planning, as well as to guide the identification of target groups for intervention (Sharma *et al.*, 2018).

The socio-demographic summary indicated that there were more males than females in the survey, that the survey population comprised predominantly of youths (younger than 35 years), who were mostly unemployed, educated (high school level and above), single, and living in households where 75% earned an annual income of less than R10 000. This shows a potentially very volatile population, but also a population with untapped potential.

(b) The CCF

The indicators identified for the resilience analysis were based on the CCF (Flora *et al.*, 2004; Jordaan, 2017). These indicators were natural, human, social, financial, infrastructural/built, political, cultural, and institutional capitals. Figure 6.7 shows the average capital score for the six capitals that were measured, which reflects the resilience score. Social capital scored the highest (2.73), followed by natural capital (2.67), human capital (2.1), infrastructural capital (2.36), political capital (2.01), and finally financial capital (1.98).

Characteristics	Sub-characteristics	Frequency	Percentage (%)
Gender	Female	131	53.25%
	Male	115	46.75%
Age group	18-25	70	28.46%
	26-34	69	28.05%
	35-44	38	15.45%
	45-54	25	10.16%
	55-64	24	9.76%
	65+	20	8.13%
Educational level	No formal education	28	11.38%
	Primary school level	20	8.13%
	High school level	109	44.31%
	College level	40	16.28%
	University level	39	15.85%
	Postgraduate level	10	4.07%
Marital status	Single	157	63.82%
	Married	40	16.26%
	Widowed	22	8.94%
	Separated	8	3.25%
	Divorced	3	1.22%
	Living together	16	6.50%
Source of income	Unemployed	96	39.02%
	Social grant	57	23.17%
	Formally employed	42	17.07%
	Self-employed	26	10.57%
	Diverse incomes	2	0.81%
	Employed	12	4.88%
	Others	11	4.47%
Household income	No response	3	1.22%
range (rands; annual)	Under R10 000	184	74.80%
	R10 801-R100 000	42	17.07%
	R101 000-R200 000	10	4.07%
	R201 000-R400 000	7	2.85%

Table 6.4. Socio-economic characteristics of household survey participants (N = 246)



Figure 6.7. Average capital scores for the measured community capitals

To determine the importance of each community capital in influencing the resilience of the QwaQwa communities to climate change impacts, a stakeholder workshop was conducted that assigned weights to each of the capitals. Table 6.5 shows the weighted capitals and the final Resilience Index.

Capital	Capital Index (from analysis)	Average weights (from stakeholders)	Standardised weight factor	Product (weight x Capital Index)	Final Index
Natural	2.60	3.9	0.83	2.158	2.16
Human	2.39	4.1	0.88	2.103	2.10
Social	2.60	3.9	0.88	2.158	2.16
Financial	2.00	3.6	0.77	1.540	1.54
Infrastructural	2.38	4.2	0.90	2.142	2.14
Political	2.02	3.7	0.79	1.596	1.60
Total		23.4	5.0		11.7
Resilience Index = 11.7/6 = <u>1.95</u>					

Table 6.5. Weighting of capitals and calculation of Resilience Index

The natural and social capitals had the highest scores after weighting (2.16), while financial capital had the lowest score (1.54). Table 6.5 shows that the overall community capital, or Resilience Index score, was calculated by multiplying the standardised weight factor and Capital Index that were obtained from the surveys. The resultant Resilience Index score was 1.95, which indicates low resilience. This implies that communities in QwaQwa will struggle to recover from the impacts of extreme weather events and climate shocks as benchmarked by the CoBRA framework (UNDP Drylands Development Centre, 2014).

Table 6.6. Resilience Index ranking

Capital scoring	Resilience rating	Remarks
Less than 1	Very low resilience	Collapses in case of disaster
1.1-2	Low resilience	Struggles to recover
2.1-3	Medium resilience	Recovers but worse than before
3.1-4	High resilience	Bounces back
4.1-5	Very high resilience	Bounces back better or bounces forward

6.3.2.4 Calculation of the Multi-Hazard Risk Index (MHRI)

The final multi-hazard risk was calculated as follows:

$$Risk = \frac{Hazard \times Exposure}{Capitals}$$
Or:

$$RISK = \frac{(Hazard)}{(Capitals)} \times \frac{(Exposure)}{(Capitals)}$$

$$MHRI = \frac{(3.5)}{(1.9)} \times \frac{(3.2)}{(1.9)}$$

$$= 1.8 \times 1.68$$

$$= 3.09$$

Based on the MHRI, QwaQwa exhibits a high risk to the identified climate-related hazards. Considering this in light of the low Resilience Index value (see previous section), efforts should focus on reducing exposure to multiple climate change-related hazards and increasing the resilience of the community, the environment, and structures by improving the existing weak assets.

6.4 KEY FINDINGS AND DISCUSSION

6.4.1 Hazard analysis

Of the 17 identified hazards, 10 were ranked high, one as medium, and six as very low. This ranking followed the South African NDMC's guidelines of 2016. All highly ranked hazards were climate related, which indicates the importance of addressing climate change impacts and climate change-related risks. It is evident from the hazard analysis that the overall risk of the selected climate-related hazards (droughts, floods, water scarcity, and wildfires) was **high** (3.52).

6.4.2 Exposure analysis

The overall rating of exposure to these hazards was also **high** (3.2). Most of the elements analysed showed high exposure to all the hazards, while the population and land use showed very high exposure to water scarcity. Water sources showed very high exposure to wildfires.

6.4.3 Resilience analysis

The Resilience Index of QwaQwa was rated as **low** at 1.9. However, social and natural capitals were ranked the highest in the community survey, while financial capital scored the lowest. The overall Community Capital Index, based on the household surveys, was initially rated as **medium** (2.3), but dropped to a **low** score of 1.95 after the stakeholder weighting was standardised. The CCF approach followed in this study described the cup as "half full", instead of "half empty", as does traditional vulnerability analyses. This serves to empower communities to improve their livelihoods using available resources. Current literature shows an increasing use of the CCF to assess and promote the empowerment of communities (Mulema *et al.*, 2021; Suárez *et al.*, 2021; Lamm *et al.*, 2020; Kline *et al.*, 2019; Himes-Cornell *et al.*, 2018; Belle *et al.*, 2017; Mayunga, 2007). This concept is appropriate to strengthen resilience by focusing on investing in available resources to reduce disaster risk to climate change impacts, especially in marginalised areas such as QwaQwa and similar regions in Africa.

The findings are supported by the risk register of the Maluti-a-Phofung Local Municipality, following a risk assessment that was conducted by the district in 2021. The results of this study are shown in Table 6.7. All the risks ranked as **extremely high** and **high** by the MHRI were climate related. Even "basic service disruptions", ranked as number 2, were mostly linked to the prevailing water crisis, among other causes. This risk assessment did not highlight water scarcity as a major risk, which this research brought to the fore.

Risk Quantification: Maluti-A-Phofung Local Municipality						
HAZARD CATEGORY				50	τı	
SCORE		Hazard Rating	Vulnerability Rating	Capacity Rating	Relative Risk Ratin	Relative Risk Priori
Natural - Biological	Veldfire	12	12	12	12,000	extremely high
Technological	Basic Services Disruption	11	12	12	11,000	extremely high
Technological	Structural fires	12	9	12	9,000	high
Natural - Hydrom.	Severe weather	10	9	12	7,500	high
Natural - Hydrom.	Floods	10	9	12	7,500	high
Natural - Hydrom.	Drought	8	11	12	7,333	high
Technological	HAZMAT: road	9	10	13	6,923	tolerable
Technological	Illegal dumping	9	10	13	6,923	tolerable
Natural - Biological	Human diseases	11	8	13	6,769	tolerable
Technological	Road incident	10	9	14	6,429	tolerable
Technological	Social unrest/conflict	9	11	16	6,188	tolerable
Technological	Crime	11	8	15	5,867	tolerable
Technological	Major Hazardous Installations (MHI)	5	13	18	3,611	tolerable

Table 6.7. Maluti-A-Phofung Local Municipality risk register

Source: Thabo Mofutsanyana District Municipality (2021)

CHAPTER 7: DISASTER RISK REDUCTION (DRR) STRATEGY AND PLAN FOR QWAQWA

7.1 INTRODUCTION

This section presents the proposed strategies for DRR and resilience building, with reference to the literature reviewed, data gathered and analysed for the three components of the study (climate and hydrological modelling, ecological risk assessment, and the community risk analysis), and recommendations by stakeholders and study participants. A summary of the recommendations by the ecological specialists (involved in the ecological risk assessment) and from the household surveys, KI interviews, and stakeholder workshop is presented, in response to the projections and key findings from the downscaled climate and hydrological models prepared for the catchment.

The disaster risk strategies were categorised into eight groups, according to the strengths and weaknesses of each community capital, in order to improve the low resilience of the QwaQwa community to climate change-related extreme events. The eight categories are: (1) natural, (2) human, (3) social, (4) financial, (5) infrastructural, (6) political, (7) cultural, and (8) institutional capitals, based on the CCF (Mattos, 2015; Emery and Flora, 2006; Flora *et al.*, 2004).

7.2 INTEGRATED SUMMARY OF THE KEY CLIMATE-RELATED RISKS IDENTIFIED FOR QWAQWA

QwaQwa has been experiencing chronic water scarcity, due to climate variability, as well as poor service delivery and local governance, for the past two decades. This was reflected in the hazard analysis, which identified water scarcity as one of the four highest-ranking hazards; the others being droughts, floods, and wildfires – all of which are climate related. Considering the future changes in climate and river flow, as projected by the downscaled climate and hydrological models developed for QwaQwa, the overall risk of these hazards is, and will remain, high for the local communities and the water resource management sector.

QwaQwa is projected to be decidedly warmer in the immediate to distant future, especially in the lowlands where Phuthaditjhaba is situated. Mean daily maximum temperatures could become approximately 1.5°C warmer in summer and in winter by the mid-2030s, and by more than 4°C for both seasons by the 2070s. Minimum temperatures in winter are also projected to increase, although the expected increase would be slightly less than for the maximum temperatures. It is possible that the number of days with frost could decrease by more than 40% in the higher-lying parts of the catchment, and by more than 70% in the lower-lying areas. This is a concerning result, considering the important role that frost plays as a natural disturbance in the ecology of grassland ecosystems; for example, frost plays a major role in inhibiting tree recruitment in grasslands (Botha *et al.*, 2020). The higher temperatures are also expected to increase potential evaporation, which is already very high, in the catchment. The projections indicate that the mean annual evaporation in QwaQwa could increase by 7% in the near future, and by more than 20% in the distant future. These projected increases are significant, especially when considering the important role that evaporation plays in river hydrology, as

well as in water loss from open water bodies, increased irrigation demands, and the more rapid drying of soil. Increased water losses due to the projected higher evaporative demand during the winter months (due to the warmer temperatures), when rainfall and streamflow are naturally low, could increase pressure on water resources in the catchment during this critical time.

Estimating increases in water temperature from air temperature projections is complex (Dallas and Rivers-Moore, 2014). However, it has been shown with high confidence that increases in water temperatures can change the physical and chemical properties of water (Abrahams *et al.*, 2013; Dallas and Rivers-Moore, 2014; Woodward *et al.*, 2016; Wang *et al.*, 2022). For example, warmer water does not only hold less DO than colder water, but can also increase the oxygen consumption and metabolism of aquatic biota, which could result in a further decline of DO concentrations in water (Abrahams *et al.*, 2013; Woodward *et al.*, 2016). This study found that even though the COD was high at most of the sampling sites, it is highly likely to increase in the future; considering that a significant relationship between water temperature and COD was identified in the catchment.

QwaQwa is projected to experience higher rainfall during the rainy season (summer), but relatively less rainfall during the dry season (winter). Rainfall increases will, however, not occur uniformly over the catchment. The projections show that the greatest changes in mean annual rainfall will occur in the mountainous headwater areas, with these areas receiving between 30 mm and 40 mm (4%) more rain in the near future, and between 140 mm and 185 mm (20%) more in the distant future. Rainfall increases could be lower during dry years and higher during wet years. However, verification of the results indicated that the projections may have overestimated the rainfall for dry years and underestimated the rainfall for wet years. This is an important conclusion for water resource managers, since it implies that the dry years may be even drier than projected and the wet years wetter. Also important to water managers is the fact that rainfall in the catchment is naturally highly variable; not only between years, but also within years (runoff and streamflow variability is even higher). This variability is expected to increase in the future, which will increase uncertainty and unpredictability around surface water availability. Another significant result is that dry spells are projected to increase in the immediate future, while wet spells could be fewer. This impact could be more pronounced in the higher-rainfall upper parts of the catchments than in the slightly drier lowland areas. This would have severe consequences for agricultural production and water resource management in QwaQwa.

Most of the runoff in the catchment is generated on the mountain slopes, with these areas producing nearly three times more runoff, on average per year, than the lowlands. The differences between the higher- and lower-lying areas may be even higher during dry years. The highest runoff generally occurs in January (summer), while August and September (winter) have the lowest runoff. MAR in the catchment is projected to increase by approximately 7% in the highlands and 15% in the lowlands in the near future. Into the distant future, runoff could increase by 36% in the highlands, and by nearly 70% in the lowlands. Stormflows in the catchment generally occur in summer (rainfall season), when they contribute approximately 70% to runoff. Stormflows are projected to increase by between 13% and 21% annually, with the greatest changes expected for the highland areas. The percentage increase in baseflows, which are the major contributors to total runoff during the dry winter season, are projected

112

to be higher in the lowlands than in the highlands. It is important to note that baseflows appear to be overestimated by the GCMs for winter, which is the critical period when baseflows dominate streamflow. This could have important repercussions for low-flow water quality assessments, since baseflows generally contain a higher proportion of chemical contaminants. In fact, extreme low flows can be more detrimental to perennial river systems than high flows, due to factors such as the potential for streamflow interruption that causes the river to fragment into a series of isolated pools, the loss of longitudinal river processes, longer residence time of water, and the loss of dilution capacity, which could lead to increases in the concentration of pollutants and toxins in the river water (Abrahams *et al.*, 2013; Woodward *et al.*, 2016). This is especially relevant for the Mphukojwane and Elands rivers, which were found to have high concentrations of toxic constituents (e.g. metals: aluminium, iron, and manganese; OMPs: BP-4, acetochlor, efavirenz, and carbamazepine) and microbes (e.g. faecal coliform bacteria and *E. coli*). Considering the serious problems with water purification, distribution, and the monitoring of drinking water quality in QwaQwa, especially during water shortages, it is clear that the residents who receive water from these water supply systems are at high risk.

Accumulated streamflows in the streams of QwaQwa are projected to increase in the future, although the increases are not expected to occur uniformly over the catchment. The accumulated streamflows vary greatly between the 19 sub-catchments, depending on the size and rainfall of the sub-catchments. It is also important to note that streamflow increases from upstream to downstream as the runoff from the upper quinary joins that of the middle quinary, to converge downstream with the local runoff from the lower quinary. Peak discharges in the catchment generally occur in late summer (February), most often after a thunderstorm event. The projections indicate that peak discharges could increase in the immediate future (2040s), with the most downstream sub-section of the catchment (lower Elands River) being at risk of flooding in the future.

The increase in stormflows and peak discharge could help to improve the water quality of the streams by diluting and flushing out polluted water. On the other hand, these flows could dislodge soil particles, especially from steep and overgrazed slopes, which will increase erosion and sediment delivery to the rivers. The projections indicate that sediment yields in QwaQwa could increase by 15% to 25% in the near future. This is particularly concerning, given the degraded state of the catchment and riparian habitats. For example, the aquatic habitat diversity of the middle Metsi-Matso River has been significantly reduced by sedimentation, with many of the critical habitats, such as riffles, being smothered by sediment. The degradation of riparian and instream habitats by the widespread and uncontrolled sand mining also causes concern, given the projected increases in rainfall, runoff, and stormflows that could wash large volumes of sediment into the rivers.

The higher rainfall and runoff projected for the future could alleviate water scarcity in the catchment, but it is important to note that increases in the annual runoff and streamflow may not necessarily reduce water stress if the infrastructure needed to capture the additional volume of water is not available (Döll *et al.,* 2015). Other factors, such as immigration into the area, higher evaporation from the soil and water bodies, higher evapotranspiration by riparian vegetation, and greater irrigation water demand, may neutralise the expected increases in streamflow. Another concern is the widespread occurrence

of alien trees in the catchment and even in the river channel. Of particular interest is the dense clumps of black wattle trees in the upper Namahadi River. These evergreen trees are notorious for reducing water yields (Dye and Jarmain, 2004), as well as during winter, which is the dry season in QwaQwa.

It is clear from the discussion above that the population in QwaQwa is highly exposed to water scarcity, not only in terms of quantity, but also with regard to quality. This was confirmed by the exposure analysis that indicated that communities in QwaQwa have very high exposure to water scarcity, and high exposure to the other three hazards (drought, floods, and wildfires) that are also related to climate change. The resilience analysis showed that QwaQwa has low resilience to climate-related hazards. It is therefore crucially important to strengthen resilience in the study area by investing in available resources to reduce disaster risk to climate change hazards and impacts.

This study has taken a different approach to traditional vulnerability analyses by using the CCF. The CCF differs from traditional approaches by identifying a community's unique strengths, and using those to empower communities. For QwaQwa, the social and natural capitals were identified as potential strengths that could be employed to improve resilience to climate-related hazards.

7.3 REDUCING THE RISKS: RISK REDUCTION STRATEGIES

The proposed strategies are categorised into policy recommendations, recommendations for the empowerment of the local communities, and recommendations to reduce disaster risks and adapt to climate change, as presented in Tables 7.1, 7.2, and 7.3. Considering the prominence of water scarcity as a hazard in the catchment, specific strategies for reducing the risks related to water quantity, water quality, and river health are presented in Table 7.4. **However, it should be noted that these recommendations overlap and interconnect.**

7.3.1 Policy recommendations

Table 7.1 lists policy recommendations for reducing disaster risk in QwaQwa.

Risk identified	Potential impact	How the impact can be exacerbated/alleviated by climate change	Proposed risk reduction strategy
Natural capital			
Inadequate water supply for the population	forces households to use water from unsafe water sources such as rivers. This can result in diseases and health issues	More nouseholds may be forced to use water from rivers and other unsafe sources of water due to more frequent dry periods, flooding, and pollution.	Rain harvesting is a feasible, cost-effective, and sustainable strategy to increase household access to water supply during droughts or water infrastructure glitches. Nel <i>et al.</i> (2017) report a reduction of up to 69% of municipal water demand when supplementary water is used. Several considerations such as tank size, roof size, and water demand should be taken into account. The initial installation of rainwater harvesting infrastructure could be very expensive for most poor households in QwaQwa; the local authority should therefore work in partnership with the private sector and NGOs to budget for and sponsor such a project. Rainwater harvesting will also contribute to reducing flood risk, saving energy required for tap water, and reducing stormwater runoff. India has recorded several rainwater harvesting success stories (Bansal, 2021).
			Increase the supply of conventional water supply; for example, water transfer from the Sterkfontein Dam to nearby QwaQwa.
			Conserve the wetlands that can increase the recharge of groundwater.
			More groundwater feasibility studies should be conducted and more boreholes should be drilled.
			According to the National Water Act, No. 36 of 1998, South Africans are provided with free basic water services (RSA, 1998). While the legislation was intended to address inequality and other ills of the apartheid system, it was taken literally and is misused to develop a culture of non-payment and misuse of the basic water and electricity privileges. To address this, residents should adhere to the

Table 7.1. Policy recommendations for reducing disaster risk in QwaQwa

		How the impact can be	
Risk identified	Potential impact	exacerbated/alleviated	Proposed risk reduction strategy
		by climate change	
			allocated amount and pay if they exceed their allotment. This is supported by
			scholars such as Maphela and Cloete (2020).
Financial capital	1		
Lack of access to	Inadequate	Increased incidences of	The top levels of public management, namely national, provincial, and district
emergency funds	financial capacity	flooding and droughts may	municipality, should ensure that more financial and human capitals are directed to
	to meet urgent	force households to use	the local level, where the actual work of reducing disaster risk and empowering
	financial needs	their limited finances,	communities occurs.
		which may potentially	Community members should be encouraged to participate in informal savings
		leave them more	groups popularly known as "stokvels" in South Africa.
		financially vulnerable to	
		disaster risks.	
Infrastructural capi	tal		
Poor land use and	Unsafe locations/	Higher and more intense	The municipality should update its land use regulations to promote climate
zoning	occupations	rainfall may destroy	resilience and DRR. Land use records should be updated in consultation with the
		informal settlements in	traditional authority to promote collaboration and to reduce the unlawful issuance
		unsafe locations.	of land and unplanned settlements. Additionally, a plan should be drawn up and
			resources allocated for the enforcement of land use planning regulations.
Political capital			
High corruption	Mismanagement of	Higher frequency of	The top levels of public management, namely national, provincial, and district
rate	public resources	flooding and drought may	municipality, should ensure that more financial and human capitals are directed to
	Depletion of	lead to the increasing	the local level, where the actual work of reducing disaster risk and empowering
	financial resources	exclusion of needy	communities occurs.
	for service delivery	households from receiving	Due to the general lack of confidence in municipal financial management, the
		necessary assistance.	organisation should hire professionals or experts who are certified in financial
Discontentn	Discontentment	Water scarcity may result	forensics to "provide a wide range of services from complete internal control
	and violent	in more violent protests.	audits and forensic analysis to general and basic consultations" (Abdulrahman,
	protests		2019).
			The municipality should be held accountable for the mismanagement of public
			resources and must provide clear and transparent financial statements and

		How the impact can be	
Risk identified	Potential impact	exacerbated/alleviated	Proposed risk reduction strategy
		by climate change	
			reports. The South African Constitution, in Chapter 13 in section 215, emphasises the importance of "transparency, accountability and effective financial management of the economy, debt and the public sector" (RSA, 1996); however, this regulation needs to be enforced and those found guilty of fraudulent activities should receive the necessary discipline.
			The municipality should manage its finances efficiently, exercise financial discipline, and implement systems for effective and timeous revenue collection, efficient debt collection, and timeous payment of its suppliers.
			Administrators should be guided by municipal regulations in all their work and conduct, and maladministration, nepotism, and corruption should be dealt with according to the established policy.
			A whistle-blowing system should be implemented to prevent corrupt and fraudulent activities. It should be efficiently and objectively managed to avoid criminalising innocent people. To encourage people to report fraudulent activities, the protection and the preservation of anonymity and confidentiality of whistle blowers are very important.
Political interference	Mismanagement of public resources	Climate change will reduce the availability of	Politicians must stick to their business and administrators should keep to their administrative work. "Politicians are elected, public administrators appointed; the
	Misdirection of resources	resources and create more rivalry among political parties, as well as between politicians and	former make decisions, the later prepare and execute them – that is roughly how the government works and should work" (Bryer, 2021). Politicians should not interfere in the work of the municipality when they provide basic needs to the community.
		administrators.	
Cultural capital	Γ		
No coordination	Inadequate	Higher incidences of dry	A culture of violence should be discouraged through the effective implementation
between traditional	attention to	periods and flooding may	of the law. Civic education is prescribed to educate the community on peaceful
and local authority	community needs	expose the community to	service delivery protests. While protesting is a constitutional right, violence
on DRR issues	in relation to DRR	more disaster risks, which	protests that have characterised QwaQwa in the recent past should be
		will result in frustration that	discouraged.

Risk identified	Potential impact	How the impact can be exacerbated/alleviated by climate change	Proposed risk reduction strategy
		breeds violent protests and crime.	

Institutional capital				
Low involvement of	Limited resources	The projected greater	South African legislation on DRR, local governance, and water and other natural	
local institutions in	to deal with	incidences of flooding and	resources management is well developed; however, there is a need for	
DRR	disaster risks	droughts may erode the	adherence and effective implementation by political and administrative	
	Working in silos does not build synergy among DRR role players	already limited resources to deal with disaster risks and leave the community struggling to bounce back. Increased use of inappropriate strategies to deal with disaster risks.	authorities. Additionally, there should be effective checks and balances by the community and other stakeholders in order to enforce compliance and accountability. The Maluti-a-Phofung Local Municipality's Disaster Management Centre should be fully capacitated in terms of human and financial resources. QwaQwa has the advantage that both the district and local disaster management centres are in the same place. Shared responsibilities and of resources between the centres are recommended.	
			Different department should work together; for example, local government (CoGTA) and the Department of Environment should cooperate and work together in handling issues such as the desilting of the Fika-Patso Dam.	

7.3.2 Recommendations for the empowerment of the local communities

Table 7.2 lists recommendations for the empowerment of the local communities in the QwaQwa area.

Dick identified		How the impact can be	
(recilionce)	Potential impact	exacerbated/alleviated	Proposed risk reduction strategy
(resilience)		by climate change	
Natural capital			
Pollution from solid	Human and	Higher rainfall and	Informal waste pickers should be integrated into formal waste management in
waste (e.g. diapers	environmental	stormflows may wash	order to complement the municipality's solid waste management activities and
in rivers)	health hazard	more solid waste into	reduce costs. The Pune Municipality in India is one example of such an initiative
		rivers and drainage	(Gupta, 2012). Guidelines should be developed to effectively integrate
		systems and contribute to	unemployed youths to carry out the operation professionally and make waste a
		more public health issues	resource to enhance community members' livelihoods. A youthful and educated
		and flooding.	population is one of the greatest capitals that QwaQwa possesses.
Human capital			
Lack of awareness	Inadequate	The community may not	Research on communities impacted by disasters has proven the important role
and/or knowledge	preparedness to	be adequately prepared for	that local and traditional knowledge can play in reducing disaster risk and
of disaster risks and	face disaster risks	extreme weather events	strengthening resilience against the impacts of climate change (Muyambo et al.,
DRR		such as more intense	2017). Such knowledge, especially on the management of natural resources,
		rainfall, stormflows,	should be gathered, taught to communities, and documented for future reference.
		flooding, drought, and	The traditional authority, as the custodian of such knowledge, should play an
		heatwaves.	active role.
			Civic education through formal classroom learning and mass media campaigns
			should be provided to promote understanding of the importance of natural
			resources such as water, climate change, and citizens' rights and responsibilities
			and to promote the involvement of the local population in local community
			development. The UFS in QwaQwa should be more visible and active in DRR
			and resilience-building initiatives.

Table 7.2. Recommendations for local community empowerment

Lack of a skilled workforce to cope with climate change risks	Poor maintenance of infrastructure	More infrastructural damage may result from more frequent and intense extreme climate events, which are exacerbated by poor maintenance.	The government should provide continuous skills improvement programmes for its workforce to build capacity and technical, managerial, and leadership skills for more efficient service delivery. Youths who were trained to repair water pipes should be engaged by the municipality, government departments, NGOs, etc. to, for example, repair water leakages that are rampant in the area. The municipality should engage in staff exchange programmes with municipalities in other parts of the country, as well as in other countries, in order to exchange ideas and learn new skills and competencies. This will cater for the shortage of skills in the municipality, as reported in the study. Community capacity building should be promoted to empower the disadvantaged		
			skills that they can use to improve their livelihoods. The proposed waste management project – Nature-based Solutions (NbS) – will be a good entry point.		
Inability of people living with disabilities to access resources	Low capacity to prepare for imminent risks. More exposure of people living with disabilities to extreme climate hazards.	Vulnerable groups may suffer more impact from more frequent and intense climate hazards.	The promotion of community-based organisations that target vulnerable groups such as the elderly, people living with disabilities, and other groups to offer meaningful activities, social interaction, and opportunities to participate and contribute to DRR and community resilience.		
Social capital					
Lack of community participation in DRR issues	Weak social networks to respond to climate emergencies	Higher frequency and intensity of climate-related hazards will weaken social ties and community support.	Community participation should be promoted because it "is the most effective in ensuring successful DRR initiatives at the local level because communities take ownership of these initiatives and gain a better understanding of their risks" (Nkombi and Wentink, 2022). Community participation is more than just consultative participation; it entails involvement in all stages of community development and DRR initiatives. It also promotes transparency and shared responsibility in the development and implementation of DRR initiatives. An online portal such as a specially created website, Facebook page, or Twitter account can be used to complement traditional means to create an interactive platform. As an example, such an undertaking was successfully launched by the Christchurch		

			City Council in New Zealand after the 2010-2011 Canterbury earthquake (UNDP, 2020)
General moderate involvement in social networks Financial capital Lack of access to emergency funds Lack of access to savings to cope with crises Low participation in informal financial groups (e.g.	Lack of support in crises	Increasing prevalence of flooding and droughts may leave more households facing disaster risks without adequate social support. Increased incidences of flooding and droughts may force households to use their limited finances and potentially leave them more financially vulnerable to disaster risks.	 2020). The formation and growth of social networks should be promoted among different groups of people in the community. Social networks are regarded as crucial in coping with and recovering from disasters (Muyambo <i>et al.</i>, 2017) because they benefit individuals who may not have any means to cope with catastrophic events to find support from their network. Even people who may not have family will receive support from the groups they belong to. The principles and practice of <i>Ubuntu</i> should be promoted. The promotion of microfinancing to help community members to access micro credit, micro insurance, and other services to help them absorb, transfer, and reduce disaster risk. Community members should be encouraged to participate in informal savings groups popularly known as "stokvels" in South Africa.
"stokvels") Low remittance flow			
Infrastructural/built	capital		
Roads in poor condition Poor drainage condition	Flooding threat to road infrastructure Flooding due to blocked drainage	Higher and more frequent rainfall may cause more damage to the road infrastructure and worsen access to essential	The municipality should collaborate or partner with the private sector and NGOs to incorporate community members such as unemployed youths to clean or repair some infrastructure such as drainage systems and roads, which will also help them to earn a decent livelihood and prevent them from resorting to crime. The municipality and other relevant government departments should ensure
Ageing water infrastructure	Water scarcity More copper theft,	services. Higher and more frequent	adequate budget allocation for the maintenance and replacement of ageing infrastructure.
Silling of dams	interruptions, and	rainfall and increased stormflows and sediment yield may cause more	Education, better policing, and strict enforcement of the law should curb cable theft, which partly results in more power supply interruptions.

Irregular electricity	more violent	drainage blockages that	
supply	protests	will result in increased	
	•	flooding and water	
		contamination	
		Stormflows and sediment yield, especially in peak discharges, may increase the rate of damage to	
		drainage systems.	
		Prolonged droughts and sedimentation affect the quantity and quality of water in dams.	
		Poorly maintained water	
		infrastructure results in	
		pipe and reservoir bursts	
		and water leakages.	
Political capital		v	
High corruption rate	Mismanagement	Higher frequency of	Both politicians and municipal administrators should be educated and trained in
Political	of public	flooding and drought may	climate change issues, DRR, and resilience building, and their responsibilities to
interference	resources	lead to the increasing	the community should guide their actions. One cannot give what one does not
Interference		exclusion of needy	have; skills and capacity development should therefore be key. The UFS, and
		households from receiving	especially the Disaster Management Centre, could play a leading role in
		necessary assistance.	community capacity building. "Knowledge is power!"
Low youth	Lack of input from	Higher frequency of	The untapped talent and energy of the educated but unemployed youth should be
participation in	the youth in DRR	flooding and drought may	developed through income-generating projects and skills and leadership
decision making	issues, yet the	lead to more youth and	development programmes to reduce marginalisation, poverty, and frustration. The
Low gender	youth are at the	gender discrimination in	youth are important in DRR and resilience building because of many reasons;
balance in public	greater risk from	decision making;	among which is the possibility of long-term benefits to the community (Rahman,
decision making	disasters	potentially exposing them	2020).
decision making		to greater disaster risks	
		and causing increased	
	Exclusion of	dissatisfaction among the	Decision making and project designs and implementation should take into
-----------------------	--------------------	------------------------------	--
	women from DRR	excluded groups.	account gender issues.
Cultural capital			
The traditional	Inadequate	Higher incidences of dry	For it to be relevant in DRR and resilience building, the traditional authority should
authority does not	attention to	periods and flooding may	be visible in the community and acquaint itself with the climate change-related
play any role in	community needs	expose the community to	issues that affect the community. DRR issues should also be included in some of
DRR	in relation to DRR	more disaster risks due to	the community gatherings called by traditional leaders.
		the limited role played by	
		the traditional authority in	
		DRR.	
No coordination	Lack of	Higher incidences of dry	The traditional authority should make use of its representation in the municipality
between traditional	coordination in	periods and flooding may	to speak for the development of the community instead of personal advancement.
and local authorities	decision making	expose the community to	As custodians of the land, the traditional authority should promote the sustainable
on DRR issues		more disaster risks due to	use and conservation of vital ecosystems such as the wetlands
		the lack of coordination	use and conservation of vital ecosystems such as the wetlands.
		between traditional and	
		local authorities.	
Institutional capital		-	
Lack of feedback	Limited	Higher frequency of	Institutions that conduct research should return to the study area to present their
from research	knowledge of	weather events such as	findings and recommendations to the stakeholders for consideration and possible
institutions	current disaster	intense rainfall,	adoption of recommended strategies. Continuous assessment of implemented
	risks	stormflows, high	strategies should be done through follow-up research. This is why this research
		temperature increases,	team organised a final workshop with the relevant stakeholders to share and
		and dry periods may be	validate the study's findings. A similar follow-up workshop at a later stage is
		overwhelming to the	recommended.
		community if no feedback	
		on previous or ongoing	
		research is given.	

7.3.3 Recommendations to reduce disaster risks and adapt to climate change

All the abovementioned recommendations would assist in reducing disaster risks, adapting to climate change, and building community resilience. In addition, the following strategies are proposed:

Risk identified (resilience)	Potential impact	How the impact can be exacerbated/alleviated by	Proposed risk reduction strategy
Natural capital		climate change	
Natural capital Soil erosion Deforestation Loss of wetlands	Reduced soil fertility Increased chance of flooding Increased chance of flooding and drought damage Poor water quality Reduced agricultural yield and increased food insecurity	Higher and more intense rainfall may contribute to more soil erosion and contribute to poor crop yields. Increased rainfall and runoff may result in more severe flooding due to loss of wetlands. Higher evaporation and dry spells may increase drought impacts and more intense flooding due to the loss of wetlands, which will have serious impacts on the	 NbS are recommended as an alternative to engineered interventions because they can contribute to climate change mitigation, adaptation, and resilience by supplying ecosystem services to the community. NbS are less expensive, community driven, and provide a win-win opportunity. NbS have successfully been used in other settings to reduce the exposure of critical elements to climate change impacts, reduce sensitivity of climate hazards, and increase adaptive capacity (Belle, 2016; Seddon <i>et al.</i>, 2020). Some of the activities that should be carried out are: Protect the land against land degradation through the conservation of the present natural ecosystem, as well as reforestation and afforestation. Restore rivers through the removal of silt, invasive plants, and solid waste deposits. Proper wetland management (rehabilitation and restoration) will contribute to flood, drought, and wildfire reduction and provide socio-ecological benefits. Agroforestry reduces both exposure and sensitivity to weather events, as
		community and their livelihoods.	well as providing timber, fruit, fuelwood, and other income sources (Belle, 2016; Quandt <i>et al.</i> , 2017).

Table 7.3. Recommendations for reducing disaster risk and developing climate change adaptation in QwaQwa

Infrastructural capita	_		
Inadequate and aged	Flooding is a	Higher and more intense	• The municipality should take stock and assess the current infrastructure, and
water pipes	threat to ageing	rainfall may increase damage	update its database.
Boodo in noor	water pipes	to ageing water pipes and	Local authorities should plan, invest, design, and build infrastructure that can
Roads in poor condition Poor drainage condition	Flooding is a threat to road infrastructure Flooding due to blocked drainage systems	result in more disruptions to water supply. Higher and more frequent rainfall may cause more damage to road infrastructure and worsen access to essential services. Higher and more frequent rainfall and increased stormflows and sediment yield may cause more drainage blockage, which will result in increased flooding, water contamination, and dam silting. Stormflows and sediment yield, especially in peak	 Local authomatics should plan, invest, design, and build innastructure that can resist multiple climate change-related events that might occur. Climate-resilient infrastructure protects lives and saves money. Resilient infrastructure usually requires a significant amount of capital investment, but the savings in repairs and replacements are often attractive (UIm and Manav, 2021). A maintenance plan must be developed that includes day-to-day repairs of reported malfunctioning or broken-down infrastructure, drainage systems, roads, water leakages, and dam silting. The local community should take proper care of the existing infrastructure and assist in their maintenance.
		discharges, may increase the	
		rate of damage to drainage	
		systems.	
Institutional capital	1		
Moderate access to	Low capacity to	Increasing frequency of	Early warning information, even about water cuts, should be sent via mass
early warning	prepare for	weather events such as	communication channels, such as social media and cell phone networks, because
systems	imminent risks	intense rainfall and	the majority of people have access to cell phones with this basic service.
Moderate to low		stormtlows may overwhelm	
response to early		an unprepared community	
		due to limited access to early	
warning systems		warning systems.	

7.3.4 Recommendations to reduce disaster risk and build resilience against the ecological risks identified in the QwaQwa study area (whole catchment)

Table 7.4 lists strategies for reducing disaster risks related to water scarcity (quantity and quality) in QwaQwa.

		How the impact can be			
Risks identified	Potential impact	exacerbated/alleviated by	Proposed risk reduction strategy		
Pollution		cilliate change			
Elevated heavy metal concentration in river sediment and water (Al, Cr, Fe, Mn, and Zn exceeding the recommended guidelines for aquatic ecosystems and domestic uses of water)	Human health impacts Health impact on aquatic biota The hardness of the water in QwaQwa decreases the bioavailability of metals to aquatic life Fish, especially benthic feeders, can take up metals via the gut (from the sediment) and gills (from water) Human health effects from eating fish with elevated concentration of metal elements due to bio-accumulation	Higher and more intense rainfall may wash more sediment containing heavy metal elements into rivers, especially from bare and eroded areas in the upper and middle quinaries. Heavy rainfall increases the solubility of metals in the water, which increases the uptake of metals by fish through their gills. Copper toxicity to fish (e.g. <i>Clarias gariepinus</i>) has been shown to be enhanced by increasing water temperature. Increasing metabolic rate due to rising water temperature may change LC50 values for aquatic species.	 Effective water treatment to remove metal elements from raw and wastewater streams. Resume regular water quality monitoring protocols as required by law. Decentralised water treatment units at potential sources of metal elements; e.g. industries. Improve access to safe and clean water for all, especially during drought periods when people from poorer households depend on river water for domestic use. Implement measures to curb sediment delivery (linked to metal elements entering streams) to the river channel: Develop and implement catchment management strategies, in collaboration with cultural leaders, to address overgrazing in the upper reaches of the catchment. Enforce existing legislation to curb the growth in unlawful and unlicensed sand mining in the riparian zones of streams. Resume biomonitoring of rivers and streams in the catchment, including monitoring the recruitment and external health of fish communities and effectbased monitoring. Train staff from local institutions and community members to conduct regular citizen monitoring. 		

 Table 7.4. Strategies for reducing disaster risks related to water scarcity in QwaQwa

Risks identified	Potential impact	How the impact can be exacerbated/alleviated by climate change	Proposed risk reduction strategy
High concentrations of certain OMPs / emergent pollutants	Range of human and environmental health impacts (see Appendix B) Environmental impacts of chronic exposure to OMPs and pharmaceuticals still unclear	Dilution due to heavy rainfall and increased runoff leads to higher streamflow. Downstream transportation of highly soluble OMPs during high-flow events. During dry spells, low flow, or absence of flows in combination with higher evaporation could result in a higher concentration of OMPs in river pools	Although conventional WWTPs cannot eliminate all OMPs (e.g. carbamazepine and nevirapine), they can remove some pollutants (efavirenz, estriol) from raw and wastewater streams. Decentralised water treatment units at potential sources; e.g. hospitals, industries, abattoir, etc. Construct small to medium wetlands for the removal of some OMPs; e.g. pharmaceuticals. Add effect-based techniques to monitoring efforts in the catchment.
Elevated levels of dioxins / dioxin-like substances in the Metsi-Matso, Namahadi, and Elands rivers	Dioxins are highly toxic; may cause reproductive and developmental problems, interfere with immune system and hormones, and are carcinogenic Negative health impact on aquatic biota; e.g. in fish it may cause skin lesions and altered liver function Bio-accumulation in aquatic biota –	Input due to heavy rainfall and increased runoff will lead to higher streamflow. Downstream transportation of persistent substances during high-flow events. During dry spells, low flow or absence of flows in combination with higher evaporation could result in higher concentrations of dioxins in river pools.	Dioxins can be removed through effective water treatment. Activated carbon has proved very successful in removing these harmful substances.

		How the impact can be			
Risks identifiedPotential impactexacerbated/alleviated		exacerbated/alleviated by	Proposed risk reduction strategy		
		climate change			
	trophic transfer in				
	food web				
Indications of high	Range of human	Although the projected	Further research on the combination of toxic substances that lead to acute		
acute toxicity of	and environmental	increase in rainfall and	toxicity, especially in the Mphukojwane and Elands rivers.		
some sites in the	health impacts (see	runoff should lead to the			
catchment	Appendix B)	dilution of pollution, the			
		increase in dry spells may			
		worsen and concentrate the			
		toxicity of river water.			
Poor water quality	Various	Due to a significant	Improve and restore the functionality of the WTPs and WWTPs.		
in the middle and lower sections of	environmental and human health	correlation between water temperature and COD found	Resume regular water quality monitoring protocols as required by law.		
the rivers; e.g.	effects (see	in the catchment, increased	Decentralised water treatment units at potential sources of pollution; e.g.		
elevated EC,	Appendix B)	water temperature could	industries, etc.		
nitrates, etc.	,	accelerate productivity in the	Implement measures to ourb sodiment delivery (linked to metal elements		
		river; e.g. increased algal	entering streams) into the river channel:		
		growth and decomposition	entering streams/ into the river channel.		
		of organic matter.	Develop and implement catchment management strategies, in collaboration with cultural leaders, to address overgrazing in the upper reaches of the		
		Warmer water could lead to	catchment		
		lower DO concentration in	Training staff from local institutions and community members to conduct		
		the water.	regular citizen monitoring.		
		Increased health risks due			
		to an expected increase in			
		fungal growth in water			
		because of an increase in			
		water temperature.			
		Although the projected			
		increase in rainfall and			
		runoff should lead to the			

	How the impact can be		
Risks identified Potential impact	exacerbated/alleviated by	Proposed risk reduction strategy	
	climate change		
	dilution of pollution, the		
	increase in dry spells may		
	worsen and concentrate the		
	toxicity of river water.		
Faecal coliform Waterborne	Warmer and wetter	Improve access to safe and clean water for all households.	
bacteria and <i>E. coli</i> diseases	conditions have been shown	Awareness campaigns to educate communities about the dangers of using	
concentrations Aesthetic effects:	to increase the prevalence	water directly from streams and education on easy ways of purifying water	
exceeding e g bad odours	of waterborne diseases.	before use: e g boiling water before use bleach etc	
recommended	Poorer households are		
guidelines by up to Reduced	forced to use untreated	Restore riparian zones; macrophytes such as sedges and reeds improve	
1 700 times in the recreational and	water from streams during	water quality.	
lower part of the cultural uses of	drought periods, which may		
catchment streams	become more frequent in		
	the future.		
Dumping of solid Public health and	Higher frequencies of floods	Improved waste removal by the Maluti-a-Phofung Local Municipality; e.g.	
waste into river environmental threat	and more intense rainfall	regular refuse removal and the provision of refuse bags and bins in informal	
channels, including due to toxic	events will wash waste	areas.	
used diapers and chemical pollution	downstream, which can	Compaigns to raise community awareness, especially among communities	
sanitary pads	cause blockages.	close to rivers	
contamination of			
water sources due		Dedicated effort to educate school children about the negative health and	
to disposable		environmental impacts of solid waste disposal.	
diapers dumped in		Develop partnerships between the UFS, the DWS, the Maluti-a-Phofung Local	
rural and urban		Municipality, and schools to establish "clean river" clubs at local schools to	
sections of rivers		teach learners about the advantages of restoring and maintaining river health	
		(using tools such as Mini-Sass, citizen monitoring, etc.).	
Attraction of flies			
and other pest		Appoint residents as waste disposal monitors around important areas.	
species		Create jobs in waste removal and waste recycling.	

		How the impact can be			
Risks identified	Potential impact	exacerbated/alleviated by	Proposed risk reduction strategy		
	Source of microplastic pollution Potential	climate change	Ensure waste workers have protective gear (e.g. rubber boots, gloves, etc.) to remove diapers from stream courses. Liaise with companies that produce disposable diapers and sanitary pads to assist with finding solutions to the problem.		
	groundwater pollution				
	Decomposition of organic waste releases methane				
	Inhibiting other recreational, cultural, and religious uses of rivers				
Poor performance of WWTPs	WWTPs in QwaQwa are in a critical state – no data are available according to the 2022 Green Drop report. (Efficiency decreased drastically from 2013 to 2022)	High water temperatures could have a negative impact on the effectivity of WWTPs; e.g. leading to a deterioration of quality of bacterial floc, increasing the TSS of effluent, etc.	Dedicated effort and political will to improve WWTPs' operations. Train WWTP staff through collaboration between the UFS, DWS, and Maluti- a-Phofung Local Municipality. Implement continuous river monitoring systems to provide real-time data on the effectivity of water treatment processes.		
	Wastewater and untreated sewage flow directly into streams and rivers				

		How the impact can be	
Risks identified	Potential impact	exacerbated/alleviated by	Proposed risk reduction strategy
		climate change	
	Human health and		
	environmental		
	impacts (see		
	Appendix B)		
Poor delivery of wat	ter-related services	r	1
Households without	Using water from	Poorer households are	Improve access to safe and clean water for all households.
access to clean	streams for	forced to use untreated	Rainwater harvesting and drilling boreholes for rural communities.
water and	domestic use	water from streams during	
sanitation	Health-related	drought periods, which may	Awareness campaigns to educate communities about the dangers of using
	impacts due to using	become more frequent in	water directly from streams and education on easy ways of purifying water
	untreated water	the future.	before use; e.g. boiling water before use, bleach, etc.
Habitat alteration / I	argely modified instre	am habitats	
Loss of critical	Loss of species	Higher rainfall and runoff in	Implement measures to curb sediment delivery (linked to metal elements
aquatic habitat	linked to specific	upper and middle quinaries	entering streams) into the river channel:
types and diversity	habitat types; e.g.	could increase sediment	Develop and implement catchment management strategies, in collaboration
due to siltation	riffles	delivery to rivers.	with cultural leaders, to address overgrazing in upper reaches of the
	Lower primary	Higher streamflows could	catchment
	production	help to flush streams.	 Enforce existing legislation to curb the growth in unlawful and unlicensed
	' influences food		sand mining in the riparian zones of streams.
	sources for		
	macroinvertebrate,		Restore and protect riparian vegetation to reduce the flow of water in the river
	fish, and other		channel.
	aquatic species		Restore and protect wetlands in the upper and middle quinaries of the
	Negative impact of		catchment to reduce the frequency and magnitude of floods.
	fish recruitment due		
	to loss of spawning		
	habitats and		
	smothering of fish		
	eggs and larvae		
Households without access to clean water and sanitation Habitat alteration / I Loss of critical aquatic habitat types and diversity due to siltation	Using water from streams for domestic use Health-related impacts due to using untreated water argely modified instre Loss of species linked to specific habitat types; e.g. riffles Lower primary production influences food sources for macroinvertebrate, fish, and other aquatic species Negative impact of fish recruitment due to loss of spawning habitats and smothering of fish eggs and larvae	Poorer households are forced to use untreated water from streams during drought periods, which may become more frequent in the future. am habitats Higher rainfall and runoff in upper and middle quinaries could increase sediment delivery to rivers. Higher streamflows could help to flush streams.	 Improve access to safe and clean water for all households. Rainwater harvesting and drilling boreholes for rural communities. Awareness campaigns to educate communities about the dangers of using water directly from streams and education on easy ways of purifying water before use; e.g. boiling water before use, bleach, etc. Implement measures to curb sediment delivery (linked to metal elements entering streams) into the river channel: Develop and implement catchment management strategies, in collaboration with cultural leaders, to address overgrazing in upper reaches of the catchment. Enforce existing legislation to curb the growth in unlawful and unlicensed sand mining in the riparian zones of streams. Restore and protect riparian vegetation to reduce the flow of water in the rive channel. Restore and protect wetlands in the upper and middle quinaries of the catchment to reduce the frequency and magnitude of floods.

			How the impact can be				
Risks identified Po		Potential impact exacerbated/alle		eviated by	Prop	posed risk reduction strategy	
			climate change				
Flow modification due to dam walls high up in the catchment Loss of surface water co downstream of the dam changes a perennial stream Loss of seasonal flows ecological cues from the		nal processes in . nutrient cycling n movement of water connectivity he dam, which nnial stream to a ream I flows and from the river	Intensification of impacts during droughts when flow releases from the dam are less likely; e.g. loss of pool habitat due to lack of surface flow and evaporation may lead to extirpation of aquatic species. Deterioration of water quality in isolated pools remaining in the river channel. Loss of sensitive species and		Release of ecological flows from the dams in the upper sections of the Metsi-Matso and Namahadi rivers.		
		downstream of the dam wall		domination of more tolerant species.			
Destruction of ripariar habitat along the rive	n rs	downstream of the dam wallLoss of habitat for speciesassociated with riparianvegetationIncreased delivery of sedimentinto the river channelIncrease in water temperaturedue to loss of shadeLoss of stability of river banksDeterioration of water qualityTrampling of river banks due towatering of cattle and small stockat rivers		Riparian ar water energe Degraded r in increase in the river, sediment d During dry riparian are water, incre from river b	eas absorb and dissipate gy during floods. iparian zones could result d flow speed and volume as well as erosion and elivery into the channel. periods, degraded eas could result in warmer eased loss of soil moisture panks, etc.	Implement buffer zones along the streams to protect riparian zones. Develop river restoration projects in critical areas, such as the upper reaches of impacted streams in collaboration with relevant tribal authorities and, for example, Working for Wetlands. Link environmental quality to ecotourism; e.g. create incentives for lodges, especially in the upper parts of the catchment, to improve their operations (e.g. improve wastewater treatment).	
Extensive erosion and removal of vegetation due to overgrazing in catchment	d 1 the	Increased sedim the river channe	ient delivery into I	The project and intensi increase ca erosion, as delivery, ar	ted increase in the volume ty of rainfall could further atchment and channel well as runoff, sediment nd turbidity.	Community engagement with tribal authorities. Restoration of areas with serious erosion gullies.	

Risks identified	Po	otential impact	How the impact can be exacerbated/alleviated by climate change		Proposed risk reduction strategy	
Encroachment of alier vegetation into the riparian and instream habitats of rivers	ו	Black wattle tree are highly invasi are water thirsty shown to reduce rivers	es grow fast and ve. These trees and have been e streamflow in	During drou further redu volume in s	ughts black wattles can uce streamflow/water streams and rivers.	Removal of invasive trees from the river banks, especially in the upper catchments. Investigate the viability of community-based wattle removal.

CHAPTER 8: CONCLUSIONS, LESSONS LEARNED, AND RECOMMENDATIONS FOR FUTURE RESEARCH AND MANAGEMENT

8.1 INTRODUCTION

This study set out to develop an understanding of how the cumulative impacts of extreme weather events may affect the vulnerability of local communities in QwaQwa and how these impacts may be ameliorated through DRR and resilience building. To fully comprehend the extent and impact of climaterelated risks in QwaQwa, a downscaled climate and hydrological model was developed for the catchment, an ecological risk assessment was conducted at nine representative sampling sites on the Namahadi, Metsi-Matso, Mphukojwane, and Elands rivers, and a disaster risk analysis was performed. The findings from the three components were used to develop a tailor-made DRR framework that addresses the specific hazards, exposures, and vulnerabilities of local communities in the study area. The strategies that were developed specifically address the improvement of policy and political issues, local community empowerment, DRR and climate change adaptation, and resilience building in the face of climate-related hazards. The study followed a systems thinking approach to recommend strategies that are interrelated and interconnected. Recommendations were also made for further research to build on and improve this project, and for the replicability of similar studies in other communities in South Africa. The innovative approach of assessing community assets, instead of community vulnerability to extreme climate change events, stands out in this study. The addition of a downscaled climate and hydrological model and an ecological risk assessment to the conventional disaster risk assessment methodology make this study unique.

8.2 CONCLUSIONS

8.2.1 Climate and hydrological modelling

A climate and hydrological model was developed to provide appropriate and reliable downscaled information for QwaQwa.

Climate projections indicate a warming trend for QwaQwa in the immediate and distant future, especially for the lower-lying area of the catchment where Phuthaditjhaba is situated. The mean daily minimum and maximum temperatures are expected to increase by approximately 1.5°C by the mid-2030s, and by more than 4°C by the 2070s. The number of frost days is projected to decrease by more than 40% in the higher-lying parts of the catchment, and by more than 70% in the lower-lying areas. Mean annual evaporation in QwaQwa could increase by 7% in the near future, and by more than 20% in the distant future.

QwaQwa is projected to experience higher rainfall during the rainy season (summer), but relatively less rainfall during the dry season (winter). Rainfall increases will, however, not occur uniformly over the catchment. The projections show that the greatest changes in mean annual rainfall will occur in the mountainous headwater areas, which could receive 30 to 40 mm more rain (a 4% increase from the present) in the near future, and between 140 mm and 185 mm more (a 20% increase from the present)

in the distant future. Rainfall increases could be lower during dry years and higher during wet years. Dry spells are projected to increase in the immediate future, while wet spells could decrease. This impact could be more pronounced in the higher-rainfall upper parts of the catchments than in the slightly drier lowland areas.

The hydrological model indicates that the bulk of the runoff in the catchment is generated on the mountain slopes, with these areas producing nearly three times more runoff, on average per year, than the lowland areas. The highest runoff generally occurs in January (summer), while August and September (winter) have the lowest runoff. The MAR in the catchment is projected to increase by approximately 7% in the highlands and 15% in the lowlands in the near future. In the distant future, runoff could increase by 36% in the highlands and by nearly 70% in the lowlands. Stormflows in the catchment generally occur in summer (rainfall season), when they contribute approximately 70% to runoff. Stormflows are projected to increase by between 13% and 21% annually, with the greatest changes expected for the highland areas. The percentage increase in baseflows, which are the major contributors to total runoff during the dry winter season, are projected to be higher in the lowlands than in the highlands.

Accumulated streamflows are projected to increase in the future, although the increases are not expected to occur uniformly over the catchment. The accumulated streamflows vary greatly between the 19 sub-catchments, depending on the size and rainfall of the sub-catchments. Peak discharges in the catchment generally occur in late summer (February) after thunderstorm activity. The projections indicate that peak discharges could increase in the immediate future (2040s), with the most downstream sub-section of the catchment (lower Elands River) being at risk of flooding in the future.

8.2.2 Ecological risk assessment

The ecological risk assessment collected data on physical and chemical water quality, ecotoxicology, and river health. The aim of these assessments was to identify potential hazards to human and environmental health associated with extreme weather events in the study area.

The negative impacts of current anthropogenic activities on the rivers and streams were clear. The results showed that the upper reaches of the rivers, situated predominantly in the more rural and less densely populated areas, were less impacted than the middle and lower reaches, which were situated in the urban and more densely populated parts of the catchment.

Water quality in general deteriorated from upstream to downstream, with the water at the most downstream sites on the Elands River being hypertrophic, acutely toxic, and having high levels of manganese, iron, and a range of emergent pollutants such as dioxins, caffeine, pharmaceuticals, and herbicides. The presence of faecal coliform bacteria, used as an indicator of faecal pollution of surface waters, also increased from upstream to downstream in the catchment. Faecal coliform bacteria concentrations were unacceptably high in the lower Namahadi, middle Mphukojwane, and the Elands rivers, where concentrations far exceeded the recommended TWQR set by the DWS by 1 700 times.

This study was the first to detect and confirm the presence of OMPs such as sulfamethoxazole, BP-4, ibuprofen, carbamazepine, atenolol, nevirapine, efavirenz (or its metabolite 8-OH-efavirenz), and the classical indicator of human impact, caffeine, in the catchment. The data showed that the lower Mphukojwane and the Elands rivers were the most affected, with the substances being present in the low to medium range. This study was also the first to conduct a series of in vitro biotests in the catchment. These tests indicated elevated estrogenic activity in the middle Mphukojwane and Elands rivers, as well as highly increased activity of the aryl-hydrocarbon in the Elands River, which can induce inflammatory responses upon exposure and may lead to chronic inflammatory diseases, including asthma, cardiovascular diseases, and increased cancer risk.

The ecotoxicological investigation found the river sediment in the Metsi-Matso, Namahadi, and Elands rivers to be contaminated with chromium above the recommended prohibition level. The Elands River was the most polluted, with chromium and zinc concentrations exceeding the recommended thresholds. A number of metals (aluminium, iron, manganese, copper, and zinc) were detected at elevated concentrations in the water samples. The Mphukojwane, lower Namahadi, and Elands rivers were the most impacted, with concentrations of aluminium and iron exceeding the TWQRs for both domestic use and aquatic ecosystems at these sites. Acute toxicity tests found that toxicity increased from upstream to downstream, with the sites in the middle and lower reaches being more toxic than those in the upper reaches. There is concern about the water of the Mphukojwane River, with the upper site being classified as having high acute toxicity (Class IV) and the middle site as having very high acute toxicity (Class V).

Riparian and instream habitats were moderately to seriously modified from their natural state. The scores for instream HI and fish assemblages were lower than the numerical limits set by the RQOs (as published in the *Government Gazette* in 2016) for the catchment at all the sites, except for the upper and middle Namahadi reaches. Of particular concern is the severe flow modification in the upper Metsi-Matso due to the lack of environmental flows being released from the Metsi-Matso Dam, severe infestation of black wattle trees in the upper Namahadi, and severe erosion and solid waste disposal in the upper Mphukojwane. Also concerning is the apparent rapid expansion of unlicensed sand mining (mostly for building and brick making) in the instream and riparian zones of streams and rivers in the catchment.

8.2.3 Disaster risk analysis

The disaster risk analysis comprised four sub-components: hazard analysis (hazard identification, probability, frequency, and intensity); exposure analysis of key elements; vulnerability or resilience analysis, using the CCF approach; and calculation of the MHRI.

The hazard analysis identified four hazards that posed a high risk to communities in QwaQwa, namely water scarcity, drought, floods, and wildfires. All these highly ranked hazards were climate related, which indicates the importance of addressing climate change impacts and climate change-related risks.

The exposure analysis showed that the overall rating of exposure to the identified hazards was high. It highlighted the fact that communities in QwaQwa have very high exposure to water scarcity, and high exposure to the other three hazards (drought, floods, and wildfires). Most of the elements analysed showed high exposure to all the hazards, while population and land use showed very high exposure to water scarcity. Water sources showed very high exposure to wildfires.

The Resilience Index value calculated for QwaQwa was low, which indicates that QwaQwa communities would struggle to recover from the impacts of extreme weather events and climate shocks. This result emphasises the importance of strengthening resilience in the area by investing in available resources to reduce disaster risk related to climate change hazards and impacts.

The MHRI score was high, which indicates that the overall risk to the combination of climate-related hazards identified for QwaQwa was high.

This study took a different approach to traditional vulnerability analyses by using the CCF. The CCF differs from traditional approaches by identifying a community's unique strengths and using those to empower communities. For QwaQwa, social and natural capitals were identified as potential strengths that could be employed to improve resilience to climate-related hazards.

8.2.4 DRR strategies

A DRR strategy was developed for QwaQwa based on the projected climate and hydrological changes and the findings from the ecological risk assessment and disaster risk analysis. The proposed strategies focused on four areas, namely policy recommendations, recommendations for the empowerment of the local communities, recommendations for DRR and climate adaptation, and recommendations for reducing the risks related to water scarcity (water quantity and quality) and river health.

8.3 LESSONS LEARNED

In terms of the climate and hydrological modelling, the following lessons were learned:

- Land use has been shown to play a significant role in hydrological responses in QwaQwa at the local level; e.g. in streamflows, sediment yields, and especially in peak discharges. Such impacts become muted as catchment size increases. This shows clearly that conducting meaningful hydrological impact studies at quaternary, and even at quinary spatial resolution, identifies little to very little impact of a specific land use. The sub-delineation of quinaries into six land uses / management-based HRUs is seen as a major step forward to meaningful land use impact studies in the South African context.
- Impacts of climate change have much more regional impacts. It is only once climate change impacts are superimposed onto land use impacts that such studies become hydrologically meaningful.
- The hydrological analysis of the impacts of actual land uses such as bare/eroded areas or formal or informal urbanisation, when compared with the natural vegetation they replace, indicates:

- that the real hydrological impacts of specific land uses need to be determined at the local level; i.e. in situ, at the scale where the actual replacement of a land use takes place; and
- that on the sub-catchment scale, and even more so at quinary or quaternary resolution, the impacts of individual land uses can become highly muted.
- Local hydrological impacts can be quite dissimilar from one sub-catchment to the next, even within the same study area, depending on the degree/intensity of the specific land use, as well as the local geographical and hydrological characteristics of the natural vegetation they replace, such as soil depth and texture, slope, rainfall interception, and seasonal growth patterns, as well as the surface and aerial characteristics of the vegetation.

8.4 RECOMMENDATIONS FOR FURTHER RESEARCH

The following recommendations are made for future research based on the experience gained during this project.

8.4.1 Climate and hydrological modelling

- Sustained station observation networks are essential for the long-term analysis of local and regional climate trends, as well as for the calibration of satellite-derived climate products. The installation of local weather stations at various altitudes in different parts of the catchment would greatly contribute to future studies.
- Although the water releases from the Fika-Patso Dam are monitored, it does not reflect the "natural" or "uncontrolled" flow in the catchment. Considering the important contribution of streams in the catchment to the Wilge River (via the Elands River), and ultimately to the Vaal River, the installation of streamflow gauging stations in the catchment should be considered.

8.4.2 Ecological risk assessment

- Information on the aquatic ecology of the streams is sparse, fragmented, and uncoordinated. In light of the projected climate and hydrological changes for the catchment, it is clear that there is a dire need for a coordinated and comprehensive ecological assessment of the streams.
- The RQOs gazetted for the Elands River catchment (RSA, 2016) are largely lacking for river water quality. The only category for which numerical limits are available is for nutrients (phosphate, nitrite, and nitrate), most probably due to a lack of information for the other categories (e.g. salts, system variables, toxins, and pathogens). The results from this study highlight the urgent need for setting numerical limits for the other classes and for conducting regular water quality monitoring, especially in the Namahadi, Mphukojwane, and Elands rivers.
- This study found that the RQOs set for the river instream habitats (Category C or higher) are not achieved for most of the middle and lower sites. Particularly concerning is the marked increased in illegal sand mining over the course of this study. This is a serious matter that requires urgent intervention by the relevant authorities, as well as further investigation.

8.4.3 Disaster risk analysis and reduction

- Stage 3 and 4 of the UNDRR (2016) risk assessment process, including monitoring and evaluation, ought to be commissioned in another research project by the WRC.
- That further, and more in-depth, research be conducted on the 17 hazards identified for QwaQwa, but not only on those related to extreme climate change events, since risks are integrated and systematic, with cascading and compounding effects.
- That the approach developed and applied in this study be tested on other climate hotspots; for example, Nelson Mandela Bay District Municipality in the Eastern Cape or eThekwini in KwaZulu-Natal.

REFERENCES

ABBAS A, SCHNEIDER I, BOLLMANN A, FUNKE J, OEHLMANN J, PRASSE C, SCHULTE-OEHLMANN U, SEITZ W, TERNES T, WEBER M and WESELY H (2019) What you extract is what you see: optimising the preparation of water and wastewater samples for in vitro bioassays. Water Research **152** 47-60.

ABDULRAHMAN S (2019) Forensic accounting and fraud prevention in Nigerian public sector: a conceptual paper. International Journal of Accounting & Finance Review **4 (2)** 13-21.

ABIODUN BJ, MAKHANYA N, PETJA B, ABATAN AA and OGUNTUNDE PG (2019) Future projection of droughts over major river basins in Southern Africa at specific global warming levels. Theoretical and Applied Climatology **137** 1785-1799.

ABRAHAMS C, BROWN L, DALE K, EDWARDS F, JEFFRIES M, KLAAR M, LEDGER M, MAY L, MILNER A and MURPHY J (2013) *Ecological issues 2013: the impact of extreme events on freshwater ecosystems*, British Ecological Society, London, UK.

ADEOLA AO and FORBES PB (2022) Antiretroviral drugs in African surface waters: prevalence, analysis, and potential remediation. Environmental Toxicology and Chemistry **41 (2)** 247-262.

ADEYINKA GC, MOODLEY B, BIRUNGI G and NDUNGU P (2018) Quantitative analyses of selected polychlorinated biphenyl (PCB) congeners in water, soil, and sediment during winter and spring seasons from Msunduzi River, South Africa. Environmental Monitoring and Assessment **190 (10)** 1-13.

ADEYINKA GC, MOODLEY B, BIRUNGI G and NDUNGU P (2019) Evaluation of organochlorinated pesticide (OCP) residues in soil, sediment and water from the Msunduzi River in South Africa. Environmental Earth Sciences **78 (6)** 1-13.

ADGER WN, BROOKS N, BENTHAM G, AGNEW M and ERIKSEN S (2004) *New indicators of vulnerability and adaptive capacity.* Technical report 7, Tyndall Centre for Climate Change Research, Norwich, UK.

AGRICULTURAL RESEARCH COUNCIL INSTITUTE FOR SOIL CLIMATE AND WATER, SOUTH AFRICA (2016) *Meteorological data*. Nearest Station: QwaQwa: UNIQWA; Station ID: 30426; Latitude: -28.48287; Longitude: 28.82522; Altitude: 1699 m.a.s.l. Date: 01/01/1986-31/08/2016.

AKSHA SK, RESLER LM, JURAN L and CARSTENSEN LW (2020) A geospatial analysis of multihazard risk in Dharan, Nepal. Geomatics, Natural Hazards and Risk **11 (1)** 88-111.

ALEVIZOU G, ALEXIOU K and ZAMENOPOULOS T (2016) *Making sense of assets: community asset mapping and related approaches*, Working paper, Open University and AHRC, London, UK.

APEH S and EKENTA OE (2012) Assessment of surface water quality of River Benue at Makurdi. Journal of Emerging Trends in Engineering and Applied Sciences **3 (6)** 904-914.

ARCHER E, ENGELBRECHT F, HÄNSLER A, LANDMAN W, TADROSS M and HELMSCHROT J (2018) Seasonal prediction and regional climate projections for Southern Africa. In: *Climate change and adaptive land management in Southern Africa – assessments, changes, challenges, and solutions,* REVERMANN R, KREWENKA KM, SCHMIEDEL U, OLWOCH JM, HELMSCHROT J and JÜRGENS N. Biodiversity & ecology, 6, Klaus Hess Publishers, Göttingen and Windhoek, Germany and Namibia.

ARISTI I, SCHILLER VD, ARROITA M, BARCELO D, PONSATI L, GARCIA-GALAN MJ, SABATER S, ELOSEGI A and ACUNA V (2015) Mixed effects of effluents from a wastewater treatment plant on river ecosystem metabolism: subsidy or stress. Freshwater Biology **60** 1398-1410.

ARNOLD SM, CLARK KE, STAPLES CA, KLECKA GM, DIMOND SS, CASPERS N and HENTGES SG (2013) Relevance of drinking water as a source of human exposure to bisphenol A. Journal of Exposure Science & Environmental Epidemiology **23 (2)** 137-144.

ARSLAN M, ULLAH I, MÜLLER JA, SHAHID N and AFZAL M (2017) Organic micropollutants in the environment: ecotoxicity potential and methods for remediation. In: *Enhancing cleanup of environmental pollutants*, ANJUM NA, GILL SS and TUTEJA N. Vol. 1: Biological approaches, Springer International Publishing, Cham, Switzerland.

ARTHINGTON AH (2012) *Environmental flows: saving rivers in the third millennium*, University of California Press, Berkeley, USA.

ASADGOL Z, MOHAMMADI H, KERMANI M, BADIRZADEH A and GHOLAMI M (2019) The effect of climate change on cholera disease: the road ahead using artificial neural network. PLOS One **14 (11)** e0224813.

AVENANT MF (2022) Assessing the ecological integrity of headwater streams in QwaQwa as a firststep towards ecosystem-based adaptation.In:Southern African mountain conference 2022,ChampagneSportsResort,SouthAfrica,14-17March2022.https://www.samc2022.africa/Documents/Abstracts.pdf. (Accessed 29 February 2023).

AVENANT MF and KOTZE PJ (2004) Fish assessment, habitat integrity and ecological importance and sensitivity: rapid reserve determination for selected sections of the Meulspruit, Grootspruit, Mopeli River, Caledon River and Van Zyl Spruit, Department of Water Affairs, Pretoria, South Africa.

BAKURE BZ, HUNDERA K and ABARA M (2022) Review on the effect of climate change on ecosystem services. *IOP Conference Series: Earth and Environmental Science* **1016 (1)** 012055.

BANSAL S (2021) Stories of change: becoming water abundant by harvesting rainwater. *India Waterportal* 12 July 2021. https://www.indiawaterportal.org/articles/stories-change-becoming-water-abundant-harvesting-rainwater. (Accessed 3rd April 2023).

BARBIER EB (2010) Poverty, development, and environment. Environment and Development Economics **15 (6)** 635-660.

BARBOUR MT (1999) *Rapid bioassessment protocols for use in wadeable streams and rivers: periphyton, benthic macroinvertebrates and fish*, 2nd ed., US Environmental Protection Agency, Office of Water, Washington, D.C., USA.

BARKHUIZEN LM (2015) An assessment of fish and fisheries in impoundments in the upper Orange-Senqu River Basin and Lower Vaal River Basin, doctoral dissertation, University of the Free State, Bloemfontein, South Africa (unpublished).

BATUCAN NSP, TREMBLAY LA, NORTHCOTT GL and MATTHAEI CD (2021) Medicating the environment? A critical review on the risks of carbamazepine, diclofenac and ibuprofen to aquatic organisms. Environmental Advances **7** 100164.

BELCHER T, GROBLER D and BARROW S (2018) *Freshwater assessment report for the Penhill Greenfields development project on Multiple portions farm 468 and 410, Eerste River (Penhill farms) in Cape Town*, Blue Science, Somerset West, South Africa. http://www.capetown.gov.za/councilonline/_layouts/OpenDocument/OpenDocument.aspx?DocumentI d=39064638-bc54-4571-ab6b-78ce39ddfa8f. (Accessed 7th February 2023).

BELLE G, FOSSEY A and ESTERHUIZEN L (2018) Use of multiple indicators to assess the pollution condition of urban streams: a case study of Bloemspruit, Free State province, South Africa. Water and Environment Journal **34 (1)** 1-13.

BELLE JA (2016) The integration of disaster risk reduction and climate change adaptation strategies into wetlands management in the Eastern Free State, South Africa, doctoral dissertation, University of the Free State, Bloemfontein, South Africa (unpublished).

BELLE JA, SITHABILE M and OGUNDEJI AA (2017) Assessing communal farmers' preparedness to drought in the Umguza District, Zimbabwe. International Journal of Disaster Risk Reduction **22** 194-203.

BENEVOLENZA MA and DERIGNE L (2019) The impact of climate change and natural disasters on vulnerable populations: a systematic review of literature. Journal of Human Behavior in the Social Environment **29 (2)** 266-281.

BERRA TM (1997) Some 20th century fish discoveries. Environmental Biology of Fishes **50 (1)** 1-12.

BEST J (2019) Anthropogenic stresses on the world's big rivers. Nature Geoscience 12 (1) 7-21.

BLOEMFONTEIN COURANT (2016) QwaQwa residents continue to queue for water tanks. *Bloemfontein Courant* 19 October 2016. https://www.bloemfonteincourant.co.za/qwaqwa-residents-continue-queue-water-tanks-cosatu/. (Accessed 10th April 2022).

BOTHA M, ARCHIBALD S and GREVE M (2020) What drives grassland-forest boundaries? Assessing fire and frost effects on tree seedling survival and architecture. Ecology and Evolution **10 (19)** 10719-10734.

BRADSHAW CD, POPE E, KAY G, DAVIE JC, COTTRELL A, BACON J, COSSE A, DUNSTONE N, JENNINGS S, CHALLINOR A and CHAPMAN S (2022) Unprecedented climate extremes in South Africa and implications for maize production. Environmental Research Letters **17 (8)** 084028.

BRYER TA (2021) From collaborative responsiveness to collaborative empowerment. In: *Handbook of collaborative public management*, MEEK JW. Elgar handbooks in public administration and management, Edward Elgar Publishing, Cheltenham, UK.

BUNN SE and ARTHINGTON AH (2002) Basic principles and ecological consequences of altered flow regimes for aquatic biodiversity. Environmental Management **30 (4)** 492-507.

CENTRE FOR ENVIRONMENTAL MANAGEMENT (2003) State of the Namahadi and Harts rivers (October 2002 to March 2003): a biomonitoring report. Report prepared for Sedibeng Water, CEM, University of the Free State, Bloemfontein, South Africa.

CESH TV (2010) *Principles of water resources history, development, management and policy*, 3rd ed., John Wiley & Sons, Hoboken, USA.

CHEN Y, JIANG Y, HUANG H, MOU L, RU J, ZHAO J and XIAO S (2018). Long-term and highconcentration heavy-metal contamination strongly influences the microbiome and functional genes in Yellow River sediments. Science of the Total Environment **637** 1400-1412.

CHOPRA S and KUMAR D (2020) Ibuprofen as an emerging organic contaminant in environment, distribution and remediation. Heliyon **6 (6)** e04087.

CHOU J, XIAN T, ZHAO R, XU Y, YANG F and SUN M (2019) Drought risk assessment and estimation in vulnerable eco-regions of China: under the background of climate change. Sustainability **11 (16)** 4463.

CITY OF CAPE TOWN (2013) *City of Cape Town: wastewater and industrial effluent by-law, 2013.* https://openbylaws.org.za/za-cpt/act/by-law/2013/wastewater-industrial-effluent/eng/. (Accessed 16th September 2019).

CLARA M, STRENN B and KREUZINGER N (2004) Carbamazepine as a possible anthropogenic marker in the aquatic environment: investigations on the behaviour of carbamazepine in wastewater treatment and during groundwater infiltration. Water Research **38 (4)** 947-954.

CLARKE B, OTTO F, STUART-SMITH R and HARRINGTON L (2022) Extreme weather impacts of climate change: an attribution perspective. Environmental Research: Climate **1 (1)** 012001.

CLOETE YC, SHADDOCK BF and NEL A (2017) The use of two microbiotests to evaluate the toxicity of sediment from Mpumalanga, South Africa. Water SA **43 (3)** 409-412.

COHEN TM, MCKINNEY M, KARK S and DOR R (2019) Global invasion in progress: modelling the past, current and potential global distribution of the common myna. Biological Invasions **21 (4)** 1295-1309.

COLVIN C, LINDLEY D, GORDON H and SCHACHTSCHNEIDER K (2016) *Water: facts and futures: rethinking South Africa's future*, World Wildlife Fund South Africa, Cape Town, South Africa.

COOPERATIVE GOVERNANCE AND TRADITIONAL AFFAIRS (2020) *CoGTA statement on the water shortages, 22 January.* http://www.cogta.gov.za/?p=7506. (Accessed 27th August 2022).

COSGROVE WJ and LOUCKS DP (2015) Water resources management: current and future challenges and research directions. Water Resources Research **51** 4823-4839.

COUNCIL FOR SCIENTIFIC AND INDUSTRIAL RESEARCH (2010) *A CSIR perspective on water in South Africa – 2010.* CSIR report no. CSIR/NRE/PW/IR/2011/0012/A, Council for Scientific and Industrial Research, Pretoria, South Africa. https://researchspace.csir.co.za/dspace/bitstream/handle/10204/5760/Wall11_2010.pdf?s. (Accessed 1st April 2023).

CRERAR S, VOLKHOLZ J and LUTZ J (2011) *Projection of impacts under plausible scenarios and guidelines on climate change adaption strategies*. ORASECOM report no. 009/20112011. https://wis.orasecom.org/content/study/GIZ/Support%20to%20the%20implementation%20of%20IWR MP%20-%20Phase%202/Documents/11WP4%20GCC_Impacts_009_2011.pdf. (Accessed 21st April 2023).

CULLIS JDS, ROSSOUW N, DU TOIT G, PETRIE D, WOLFAARDT G, DE CLERCQ W and HORN A (2018) Economic risks due to declining water quality in the Breede River catchment. Water SA **44 (3)** 464-473.

CUTTER SL, BORUFF BJ and SHIRLEY LW (2003) Social vulnerability to environmental hazards. Social Science Quarterly **84 (2)** 242-261.

DABROWSKI JM, PEALL SKC, REINECKE AJ, LIESS M and SCHULZE RE (2002) Runoff-related pesticide input into the Lourens River, South Africa: basic data for exposure assessment and risk mitigation at the catchment scale. Water, Air, and Soil Pollution **135** 265-283.

DALLAS HF and RIVERS-MOORE N (2014) Ecological consequences of global climate change for freshwater ecosystems in South Africa. South African Journal of Science **110 (5-6)** 1-11.

DASGUPTA P (2010) Nature's role in sustaining economic development. Philosophical Transactions of the Royal Society B **365** 5-11.

DAVIS-REDDY CL (2017) Observed trends in climate change over Southern Africa. In: *Climate risk and vulnerability: a handbook for Southern Africa*, DAVIS-REDDY CL and VINCENT K, 2nd ed., CSIR, Pretoria, South Africa.

DAVIS-REDDY CL and VINCENT K (2017) *Climate risk and vulnerability: a handbook for Southern Africa*, 2nd ed., CSIR, Pretoria, South Africa.

DEATH RG, FULLER IC and MACKLIN MG (2015) Resetting the river template: the potential for climate-related extreme floods to transform river geomorphology and ecology. Freshwater Biology **60** (12) 2477-2496.

DEPARTMENT OF ENVIRONMENTAL AFFAIRS AND TOURISM (1997) *Environmental potential atlas for South Africa (ENPAT)*, DEAT, Pretoria, South Africa.

DEPARTMENT OF FORESTRY, FISHERIES AND THE ENVIRONMENT (n.d.) South African nationallandcover2018dataset.Availablefordownloadfromhttps://egis.environment.gov.za/gis_data_downloads. (Accessed 10th March 2021).

DEPARTMENT OF WATER AFFAIRS (2011) *Blue drop report: Free State province*, DWA, Pretoria, South Africa. http://www.humansettlements.fs.gov.za/wp-content/uploads/2012/07/2011-Blue-Drop-Report-Free-State.pdf. (Accessed 27th March 2023).

DEPARTMENT OF WATER AFFAIRS AND FORESTRY (1996a) *South African water quality guidelines, Vol. 1: domestic water use*, 2nd ed., Directorate Water Quality Management, DWAF, Pretoria, South Africa.

DEPARTMENT OF WATER AFFAIRS AND FORESTRY (1996b) South African water quality guidelines, Vol. 7: aquatic ecosystems. Directorate Water Quality Management, DWAF, Pretoria, South Africa.

DEPARTMENT OF WATER AFFAIRS AND FORESTRY (1999) *Resource-directed measures for protection of water resources*. River ecosystems, Vol. 3, version 1.0, DWAF, Pretoria, South Africa.

DEPARTMENT OF WATER AFFAIRS AND FORESTRY (2004) *Upper Vaal water management area: internal strategic perspective*. Report prepared by PDNA, WRP Consulting Engineers (Pty) Ltd., WMB and Kwezi-V3 for the DWAF; DWAF report no. P WMA 08/000/00/0304, DWAF, Pretoria, South Africa.

DEPARTMENT OF WATER AFFAIRS AND FORESTRY (2007) *Resource management plan for Fika-Patso Dam and Metsi-Matso Dam: research report (report 3).* Report prepared by Vela VKE Consulting Engineers for the DWAF, DWAF, Pretoria, South Africa.

DEPARTMENT OF WATER AFFAIRS AND FORESTRY (2014) *Bi-annual performance evaluation report for Sterkfontein-QwaQwa regional bulk water supply scheme Free State Chief Directorate: bulk infrastructure programme period: 1 April 2014 to 30 September 2014,* DWAF, Pretoria, South Africa.

DEPARTMENT OF WATER AND SANITATION (2017) *Water supply and drought operating rules for stand-alone schemes: central cluster, Phuthaditjhaba.* DWS report no. P RSA 0000/00/22016, Directorate Water Resource Planning Systems, DWS, Pretoria, South Africa.

DEPARTMENT OF WATER AND SANITATION (2022a) *Blue drop 2022: progress report*, DWS, Pretoria, South Africa. https://ws.dws.gov.za/IRIS/releases/2021_BD_PAT_report_final-28Mar22_MN_web.pdf. (Accessed 8th January 2023).

DEPARTMENT OF WATER AND SANITATION (2022b) *Green drop 2022: national report*, DWS, Pretoria, South Africa. https://ws.dws.gov.za/iris/releases/Report_DPW_Rev02_29Mar22_MN%20web.pdf. (Accessed 8th January 2023).

DEPARTMENT FOR INTERNATIONAL DEVELOPMENT (2011) *Defining disaster resilience: a DFID approach paper*, DFID, London, UK.

DEUTSCHES INSTITUT FÜR NORMUNG (2019) *DIN EN 1484:2019-04: water analysis – guidelines for the determination of total organic carbon (TOC) and dissolved organic carbon (DOC).* https://www.beuth.de/en/standard/din-en-1484/301999219. (Accessed 10th April 2021).

DIETER HH (2014) Health related guide values for drinking-water since 1993 as guidance to assess presence of new analytes in drinking-water. International Journal of Hygiene and Environmental Health **217 (2-3)** 117-132.

DÖLL P and SCHMIED HM (2012) How is the impact of climate change on river flow regimes related to the impact on mean annual runoff? A global-scale analysis. Environmental Research Letters **7 (1)** 014037.

DÖLL P, JIMÉNEZ-CISNEROS B, OKI T, ARNELL NW, BENITO G, COGLEY JG, JIANG T, KUNDZEWICZ ZW, MWAKALILA S and NISHIJIMA A (2015) Integrating risks of climate change into water management. Hydrological Sciences Journal **60 (1)** 4-13.

DRZYMAŁA, J and KALKA J (2020) Ecotoxic interactions between pharmaceuticals in mixtures: diclofenac and sulfamethoxazole. Chemosphere **259** 127407.

DU Y, WANG WQ, PEI ZT, AHMAD F, XU RR, ZHANG YM and SUN LW (2017) Acute toxicity and ecological risk assessment of benzophenone-3 (BP-3) and benzophenone-4 (BP-4) in ultraviolet (UV)-filters. International Journal of Environmental Research and Public Health **14 (11)** 1414.

DUDGEON D, ARTHINGTON AH, GESSNER MO, KAWABATA ZI, KNOWLER DJ, LÉVÊQUE C, NAIMAN RJ, PRIEUR-RICHARD AH, SOTO D, STIASSNY ML and SULLIVAN CA (2006) Freshwater biodiversity: importance, threats, status and conservation challenges. Biological Reviews **81 (2)** 163-182.

DUNNE D (2022) Analysis: Africa's unreported extreme weather in 2022 and climate change. *CarbonBrief* 26 October 2022. https://www.carbonbrief.org/analysis-africas-unreported-extreme-weather-in-2022-and-climate-change/. (Accessed 27th March 2023).

DYE P and JARMAIN C (2004) Water use by black wattle (*Acacia mearnsii*): implications for the link between removal of invading trees and catchment streamflow response: working for water. South African Journal of Science **100 (1)** 40-44.

EDOKPAYI JN, ODIYO JO and DUROWOJU OS (2017) Impact of wastewater on surface water quality in developing countries: a case study of South Africa. Water Quality **10** 66561.

EMERY M and FLORA C (2006) Spiraling-up: mapping community transformation with community capitals framework. Community Development **37 (1)** 19-35.

ENGELBRECHT F, ADEGOKE J, BOPAPE MM, NAIDOO M, GARLAND R, THATCHER M, MCGREGOR J, KATZFEY J, WERNER M, ICHOKU C and GATEBE C (2015) Projections of rapidly rising surface temperatures over Africa under low mitigation. Environmental Research Letters **10 (8)** 085004

ENGELBRECHT F, DEDEKIND Z, MUTHIGE M, MALHERBE J, BERAKI A, ENGELBRECHT C, NGWANA I, LUMSDEN T, LANDMAN W, MAISHA R and MPHESHEA L (2020) *Predictability of hydroclimatic variability over eastern South Africa under climate change*. WRC report no. K5/2457, Water Research Commission, Pretoria, South Africa.

ENGELBRECHT F, MAREAN CW, COWLING RM, ENGELBRECHT CJ, NEUMANN FH, SCOTT L, NKOANA R, O'NEAL D, FISHER E, SHOOK E and FRANKLIN J (2019) Downscaling last glacial maximum climate over Southern Africa. Quaternary Science Reviews **226** 105879.

FENOGLIO S, BO T, CUCCO M, MERCALLI L and MALACARNE G (2010) Effects of global climate change on freshwater biota: a review with special emphasis on the Italian situation. Italian Journal of Zoology **77 (4)** 374-383.

FIELD CB (2012) Managing the risks of extreme events and disasters to advance climate change adaptation: special report of the Intergovernmental Panel on Climate Change, Cambridge University Press, Cambridge, UK.

FILIPE AF, LAWRENCE JE and BONADA N (2013) Vulnerability of stream biota to climate change in Mediterranean climate regions: a synthesis of ecological responses and conservation challenges. Hydrobiologia **719 (1)** 331-351.

FITCHETT J (2021) Climate change has already hit Southern Africa: here's how we know. *The Conversation* 24 October 2021. https://theconversation.com/climate-change-has-already-hit-southern-africa-heres-how-we-know-169062. (Accessed 27th March 2023).

FITZGIBBON C, KURAUCHI Y, OPIYO F and VENTON CC (2014) *Building disaster resilience for sustainable human development: lessons learnt from community based resilience analysis in the Horn of Africa*, UNDP Drylands Development Centre, UN Office for Disaster Risk Reduction, Nairobi, Kenya.

FLORA CB, FLORA JL and FEY S (2004) *Rural communities: legacy and change*, 2nd ed., Westview Press, Colorado, USA.

FOUNTAIN MT and HOPKIN SP (2005) *Folsomia candida* (Collembola): a "standard" soil arthropod. Annual Review of Entomology **50** 201-222.

FREE STATE DEPARTMENT OF COOPERATIVE GOVERNANCE AND TRADITIONAL AFFAIRS (2007) *Free State spatial development framework (final draft)*, Department of Local Government and Housing, Free State province, South Africa. http://www.cogta.fs.gov.za/wp-content/uploads/2012/01/Spatial-Development-Framework-2007.pdf. (Accessed 27th March 2023).

FREE STATE DEPARTMENT OF COOPERATIVE GOVERNANCE AND TRADITIONAL AFFAIRS AND CHELL ENGINEERING (2018) *Updating of the water master plan for Maluti-a-Phofung-Local Municipality*. Report prepared for FS COGTA, COGTA, Bloemfontein, South Africa.

FU S, YANG Y, LIU B, LIU H, LIU J, LIU L and LI P (2020) Peak flow rate response to vegetation and terraces under extreme rainstorms. Agriculture, Ecosystems & Environment **288** 106714.

GALLINA V (2015) An advanced methodology for the multi-risk assessment: an application for climate change impacts in the North Adriatic case study, doctoral dissertation, Ca' Foscari University of Venice, Italy (unpublished).

GALLINA V, TORRESAN S, CRITTO A, SPEROTTO A, GLADE T and MARCOMINI A (2016) A review of multi-risk methodologies for natural hazards: consequences and challenges for a climate change impact assessment. Journal of Environmental Management **168** 123-132.

GALVIN RM (1996) Occurrence of metals in waters: an overview. Water SA 22 (1) 7-18.

GARNER G, VAN LOON AF, PRUDHOMME C and HANNAH DM (2015) Hydroclimatology of extreme river flows. Freshwater Biology **60 (12)** 2461-2476.

GERMAN FEDERAL ENVIRONMENT AGENCY (2003). *Health-orientated guidance values*. www.umweltbundesamt.de/sites/default/files/medien/374/dokumente/gow-empfehlung_2003_46.pdf. (Accessed 30th March 2023).

GOUWS CM, MBAMBO B, MOEKETSI I, MOROTOLO M, MOTLOUNG S and TEMPELHOFF JWN, (2011) What about the votes? Water, sanitation and civil disorientation: the case of Maluti-a-Phofung local municipality (Harrismith). Research report by the Research Group for the Cultural Dynamic of Water, North-West University, Vanderbijlpark. https://scholar.google.com/citations?view_op=view_citation&hl=en&user=OvSH-P4AAAAJ&citation _for_view=OvSH-P4AAAAJ:HDshCWvjkbEC. (Accessed 14th July 2022).

GUDMUNDSSON L, BOULANGE J, DO HX, GOSLING SN, GRILLAKIS MG, KOUTROULIS AG, LEONARD M, LIU J, MÜLLER SCHMIED H, PAPADIMITRIOU L and POKHREL Y (2021) Globally observed trends in mean and extreme river flow attributed to climate change. Science **371 (6534)** 1159-1162.

GUPTA SK (2012) Integrating the informal sector for improved waste management. Private Sector and Development **15** 12-17.

HARRIS JH (1995) The use of fish in ecological assessments. Australian Journal of Ecology **20 (1)** 65-80.

HASSEN A (2008) Vulnerability to drought risk and famine: local responses and external interventions among the Afar of Ethiopia, a study on the Aghini pastoral community, doctoral dissertation, University of Bayreuth, Ethiopia (unpublished).

HAUGER MB, MOUCHEL JM and MIKKELSEN P (2006) Indicators of hazard, vulnerability and risk in urban drainage. Water Science and Technology: A Journal of the International Association on Water Pollution Research **54** 441-50.

HEINO J, VIRKKALA R and TOIVONEN H (2009) Climate change and freshwater biodiversity: detected patterns, future trends and adaptations in northern regions. Biological Reviews **84 (1)** 39-54.

HEWITSON B, TADROSS M and JACK C (2005) Climate change scenarios: conceptual foundations, large scale forcing, uncertainty and the climate context. In: *Climate change and water resources in Southern Africa: studies on scenarios, impacts, vulnerabilities and adaptation*, SCHULZE RE, WRC report no. 1430(1), Water Research Commission, Pretoria, South Africa.

HIMES-CORNELL A, ORMOND C, HOELTING K, NATALIE C, BAN NC, KOEHN JZ, ALLISON EH, LARSON EC, MONSON DH, HUNTINGTON HP and OKEY TA (2018) Factors affecting disaster preparedness, response, and recovery using the community capitals framework. Coastal Management **46 (5)** 335-358.

HOBBS P, COLVIN C, MAHERRY A, BUGAN R, CLARKE S, PETERSEN C, ROSE R and CARSTENS M (2013) *Groundwater reserve determination for the Upper Vaal water management area*. WRC report no. KV 312/13, Water Research Commission, Pretoria, South Africa.

HOEGH-GULDBERG OD, JACOB D, TAYLOR MA, BINDI M, BROWN S, CAMILLONI I, DIEDHIOU A, DJALANTE R, EBI KL, ENGELBRECHT F, GUIOT J, HIJIOKA Y, MEHROTRA S, PAYNE A, SENEVIRATNE SI, THOMAS A, WARREN R and ZHOU G (2018) Impacts of 1.5°C global warming on natural and human systems. In: *Global warming of 1.5°C: an IPCC special report on the impacts of global warming of 1.5°C above pre-industrial levels and related global greenhouse gas emission pathways, in the context of strengthening the global response to the threat of climate change, sustainable development, and efforts to eradicate poverty,* MASSON-DELMOTTE V, ZHAI P, PÖRTNER HO, ROBERTS D, SKEA J, SHUKLA PR, PIRANI A, MOUFOUMA-OKIA W, PÉAN C, PIDCOCK R, CONNORS S, MATTHEWS JBR, CHEN Y, ZHOU X, GOMIS MI, LONNOY E, MAYCOCK T and WATERFIELD T, Cambridge University Press, Cambridge, UK.

HUMPHRIES P and BALDWIN DS (2003) Drought and aquatic ecosystems: an introduction. Freshwater Biology **48 (7)** 1141-1146.

INTERGOVERNMENTAL PANEL ON CLIMATE CHANGE (2012) Managing the risks of extreme events and disasters to advance climate change adaptation: a special report of Working Groups I and II of the Intergovernmental Panel on Climate Change, Cambridge University Press, Cambridge, UK.

INTERGOVERNMENTAL PANEL ON CLIMATE CHANGE (2014) Summary for policymakers. In: *Climate change 2014: impacts, adaptation, and vulnerability. Part A: global and sectoral aspects. Working Group II contribution to the fifth assessment report of the Intergovernmental Panel on Climate Change,* FIELD CB, BARROS VR, DOKKEN DJ, MACH KJ, MASTRANDREA MD, BILIR TE, CHATTERJEE M, EBI KL, ESTRADA YO, GENOVA RC, GIRMA B, KISSEL ES, LEVY AN, MACCRACKEN S, MASTRANDREA PR and WHITE LL, Cambridge University Press, Cambridge, UK.

INTERGOVERNMENTAL PANEL ON CLIMATE CHANGE (2022) Summary for policymakers. In: *Climate change 2022: impacts, adaptation and vulnerability. Contribution of Working Group II to the sixth assessment report of the Intergovernmental Panel on Climate Change*, PÖRTNER HO, ROBERTS DC, TIGNOR MMB, POLOCZANSKA E, MINTENBECK K, ALEGRÍA A, CRAIG M, LANGSDORF S, LÖSCHKE S, MÖLLER V, OKEM A and RAMA B, Cambridge University Press, Cambridge, UK.

INTERNATIONAL ORGANIZATION FOR STANDARDIZATION (2005) *ISO/IEC* 17025:2005: general requirements for the competence of testing and calibration laboratories. https://www.iso.org/standard/39883.html#:~:text=ISO%2FIEC%2017025%3A2005%20specifies,%2C %20and%20laboratory%2Ddeveloped%20methods. (Accessed 28th June 2021).

INTERNATIONAL ORGANIZATION FOR STANDARDIZATION (2006) *ISO* 21427-2:2006: water quality – evaluation of genotoxicity by measurement of the induction of micronuclei – part 2: mixed population method using the cell line V79. https://www.iso.org/standard/39681.html. (Accessed 28th June 2021).

INTERNATIONAL ORGANIZATION FOR STANDARDIZATION (2012a) *ISO* 6341:2012: water quality – determination of the inhibition of the mobility of Daphnia magna Straus (Cladocera, Crustacea) – acute toxicity test. https://www.iso.org/standard/54614.html. (Accessed 28th June 2021).

INTERNATIONAL ORGANIZATION FOR STANDARDIZATION (2012b) *ISO 11350:2012* (Accessed 28th June 2021).

INTERNATIONAL ORGANIZATION FOR STANDARDIZATION (2014) *ISO* 11267:2014(*en*) *soil quality* – *inhibition* of reproduction of Collembola (Folsomia candida) by *soil* contaminants. https://www.iso.org/obp/ui/#iso:std:iso:11267:ed-2:v1:en. (Accessed 28th June 2021).

INTERNATIONAL ORGANIZATION FOR STANDARDIZATION (2018) *ISO* 19040-1:2018 water quality – determination of the estrogenic potential of water and waste water – part 1: yeast estrogen screen (Saccharomyces cerevisiae). https://iso.org/standard/64450.html. (Accessed 10th March 2022).

JAEGER KL, OLDEN JD and PELLAND NA (2014) Climate change poised to threaten hydrologic connectivity and endemic fishes in dryland streams. Proceedings of the National Academy of Sciences **111 (38)** 13894-13899.

JENKINS C (2021) David Attenborough to UN: climate change is biggest threat modern humans have ever faced. *The Hill* 23 February 2021. https://thehill.com/homenews/news/540058-david-attenborough-warns-un-security-council-on-climate-change-i-dont-envy-you/. (Accessed 24th March 2023).

JOACHIM S, BEAUDOUIN R, DANIELE G, GEFFARD A, BADO-NILLES A, TEBBY C, PALLUEL O, DEDOURGE-GEFFARD O, FIEU M, BONNARD M and PALOS-LADEIRO M (2021) Effects of diclofenac on sentinel species and aquatic communities in semi-natural conditions. Ecotoxicology and Environmental Safety **211** 111812.

JORDAAN A (2017) Vulnerability, adaptation to and coping with drought: the case of commercial and subsistence rain fed farming in the Eastern Cape, Vol. I. WRC report no. TT716/1/17, Water Research Commission, Pretoria, South Africa.

KAPPES MS, KEILER M, VON ELVERFELDT K and GLADE T (2012) Challenges of analyzing multihazard risk: a review. Natural Hazards **64** 1925-1958. KAZA U, KNIGHT AK and BAHNA SL (2007) Risk factors for the development of food allergy. Current Allergy and Asthma Reports **7(3)** 182-186.

KAŹMIERCZAK A and CAVAN G (2011) Surface water flooding risk to urban communities: analysis of vulnerability, hazard and exposure. Landscape and Urban Planning **103 (2)** 185-197.

KHALANE S (2020) QwaQwa water: punish the poor, enrich the powerful. *New Frame* 11 February 2020.

KIENZLE SW, LORENTZ SA and SCHULZE RE (1997) *Hydrology and water quality of the Mgeni catchment*. WRC report no. TT87/97, Water Research Commission, Pretoria, South Africa.

KINGS, S (2017) It's raining but QwaQwa hopes dry up. Mail & Guardian 23 February 2017.

KLASIC M, EKSTROM J, LUBELL M and BEDSWORTH L (2017) *Water quality and extreme events in the face of climate change: does science supply meet demand?* Poster presentation at the 2017 World Water Week in Stockholm, Sweden.

KLEYNHANS CJ (1996) A qualitative procedure for the assessment of the habitat integrity status of the Luvuvhu River (Limpopo system, South Africa). Journal of Aquatic Ecosystem Health **5 (1)** 41-54.

KLEYNHANS CJ (1999) The development of a fish index to assess the biological integrity of South African rivers. Water SA **25** 265-278.

KLEYNHANS CJ (2000) Desktop estimates of the ecological importance and sensitivity categories (EISC), default ecological management classes (DEMC), present ecological status categories (PESC), present attainable ecological management class (present AEMC) and best attainable ecological management class (best AEMC) for quaternary catchments in South Africa, Institute for Water Quality Studies, Department of Water Affairs and Forestry, Pretoria, South Africa.

KLEYNHANS CJ (2007) Module D: volume 1: fish response assessment index (FRAI). In: *River ecoclassification: manual for ecostatus determination (version 2)*, KLEYNHANS CJ, Joint Water Research Commission and Department of Water Affairs and Forestry report, WRC report no. TT330/08, Water Research Commission, Pretoria, South Africa.

KLEYNHANS CJ, LOUW MD and MOOLMAN J (2007) *Module D: volume 2: reference frequency of occurrence of fish species in South Africa: manual for ecostatus determination (version 2)*. Joint Water Research Commission and Department of Water Affairs and Forestry report, WRC report no. TT330/08, Water Research Commission, Pretoria, South Africa.

KLEYNHANS CJ, THIRION C and MOOLMAN J (2005) *A level I river ecoregion classification system for South Africa, Lesotho and Swaziland*. Report no. N/0000/00/REQ0104, Resource Quality Services, Department of Water Affairs and Forestry, Pretoria, South Africa.

KLEYNHANS I (2021) *Maluti-a-Phofung fails to supply water in a COVID-19 hotspot*. Press release by the Democratic Alliance, 26 January 2021. https://fs.da.org.za/2021/01/maluti-a-phofung-fails-to-supply-water-in-a-COVID-19-hotspot. (Accessed 2nd April 2023).

KLINE C, MCGEHEE N and DELCONTE J (2019) Built capital as a catalyst for community-based tourism. Journal of Travel Research **58 (6)** 899-915.

KNOESEN D, DICKENS C, KUNZ R, PRINGLE C, SCHULZE R and SUMMERTON M (2009) *Water for the future: impacts of climate change on water resources in the Orange-Senqu River basin*, Institute of Natural Resources, Pietermaritzburg, South Africa.

KNOESEN DM (2011) Integrating hydro-climatic hazards and climate change as a tool for adaptive water resources management in the Orange River catchment, doctoral dissertation, University of KwaZulu-Natal, Pietermaritzburg, South Africa (unpublished).

KOMENDANTOVA N, MRZYGLOCKI R, MIGNAN A, KHAZAI B, WENZEL F, PATT A and FLEMING K (2014) Multi-hazard and multi-risk decision-support tools as a part of participatory risk governance: feedback from civil protection stakeholders. International Journal of Disaster Risk Reduction **8** 50-67.

KÖNEMANN S, KASE R, SIMON E, SWART K, BUCHINGER S, SCHLÜSENER M, HOLLERT H, ESCHER BI, WERNER I, AÏT-AÏSSA S and VERMEIRSSEN E (2018) Effect-based and chemical analytical methods to monitor oestrogens under the European Water Framework Directive. TrAC Trends in Analytical Chemistry **102** 225-235.

KRAHNSTÖVER T, SANTOS N, GEORGES K, CAMPOS L and ANTIZAR-LADISLAO B (2022) Lowcarbon technologies to remove organic micropollutants from wastewater: a focus on pharmaceuticals. Sustainability **14 (18)** 11686.

KRUGER A, PIETERS R, HORN S, VAN ZIJL C and ANECK-HAHN N (2022) The role of effect-based methods to address water quality monitoring in South Africa: a developing country's struggle. Environmental Science and Pollution Research **29 (56)** 84049-84055.

KRUGER AC and SEKELE SS (2013) Trends in extreme temperature indices in South Africa: 1962-2009. International Journal of Climatology **33 (3)** 661-676.

KUMAR P and YASHIRO M (2014) The marginal poor and their dependence on ecosystem services: evidence from South Asia and sub-Saharan Africa. In: *Marginality*, VON BRAUN J and GATZWEILER F, Springer, Dordrecht, the Netherlands.

KUNZ PY, SIMON E, CREUSOT N, JAYASINGHE BS, KIENLE C, MALETZ S, SCHIFFERLI A, SCHÖNLAU C, AÏT-AÏSSA S, DENSLOW ND and HOLLERT H (2017) Effect-based tools for monitoring estrogenic mixtures: evaluation of five in vitro bioassays. Water Research **110** 378-388.

KUSANGAYA S, MUNYARADZI DS and MBENGO I (2017) Water resources. In: *Climate: risk and vulnerability: a handbook for Southern Africa*, DAVIS-REDDY CL and VINCENT K, 2nd ed., CSIR, Pretoria, South Africa.

LAMM KW, BORRON A and ATKINS K (2020) The community diagnostics and social impact toolkit: development and validation of a reliable measure. Journal of Agricultural Education **61 (4)** 249-265.

LAURA M, TARTARI G, SALERNO F, VALSECCHI L, BRAVI C, LORENZI E, GENONI P and GUZZELLA L (2017) Climate change impacts on sediment quality of subalpine reservoirs: implications on management. Water **9 (9)** 680.

LEEDY PD and ORMROD JE (2021) *Practical research: planning and design*, 12th ed., Pearson Education Limited, UK.

LEIGH C, BUSH A, HARRISON ET, HO SS, LUKE L, ROLLS RJ and LEDGER ME (2015) Ecological effects of extreme climatic events on riverine ecosystems: insights from Australia. Freshwater Biology **60 (12)** 2620-2638.

LI S, WEN J, HE B, WANG J, HU X and LIU J (2020) Occurrence of caffeine in the freshwater environment: implications for ecopharmacovigilance. Environmental Pollution **263**, 114371.

LIU C, WALLING DE and HE Y (2018) The International Sediment Initiative case studies of sediment problems in river basins and their management. International Journal of Sediment Research **33 (2)** 216-219.

LUMSDEN TG, SCHULZE RE and HEWITSON BC (2009) Evaluation of potential changes in hydrologically relevant statistics of rainfall in Southern Africa under conditions of climate change. Water SA **35 (5)** 649-656.

LYNCH SD (2004) *Development of a raster database of annual, monthly and daily rainfall for Southern Africa.* WRC report no. 1156/1/04, Water Research Commission, Pretoria, South Africa.

MACKELLAR N, NEW M and JACK C (2014) Observed and modelled trends in rainfall and temperature for South Africa: 1960-2010. South African Journal of Science **110 (7-8)** 1-13.

MACUPE B (2020) Impossible choices in QwaQwa: what it takes to get water. *Mail & Guardian* 7 February 2020. https://mg.co.za/article/2020-02-07-impossible-choices-in-QwaQwa-what-it-takes-to-get-water/. (Accessed 2nd April 2023).

MAGDEBURG A, STALTER D, SCHLÜSENER M, TERNES T and OEHLMANN J (2014) Evaluating the efficiency of advanced wastewater treatment: target analysis of organic contaminants and (geno-) toxicity assessment tell a different story. Water Research **50** 35-47.

MAIL & GUARDIAN (2004) Teenager dies after Harrismith protest. *Mail & Guardian* 31 August 2004. https://mg.co.za/article/2004-08-31-teenager-dies-after-harrismith-protest/. (Accessed 22nd February 2023).

MALUTI-A-PHOFUNG LOCAL MUNICIPALITY (2007) *Integrated development plan 2007/2008*, Malutia-Phofung Local Municipality, Witsieshoek.

MALUTI-A-PHOFUNG LOCAL MUNICIPALITY (2018) *Final integrated development plan, 2018/2019*, Maluti-a-Phofung Local Municipality, Witsieshoek.

MAPHELA B and CLOETE F (2020) Johannesburg's implementation of the National Water Act, 1998 in Soweto, South Africa. Development Southern Africa **37 (4)** 535-552.

MASON SJ, WAYLEN PR, MIMMACK GM, RAJARATNAM B and HARRISON JM (1999) Changes in extreme rainfall events in South Africa. Climatic Change **41** 249-257.

MASSON V, LEMONSU A, HIDALGO J and VOOGT J (2020) Urban climates and climate change. Annual Review of Environment and Resources **45** 411-444.

MASUABI Q (2020) There are two ANCs here. City Press 26 January 2020.

MATTOS D (2015) Community capitals framework as a measure of community development. Cornhusker Economics **811** 2 September. https://digitalcommons.unl.edu/agecon_cornhusker/811. (Accessed 2nd April 2023).

MAÚRE G, PINTO I, NDEBELE-MURISA M, MUTHIGE M, LENNARD C, NIKULIN G, DOSIO A and MEQUE A (2018) The Southern African climate under 1.5°C and 2°C of global warming as simulated by CORDEX regional climate models. Environmental Research Letters **13 (6)** 065002.

MAYUNGA JS (2007) Understanding and applying the concept of community disaster resilience: a capital-based approach. Summer Academy for Social Vulnerability and Resilience Building **1 (1)** 1-16.

MDLALANE A (2020) QwaQwa water crises blamed on poor infrastructure. *SABC News* 1 February 2020. https://www.sabcnews.com/sabcnews/QwaQwa-water-crises-blamed-on-poor-infrastructure/. (Accessed 2nd April 2023).

MELORE TW (2017) Improving the climate change resilience of informal settlements in mountainous regions of Africa: comparative case studies of QwaQwa in South Africa and Konso in Ethiopia, doctoral dissertation, University of the Free State, Bloemfontein, South Africa (unpublished).

MELORE TW and NEL V (2020) Resilience of informal settlements to climate change in the mountainous areas of Konso, Ethiopia and QwaQwa, South Africa. Jàmbá: Journal of Disaster Risk Studies **12 (1)** 1-9.

MERZ SK (2004) *Impacts of bushfires on flood hydrology and erosion rates*, Department of Sustainability and Environment, Victoria, Australia.

MILLENNIUM ECOSYSTEM ASSESSMENT (2005) *Ecosystems and human well-being: synthesis*, Island Press, Washington, D.C., USA.

MILLENNIUM ECOSYSTEM ASSESSMENT (2008) *Living beyond our means: natural assets and human well-being*, Island Press, Washington, D.C., USA.

MISRA AK (2014) Climate change and challenges of water and food security. International Journal of Sustainable Built Environment **3 (1)** 153-165.

MNISI N (2020) *Water scarcity in South Africa: a result of physical or economic factors?* https://hsf.org.za/publications/hsf-briefs/water-scarcity-in-south-africa-a-result-of-physical-or-economic-factors. (Accessed 23rd March 2021).

MOCWAGAE K (2020) Exploring the QwaQwa water crisis for effective planning in post-apartheid South Africa, doctoral dissertation, University of the Free State, Bloemfontein, South Africa (unpublished).

MOHAMED AA and MUKWADA G (2019) Temperature changes in the Maloti-Drakensberg region: an analysis of trends for the 1960-2016 period. Atmosphere **10 (8)** 471.

MOLOI M, OGBEIDE O and VOUA OTOMO P (2020) Probabilistic health risk assessment of heavy metals at wastewater discharge points within the Vaal River Basin, South Africa. International Journal of Hygiene and Environmental Health **224** 113421.

MORRONGIELLO JR, BEATT SJ, BENNETT JC, CROOK DA, IKEDIFE DN, KENNARD MJ, KEREZSY A, LINTERMANS M, MCNEIL DG, PUSEY BJ and RAYNER T (2011) Climate change and its implications for Australia's freshwater fish. Marine and Freshwater Research **62 (9)** 1082-1098.

MOSOLLOANE PM, BREDENHAND E and VOUA OTOMO P (2018) Laboratory assessment of the toxic effects of sewage sludge from the Maluti-Drakensberg region on a terrestrial oligochaete species. Ecotoxicology **28** 8-91.

MOTAUNG M (2020) Fury over QwaQwa water crisis. The Star 24 January 2020.

MOTHOLO LM (2014) Characterization of macro- and micro-invertebrates and assessment of water quality in dams and rivers of QwaQwa, master's thesis, University of the Free State, Bloemfontein, South Africa (unpublished).

MUCINA L and RUTHERFORD MC (2006) *The vegetation of South Africa, Lesotho and Swaziland*, South African National Biodiversity Institute, Pretoria, South Africa.

MUKHEIBIR P (2008) Water resources management strategies for adaptation to climate-induced impacts in South Africa. Water Resources Management **22 (9)** 1259-1276.

MUKWADA G and MANATSA D (2018) Spatiotemporal analysis of the effect of climate change on vegetation health in the Drakensberg Mountain Region of South Africa. Environmental Monitoring and Assessment **190 (6)** 1-21.

MUKWADA G and MUTANA S (2023) Surviving the limits imposed by a changing climate: the case of urban drought and water supply sustainability in Phuthaditjhaba. In: *Sustainable futures in Southern Africa's mountains: multiple perspectives on an emerging city*, MEMBRETTI A, TAYLOR SJ and DELVES JL, Springer International Publishing, Cham, Switzerland.

MULEMA, AA, BOONABAANA B, DEBEVEC L, NIGUSSIE L, ALEMU M and KAARIA S (2021) Spiraling up and down: mapping women's empowerment through agricultural interventions using the Community Capitals Framework in rural Ethiopia. Community Development **52 (1)** 113-130.

MUNICIPALITIES OF SOUTH AFRICA (2022) *Maluti-A-Phofung Local Municipality (FS194)*. https://municipalities.co.za/overview/1051/Maluti-A-Phofung-local-municipality. (Accessed 25th July 2022).

MUYAMBO F, JORDAAN A and BAHTA Y (2017) The role of indigenous knowledge in drought risk reduction: a case of communal farmers in South Africa. Jàmbá: Journal of Disaster Risk Studies **9 (1)** 1-6.

NATIONAL DISASTER MANAGEMENT AUTHORITY (2014) *National policy guidelines on vulnerable groups in disasters*. https://faolex.fao.org/docs/pdf/pak173386.pdf. (Accessed 27th July 2022).

NATIONAL DISASTER MANAGEMENT CENTRE (2016) *Guideline: conducting comprehensive disaster risk assessments: part 1*, Department of Cooperative Governance and Traditional Affairs, Pretoria, South Africa.

NATIONAL DISASTER MANAGEMENT CENTRE (2017) *Stakeholder management and communication*, Department of Cooperative Governance and Traditional Affairs, Pretoria, South Africa.

NATIONAL PLANNING COMMISSION (2012) *National Development Plan 2030: our future – make it work.* https://www.gov.za/sites/default/files/gcis_document/201409/ndp-2030-our-future-make-it-workr.pdf. (Accessed 8th January 2022).

NAYDENOV N (2016) *Political capital conceptualization: reclaiming the heart of democracy.* https://papers.ssrn.com/sol3/papers.cfm?abstract_id=2041998. (Accessed 25th July 2022).

NDLOVU S and NAIDOO S (2022) *Community-led waste disposal in the Umkhomazi catchment in KwaZulu-Natal, South Africa: a situational analysis and embedded awareness-raising campaign for nappy pollution*, Institute of Natural Resources NPC, Pietermaritzburg, South Africa. https://data.jncc.gov.uk/data/bd078383-adb7-4fd4-a3ed-5c90fe67a477/rptp-report-nappy-waste-vs0.2.pdf. (Accessed 25th July 2022).

NEL JL, DRIVER A, STRYDOM WF, MAHERRY A, PETERSEN C, HILL L, ROUX DJ, NIENABER S, VAN DEVENTER H, SWARTZ E and SMITH-ADAO LB (2011) *Atlas of freshwater ecosystem priority areas in South Africa: maps to support sustainable development of water resources.* WRC report no. TT500/11, Water Research Commission, Pretoria, South Africa.

NEL JL, LE MAITRE DC, ROUX DJ, COLVIN C, SMITH JS, SMITH-ADAO LB, MAHERRY A and SITAS N (2017) Strategic water source areas for urban water security: making the connection between protecting ecosystems and benefiting from their services. Ecosystem Services **28** 251-259.

NIANG I, RUPPEL OC, ABDRABO MA, ESSEL A, LENNARD C, PADGHAM J and URQUHART P (2014) Africa. In: *Climate change 2014: impacts, adaptation, and vulnerability. Part B: regional aspects. Working Group II contribution to the Intergovernmental Panel on Climate Change fifth assessment report*, BARROS VR, FIELD CB, DOKKEN DJ, MASTRANDREA MD, MACH KJ, BILIR TE, CHATTERJEE M, EBI KL, ESTRADA YO, GENOVA RC, GIRMA B, KISSEL ES, LEVY AN, MACCRACKEN S, MASTRANDREA PR and WHITE LL, Cambridge University Press, Cambridge, UK.

NIBAMUREKE MCU, BARNHOORN IE and WAGENAAR GM (2019) Assessing the potential effects of nevirapine in South African surface water on fish growth: a chronic exposure of *Oreochromis mossambicus*. South African Journal of Science **115 (3-4)** 1-6.

NKOMBI Z and WENTINK GJ (2022) The role of public participation in disaster risk reduction initiatives: the case of Katlehong township. Jàmbá: Journal of Disaster Risk Studies **14 (1)** 1203.

NOBLE M and WRIGHT G (2013) Using indicators of multiple deprivation to demonstrate the spatial legacy of apartheid in South Africa. Social Indicators Research **112 (1)** 187-201.

NYAM YS, KOTIR JH, JORDAAN AJ, OGUNDEJI AA and TURTON AR (2020) Drivers of change in sustainable water management and agricultural development in South Africa: a participatory approach. Sustainable Water Resources Management **6** 62. https://doi.org/10.1007/s40899-020-00420-9. (Accessed 20th April 2021).

OLIVEIRA S, ROCHA J and SÁ A (2021) Wildfire risk modelling. Current Opinion in Environmental Science & Health **23** 100274.

ORANGE-SENQU RIVER COMMISSION (2013) *The Orange-Senqu river basin infrastructure catalogue*. ORASECOM report no. 001/2013, ORASECOM, Centurion, South Africa.

ORGANISATION FOR ECONOMIC CO-OPERATION AND DEVELOPMENT (2004) *Earthworm reproduction test: guideline for testing chemicals, no.* 222, OECD, Paris, France.

ORGANISATION FOR ECONOMIC CO-OPERATION AND DEVELOPMENT (2008) Natural resources and pro-poor growth: the economics and politics, OECD, Paris, France.

ORGANISATION FOR ECONOMIC CO-OPERATION AND DEVELOPMENT (2012) *Disaster risk assessment and risk financing: a G20 / OECD methodological framework.* https://www.oecd.org/gov/risk/g20oecdframeworkfordisasterriskmanagement.htm#:~:text=G20%2FO ECD%20methodological%20framework%20on%20disaster%20risk%20assessment%20and%20risk %20financing,-%E2%80%8C&text=It%20is%20recognised%20that%20disasters,potential%20 cascading%20and%20global%20effects. (Accessed 20th April 2021).

ORGANISATION FOR ECONOMIC CO-OPERATION AND DEVELOPMENT (2016) *OECD guidelines* for the testing of chemicals, section 4, test no. 487: in vitro mammalian cell micronucleus test, OECD Publishing, Paris, France. https://doi.org/10.1787/9789264264861-en. (Accessed 20th April 2021).

ORGANISATION FOR ECONOMIC CO-OPERATION AND DEVELOPMENT (2020) *OECD guidelines for the testing of chemicals, section 4, test no. 471: bacterial reverse mutation test*, OECD Publishing, Paris, France. https://doi.org/10.1787/9789264071247-en. (Accessed 20th April 2021).

OWEN P (2017) Disposable diapers a growing threat to African rivers. *Green Times* 14 November 2017. https://thegreentimes.co.za/disposable-diapers-growing-threat-african-rivers/. (Accessed 25th April 2021).

PERKINS-KIRKPATRICK SE and GIBSON PB (2017) Changes in regional heatwave characteristics as a function of increasing global temperature. Scientific Reports **7 (1)** 1-12.

PERSOONE G, MARSALEK B, BLINOVA I, TÖRÖKNE A, ZARINA D, MANUSADZIANAS L, NALECZ-JAWECKI G, TOFAN L, STEPANOVA N, TOTHOVA L and KOLAR B (2003) A practical and userfriendly toxicity classification system with microbiotests for natural waters and wastewaters. Environmental Toxicology: An International Journal **18 (6)** 395-402.

PETERS DJ (2016) *Community Capital Framework*, Guest lecture at DSN S 546 Interdisciplinary Design Studio, Iowa State University, USA.

PETERSON GCL, PRINCE SE and RAPPOLD AG (2021) Trends in fire danger and population exposure along the wildland-urban interface. Environmental Science & Technology **55 (23)** 16257-16265.

PICKLES J and WOODS J (1992) South Africa's homelands in the age of reform: the case of QwaQwa. Annals of the Association of American Geographers **82 (4)** 629-652.

PIRES AM, MAGALHÃES MF, MOREIRA DA COSTA L, ALVES MJ and COELHO MM (2008) Effects of an extreme flash flood on the native fish assemblages across a Mediterranean catchment. Fisheries Management and Ecology **15 (1)** 49-58.

PLETTERBAUER F, MELCHER A and GRAF W (2018) Climate change impacts in riverine ecosystems. In: *Riverine ecosystem management*, SCHMUTZ S and SENDZIMIR J, Aquatic ecology series 8, Springer International Publishing, Cham, Switzerland.

POFF NL and ZIMMERMAN JK (2010) Ecological responses to altered flow regimes: a literature review to inform the science and management of environmental flows. Freshwater Biology **55 (1)** 194-205.

POSTEL S and RICHTER B (2003) *Rivers for life: managing water for people and nature*, Island Press, Washington, D.C., USA.

QUANDT A, NEUFELDT H and MCCABE JT (2017) The role of agroforestry in building livelihood resilience to floods and drought in semiarid Kenya. Ecology and Society **22 (3)** 10.

RAHMAN HA (2020) Malaysian youth and environmental sustainability: a review. Perspektif Jurnal Sains Sosial dan Kemanusiaan **12 (2)** 43-54.

RANDWATER (2019) Integrated annual report, Randwater, Johannesburg, South Africa. https://www.randwater.co.za/Annual%20Reports/Annual%20Reports/2018-2019%20Annual%20Report%20Final/10374%20RANDWATER%20AR%20FINAL%20Electronic%20 2019.pdf. (Accessed 3rd February 2022).

REIFFERSCHEID G, DILL F, FIEBLINGER D, GMINSKI R, GRUMMT HJ, HAFNER C, HOLLERT H, KUNZ S, RODRIGO G, STOPPER H and ZIEMANN C (2007) Measurement of genotoxicity in waste water samples – results of a collaborative study on the in vitro micronucleus test in the context of standardisation according to ISO. Umweltwissenschaften und Schadstoff-Forschung **19** 7-16.

REPUBLIC OF SOUTH AFRICA (1996) *Constitution of the Republic of South Africa, no. 108 of 1996*, Government Printer, Pretoria, South Africa.

REPUBLIC OF SOUTH AFRICA (1998) National Water Act, no. 36 of 1998, Government Printer, Pretoria, South Africa.

REPUBLIC OF SOUTH AFRICA (2002) *Disaster Management Act, no. 57 of 2002*, Government Printer, Pretoria, South Africa.

REPUBLIC OF SOUTH AFRICA (2015) *Disaster Management Amendment Act, no. 16 of 2015,* Government Printer, Pretoria, South Africa.

REPUBLIC OF SOUTH AFRICA (2016) National Water Act: classes and resource quality objectives of water resources for catchments of the Upper Vaal. Government Gazette, **610 (39943)** 618-642.

REVERMANN R, KREWENKA KM, SCHMIEDEL U, OLWOCH JM, HELMSCHROT J and JURGENS N (2018) *Climate change and adaptive land management in Southern Africa*, Klaus Hess Publishers, Gottingen and Windhoek, Germany and Namibia.

RIVER HEALTH PROGRAMME (2003) *State of the rivers report: Free State region river systems*, Department of Water Affairs and Forestry, Pretoria, South Africa.

ROOSEBOOM A, VERSTER E, ZIETSMAN HL and LOTRIET HH (1992) *The development of the new sediment yield map of Southern Africa.* WRC report no. 297/2/92, Water Research Commission, Pretoria, South Africa.

ROWNTREE KM and WADESON RA (1999) A hierarchical framework for categorising the geomorphology of selected South African rivers. WRC report no. 497/1/99, Water Research Commission, Pretoria, South Africa.

SAFE SH and GAIDO K (1998) Phytoestrogens and anthropogenic estrogenic compounds. Environmental Toxicology and Chemistry **17 (1)** 119-126.

SA NEWS (2020) Task team to resolve Maluti-a-Phofung and Eskom woes. *South African Government News Agency* 27 November 2020. https://www.sanews.gov.za/south-africa/task-team-resolve-maluti-phofung-and-eskom-woes. (Accessed 25th February 2021).

SCHANZE, J (2016) Resilience in flood risk management – exploring its added value for science and practice. E3S Web of Conferences **7** 08003.

SCHMIDT EJ and SCHULZE RE (1984) *Improved estimates of peak flow rates using modified SCS lag equations.* Agricultural Catchments Research Units report no. 17, Department of Agricultural Engineering, University of Natal, Durban, South Africa. https://www.wrc.org.za/wp-content/uploads/mdocs/63-1-84.pdf. (Accessed 20th March 2022).

SCHOLES R and ENGELBRECHT F (2021) *Climate impacts in Southern Africa during the 21st century*. Report to the Centre for Environmental Rights, University of the Witwatersrand, Johannesburg. https://cer.org.za/wp-content/uploads/2021/09/Climate-impacts-in-South-Africa_Final_September_ 2021.FINAL_.pdf. (Accessed 27th March 2023).

SCHULZE RE (1979) *Hydrology and water resources of the Drakensberg*, Natal Town and Regional Planning Commission, Pietermaritzburg, South Africa.

SCHULZE RE (1995) *Hydrology and agrohydrology: a text to accompany the ACRU 3.00 agro-hydrological modelling system.* WRC report no. TT69/95, Water Research Commission, Pretoria, South Africa.

SCHULZE RE (2001) Comparison of spray drift-and runoff-related input of azinphos-methyl and endosulfan from fruit orchards into the Lourens River, South Africa. Chemosphere **45** 543-551.

SCHULZE RE (2004) Determination of baseline land cover variables for applications in assessing land use impacts on hydrological responses in South Africa. In: *Development and evaluation of an installed hydrological modelling system*, SCHULZE RE and PIKE A, WRC report no. 1155/1/04, Water Research Commission, Pretoria, South Africa.

SCHULZE RE (2008) *South African atlas of climatology and agrohydrology.* WRC report no. 1489/1/08, Water Research Commission, Pretoria, South Africa [interactive DVD].

SCHULZE RE (2012) A 2011 perspective on climate change and the South African water sector. WRC report no. TT518/12, Water Research Commission, Pretoria, South Africa.

SCHULZE RE (2022) South African and international verification studies of the ACRU daily time-step model across a range of processes, applications and spatial scales (report 3). In: *A national assessment of potential climate change impacts on the hydrological yield of different hydro-climatic zones of South Africa*, SCHÜTTE S, SCHULZE RE and CLARK DJ, WRC report no. 2833, Water Research Commission, Pretoria, South Africa.

SCHULZE RE and HORAN MJC (2010) Methods 1: delineation of South Africa, Lesotho and Swaziland into quinary catchments. Cited in: SCHULZE RE, HEWITSON BC, BARICHIEVY KR, SCHULZE RE, HORAN MJC, KUNZ RP, LUMSDEN TG and KNOESEN DM (2010) Methods 2: development of the Southern African Quinary catchments database. In: *Methodological approaches to assessing eco-hydrological responses to climate change in South Africa*, SCHULZE RE, HEWITSON BC, BARICHIEVY KR, TADROSS M, KUNZ RP, HORAN MJC and LUMSDEN TG, WRC report no. 1562/1/10, Water Research Commission, Pretoria, South Africa.
SCHULZE RE, HORAN MJC, KUNZ RP, LUMSDEN TG and KNOESEN DM (2010) Methods 2: development of the Southern African quinary catchments database. In: *Methodological approaches to assessing eco-hydrological responses to climate change in South Africa*, SCHULZE RE, HEWITSON BC, BARICHIEVY KR, TADROSS M, KUNZ RP, HORAN MJC and LUMSDEN TG, WRC report no. 1562/1/10, Water Research Commission, Pretoria, South Africa.

SCHULZE RE and MAHARAJ M (2004) *Development of a database of gridded daily temperatures for Southern Africa.* WRC report no. 1156/2/04, Water Research Commission, Pretoria, South Africa.

SCHULZE RE and SCHMIDT EJ (1995) Peak discharge. In: *Hydrology and agrohydrology: a text to accompany the ACRU 3.00 agrohydrological modelling system*, SCHULZE RE, WRC report no. TT69/95, Water Research Commission, Pretoria, South Africa.

SCHULZE RE and SMITHERS JC (2004) The ACRU agrohydrological modelling system as of 2002: background, concepts, structure, output, typical applications and operations. In: *Modelling as a tool in integrated water resources management: conceptual issues and case study applications*, SCHULZE RE, WRC report no. 749/1/04, Water Research Commission, Pretoria, South Africa.

SCOTT DF (1993) The hydrological effects of fire in South African mountain catchments. Journal of Hydrology **150 (2-4)** 409-432.

SEDDON N, CHAUSSON A, BERRY P, GIRARDIN CA, SMITH A and TURNER B (2020) Understanding the value and limits of nature-based solutions to climate change and other global challenges. Philosophical Transactions of the Royal Society B **375 (1794)** 20190120.

SEKHELE N and VOUA OTOMO P (2023) Afromontane community's dependence on the water and climate change nexus of the Maloti-Drakensberg mountain range: the case of Phuthaditjhaba. In: *Sustainable futures in Southern Africa's mountains: multiple perspectives on an emerging city*, MEMBRETTI A, TAYLOR SJ and DELVES JL, Springer International Publishing, Cham, Switzerland.

SHARMA J, INDU KM, ESTEVES T, NEGI P, SUSHMA S, DASGUPTA S, BARUA A, BALA G and RAVINDRANATH NH (2018) *Vulnerability and risk assessment: framework, methods and guidelines for the Indian Himalayan region.* https://www.weadapt.org/system/files_force/vulnerability_manual_iisc_ihcap_0.pdf. (Accessed 3rd March 2022).

SHE J, IP HSS, GUAN Y, CHEN D, MOON HB, ROOS A, WANG D, ZHANG J, BERGER ML and SHAW SD (2016) Levels, trends, and health effects of dioxins and related compounds in aquatic biota. In: ALAEE M, *Dioxin and related compounds: the handbook of environmental chemistry*, vol. 49, Springer, Cham, Switzerland.

SHEIKHBARDSIRI H, YARMOHAMMADIAN MH, REZAEI F and MARACY MR (2017) Rehabilitation of vulnerable groups in emergencies and disasters: a systematic review. World Journal of Emergency Medicine **8 (4)** 253-263.

SHIKLOMANOV IA (2000) The world's water resources. Proceedings of the International Symposium to Commemorate 1991 **25** 93-126.

SIKORA MJ, RAE JM, JOHNSON MD and DESTA Z (2010) Efavirenz directly modulates the oestrogen receptor and induces breast cancer cell growth. HIV Medicine **11** 603-607.

SIMONS DB and ŞENTÜRK F (1992) *Sediment transport technology: water and sediment dynamics*, Water Resources Publication, Littleton, Colorado, USA.

SLATER R (2002) Differentiation and diversification: changing livelihoods in QwaQwa, South Africa, 1970-2000. Journal of Southern African Studies **28 (3)** 599-614.

SMITHERS JC and SCHULZE RE (1995) *ACRU agrohydrological modelling system, user manual: version 3.0.* WRC report no. TT70/95, Water Research Commission, Pretoria, South Africa.

SOMMAGGIO LRD, MAZZEO DEC, PAMPLONA-SILVA MT and MARIN-MORALES MA (2018) Evaluation of the potential agricultural use of biostimulated sewage sludge using mammalian cell culture assays. Chemosphere **199** 10-15.

SOUTH AFRICAN NATIONAL BIODIVERSITY INSTITUTE (2012a) *Vegetation map app [Vector]*. Available for download from http://bgis.sanbi.org. (Accessed 25th February 2021).

SOUTH AFRICAN NATIONAL BIODIVERSITY INSTITUTE (2012b) Vegetation map of South Africa, Lesotho and Swaziland, South African National Biodiversity Institute, NVM2012beta2_wgs84_Geo. [Dataset], SANBI, Pretoria, South Africa.

SOUTH AFRICAN NATIONAL STANDARDS (2015) SANS 241-1 Drinking water. Part 1: microbiological, physical, aesthetic and chemical determinants, 2nd ed., South African Bureau of Standards, Pretoria, South Africa.

STATISTICSSOUTHAFRICA(2011)2011census.http://www.statssa.gov.za/publications/P03014/P030142011.pdf.(Accessed 15th March 2022).

STATISTICS SOUTH AFRICA (2012) *Maluti a Phofung.* https://www.statssa.gov.za/?page_id=993&id=maluti-a-phofung-municipality. (Accessed 15th March 2022).

STEFFEN H, BOSCH C, WOLFAARDT G and BOTHA A (2022) Rising environmental temperatures and polluted surface waters: the prelude to the rise of mycoses in South Africa. Water SA **48 (2)** 199-216.

STOFFERAHN CW (2012) Community capitals and disaster recovery: Northwood ND recovers from an EF 4 tornado. Community Development **43 (5)** 581-598.

STONE W (2000) Social capital and social security: lessons from research. Family Matters 57 10-14.

STRYDOM S and SAVAGE MJ (2016) A spatio-temporal analysis of fires in South Africa. South African Journal of Science **112 (11-12)** 1-8.

SUÁREZ AE, GUTIÉRREZ-MONTES I, ORTIZMOREA FA, ORDOÑEZ C, SUÁREZ JC and CASANOVES F (2021) Dimensions of social and political capital in interventions to improve household well-being: implications for coffee-growing areas in southern Colombia. PLOS One **16 (1)** e0245971.

SUTTON J (2010) *Strengthening community resiliency: new media, social networks and the critical infrastructure.* http://www.jeannettesutton.com/uploads/Strengthening_Community_Resiliency.pdf. (Accessed 24th June 2021).

SZKLAREK S, KIEDRZYŃSKA E, KIEDRZYŃSKI M, MANKIEWICZ-BOCZEK J, MITSCH WJ and ZALEWSKI M (2021) Comparing ecotoxicological and physicochemical indicators of municipal wastewater effluent and river water quality in a Baltic Sea catchment in Poland. Ecological Indicators **126** 107611.

TANNER JL, MANTEL S, PAXTON B, SLAUGHTER A and HUGHES D (2022) Impacts of climate change on rivers and biodiversity in a water-scarce semi-arid region of the Western Cape, South Africa. Frontiers in Water **4** 143.

TAYLOR SJ (2023) Exploring the establishment of an urban forest in Phuthaditjhaba to create a more sustainable future urban environment. In: *Sustainable futures in Southern Africa's mountains: multiple perspectives on an emerging city*, MEMBRETTI A, TAYLOR SJ and DELVES JL, Springer International Publishing, Cham, Switzerland.

THE WATER PROJECT (2021) *Water in crisis – South Africa.* https://thewaterproject.org/water-crisis/water-in-crisis-south-africa. (Accessed 23rd March 2021).

THOMPSON JR, GOSLING SN, ZAHERPOUR J and LAIZÉ CLR (2021) Increasing risk of ecological change to major rivers of the world with global warming. Earth's Future **9 (11)** 2021EF002048.

THOMPSON MP, FREEBORN P, RIECK JD, CALKIN DE, GILBERTSON-DAY JW, COCHRANE MA and HAND MS (2016) Quantifying the influence of previously burned areas on suppression effectiveness and avoided exposure: a case study of the Las Conchas fire. International Journal of Wildland Fire **25 (2)** 167-181.

KARRIM A (2020) Department falls short on promise to deliver 5000 water tankers to drought-stricken QwaQwa. *News 24* 5 February 2020.

Thabo Mofutsanyana District Municipality (2021) *Disaster risk assessment 2020/21*. Thabo Mofutsanyana District Municipality, Phuthaditjhaba, South Africa.

ULM FJ and MANAV IB (2021) Climate-resilient infrastructure. *Climate Portal* 20 September 2021. https://climate.mit.edu/explainers/climate-resilient-infrastructure. (Accessed 3rd April 2023).

UNITED NATIONS DEVELOPMENT PROGRAMME (2004) *Reducing disaster risk: a challenge for development*, UNDP, Bureau for Crisis Prevention and Recovery, New York, USA.

UNITED NATIONS DEVELOPMENT PROGRAMME (2010) *Disaster risk assessment*. https://www.undp.org/arab-states/publications/disaster-risk-assessment. (Accessed 25th February 2022).

UNITED NATIONS DEVELOPMENT PROGRAMME (2015) *A new sustainable development agenda: the SDGs.* https://www.undp.org/sustainable-development-goals#:~:text=The%20Sustainable%20 Development%20Goals%20(SDGs)%2C%20also%20known%20as%20the,people%20enjoy%20peac e%20and%20prosperity. (Accessed 25th February 2022).

UNITED NATIONS DEVELOPMENT PROGRAMME (2020) *Guidelines for community participation in disaster recovery*. https://reliefweb.int/report/world/guidelines-community-participation-disaster-recovery. (Accessed 27th July 2022).

UNITED NATIONS DEVELOPMENT PROGRAMME DRYLANDS DEVELOPMENT CENTRE (2014) *Community based resilience analysis (CoBRA) conceptual framework and methodology.* https://www.undp.org/sites/g/files/zskgke326/files/publications/CoBRRA_Conceptual_Framework.pdf. (Accessed 27th July 2022).

UNITED NATIONS EDUCATIONAL, SCIENTIFIC AND CULTURAL ORGANISATION (2015) *The United Nations world water development report 2015: water for a sustainable world*, UNESCO, Paris, France. https://www.unwater.org/publications/un-world-water-development-report-2015. (Accessed 16th April 2019).

UNITED NATIONS ENVIRONMENT PROGRAMME (2007) *Cities and urban vulnerability: in the context of urban environmental management.* https://www.unisdr.org/files/5453_092UNE.pdf. (Accessed 27th July 2022).

UNITED NATIONS ENVIRONMENT PROGRAMME (2013) Africa environment outlook 3: our environment, our health, UNEP, Nairobi, Kenya.

UNITED NATIONS ENVIRONMENT PROGRAMME (2016) A snapshot of the world's water quality: towards a global assessment, UNEP, Nairobi, Kenya.

UNITED NATIONS FRAMEWORK CONVENTION ON CLIMATE CHANGE (2016) *The Paris Agreement*. https://unfccc.int/documents/184656. (Accessed 24th March 2023).

UNITED NATIONS GENERAL ASSEMBLY (2015) *Transforming our world: the 2030 agenda for sustainable development*. https://www.refworld.org/docid/57b6e3e44.html. (Accessed 27th July 2022).

UNITED NATIONS INTERNATIONAL STRATEGY FOR DISASTER REDUCTION (2015) *The Sendai framework for disaster risk reduction 2015-2030*, UNISDR, Geneva, Switzerland.

UNITED NATIONS OFFICE FOR DISASTER RISK REDUCTION (2016) *Terminology*. https://www.undrr.org/terminology (Accessed 10th February 2022).

UNITED NATIONS OFFICE FOR DISASTER RISK REDUCTION (2022) *Understanding disaster risk: poorly planned urban development*. https://www.preventionweb.net/understanding-disaster-risk/risk-drivers/poorly-planned-urban-development (Accessed 27th July 2022).

UNITED NATIONS OFFICE OF THE HIGH COMMISSIONER FOR HUMAN RIGHTS (2022) *Climate change the greatest threat the world has ever faced, UN expert warns.* https://www.ohchr.org/en/press-releases/2022/10/climate-change-greatest-threat-world-has-ever-faced-un-expert-warns. (Accessed 24th March 2023).

UNITED NATIONS SECURITY COUNCIL (2021) *Climate change "biggest threat modern humans have ever faced", world renowned naturalist tells security council, calls for greater global cooperation.* https://press.un.org/en/2021/sc14445.doc.htm. (Accessed 24th March 2023).

UNITED STATES AGENCY FOR INTERNATIONAL DEVELOPMENT (2016) *Climate vulnerability assessment: an annex to the USAID climate-resilient development framework.* https://pdf.usaid.gov/pdf_docs/PA00KZ84.pdf. (Accessed 10th January 2022).

UNITED STATES DEPARTMENT OF AGRICULTURE, SOIL CONSERVATION SERVICE (1972) *National engineering handbook section 4, hydrology*, USDA-SCS, Washington, D.C., USA.

VAN DER LAAN M, VAN ANTWERPEN R and BRISTOW KL (2012) River water quality in the northern sugarcane-producing regions of South Africa and implications for irrigation: a scoping study. Water SA **38 (1)** 87-96.

VAN NIEKERK W and LE ROUX A (2017) Human settlements. In: *Climate risk and vulnerability: a handbook for Southern Africa*, DAVIS-REDDY CL and VINCENT K, 2nd ed., CSIR, Pretoria, South Africa.

VAN VLIET MT, FRANSSEN WH, YEARSLEY JR, LUDWIG F, HADDELAND I, LETTENMAIER DP and KABAT P (2013) Global river discharge and water temperature under climate change. Global Environmental Change **23 (2)** 450-464.

VAN VUREN JHJ, DU PREEZ HH and DEACON AR (1994) *Effects of pollutants on the physiology of fish in the Olifants River (Eastern Transvaal)*. WRC report no. 350/1/94, Water Research Commission, Pretoria, South Africa.

VEZZONE M, CESAR R, DE SOUZA ABESSA DM, SERRANO A, LOURENÇO R, CASTILHOS Z, RODRIGUES AP, PERINA FC and POLIVANOV H (2019) Metal pollution in surface sediments from Rodrigo de Freitas Lagoon (Rio de Janeiro, Brazil): toxic effects on marine organisms. Environmental Pollution **252** 270-280.

VILJOEN G and VAN DER WALT K (2018) South Africa's water crisis – an interdisciplinary approach. Tydskrif vir Geesteswetenskappe **58 (3)** 483-500.

VÖLKER J, CASTRONOVO S, WICK A, TERNES TA, JOSS A, OEHLMANN J and WAGNER M (2016) Advancing biological wastewater treatment: extended anaerobic conditions enhance the removal of endocrine and dioxin-like activities. Environmental Science & Technology **50 (19)** 10606-10615.

VOUA OTOMO P, OWOJORI OJ, REINECKE SA, DANIELS S and REINECKE AJ (2011) Using estimates of metal bioavailability in the soil and genetic variation of allozymes to investigate heavy metal tolerance in the earthworm *Eisenia fetida* (Oligochaeta). Ecotoxicology and Environmental Safety **74** (7) 2070-2074.

WADESON RA and ROWNTREE KM (2001) The application of a hydraulic biotope matrix to the assessment of available habitat: potential application to IFRs and river health monitoring. Southern African Journal of Aquatic Sciences **26 (1)** 67-73.

WAGNER M (2019) *Summer school South Africa* — *chair of hydrology* — *TU Dresden.* https://tudresden.de/bu/umwelt/hydro/ihm/hydrologie/forschung/summerschool-south-africa. (Accessed 15th April 2019).

WAGNER M, VERMEIRSSEN EL, BUCHINGER S, BEHR M, MAGDEBURG A and OEHLMANN J (2013) Deriving bio-equivalents from in vitro bioassays: assessment of existing uncertainties and strategies to improve accuracy and reporting. Environmental Toxicology and Chemistry **32 (8)** 1906-1917.

WANG D, CHEN Y, JARIN M and XIE X (2022) Increasingly frequent extreme weather events urge the development of point-of-use water treatment systems. NPJ Clean Water **5** (1) 36.

WARBURTON-TOUCHER ML, RAMJEAWON M, MCNAMARA MA, ROUGET M, BULCOCK H, KUNZ RP, MOONSAMY J, MENGISTU M, NAIDOO T, VATHER T and ALDWORTH TA (2019) *Resetting the baseline land cover against which streamflow reduction activities and the hydrological impacts of land use change are assessed.* WRC report no. 2437/1/19, Water Research Commission, Pretoria, South Africa.

WHITEHEAD PG, WILBY RL, BATTARBEE RW, KERNAN M and WADE AJ (2009) A review of the potential impacts of climate change on surface water quality. Hydrological Sciences Journal **54 (1)** 101-123.

WILLIAMS JR (1975) Sediment-yield prediction with universal equation using runoff energy factor. In: *Present and prospective technology for predicting sediment yield and sources*, UNITED STATES DEPARTMENT OF AGRICULTURE: AGRICULTURAL RESEARCH SERVICE, United States Department of Agriculture, Washington, D.C., USA.

WISCHMEIER WH and SMITH DD (1978) *Predicting rainfall erosion losses: a guide to conservation planning (no. 537)*, United States Department of Agriculture, Science and Education Administration, Washington, D.C., USA.

WISNER B, BLAIKIE PM, CANNON T and DAVIS I (2004) *At risk: natural hazards, people's vulnerability and disasters*, 2nd ed., Routledge, London, UK.

WOLSKI P, CONRADIE S, JACK C and TADROSS M (2021) Spatio-temporal patterns of rainfall trends and the 2015-2017 drought over the winter rainfall region of South Africa. International Journal of Climatology **41** E1303-E1319.

WOLSKI P, LUMSDEN TG, KUNZ RP, CLARK DJ, SCHÜTTE S and SCHULZE RE (2022) Establishing climate scenario projections. In: *A national assessment of potential climate change impacts on the hydrological yield of different hydro-climatic zones of South Africa*, SCHÜTTE S, SCHULZE RE and CLARK DJ, WRC report no. 2833, Water Research Commission, Pretoria, South Africa.

WONDIM YK (2016) Flood hazard and risk assessment using GIS and remote sensing in lower Awash sub-basin, Ethiopia. Journal of Environment and Earth Science **6** 69-85.

WONGBUSARAKUM S and LOPER C (2011). *Indicators to assess community level social vulnerability to climate change: an addendum to SocMon and SEM-Pasifika regional socioeconomic monitoring guidelines (first draft)*. https://reefresilience.org/pdf/SocMon_Climate_change_guidlelines_FINAL_april_2011.pdf. (Accessed 24th March 2023).

WOODFORDC(2019)Waterpollution:anintroduction.https://www.explainthatstuff.com/waterpollution.html. (Accessed 6th September 2019).

WOODWARD G, BONADA N, BROWN LE, DEATH RG, DURANCE I, GRAY C, HLADYZ S, LEDGER ME, MILNER AM, ORMEROD SJ and THOMPSON RM (2016) The effects of climatic fluctuations and extreme events on running water ecosystems. Philosophical Transactions of the Royal Society B: Biological Sciences **371 (1694)** 20150274.

WORLD BANK (2021) *Climate risk country profile: South Africa*, World Bank Group, Washington, D.C., USA.

WORLD ECONOMIC FORUM (2020) *Climate change hits the poor hardest. Here is how to protect them.* https://www.weforum.org/agenda/2020/10/climate-change-poor-hardest-how-protect-them/. (Accessed 24th March 2023).

WORLD ECONOMIC FORUM (2023) *The global risks report 2019*, 18th ed., WEF, Geneva, Switzerland. https://www3.weforum.org/docs/WEF_Global_Risks_Report_2023.pdf. (Accessed 24th March 2023). WORLD WILDLIFE FUND (2016) *Water: facts and figures.* https://www.wwf.org.za/our_research/publications/?25181/Water-Facts-and-Futures (Accessed 24th March 2023).

WU Z, ZHANG D, CAI Y, WANG X, ZHANG L and CHEN Y (2017) Water quality assessment based on the water quality index method in Lake Poyang: the largest freshwater lake in China. Scientific Reports **7** 17999.

YAN DH, WANG G, WANG H and QIN TL (2012) Assessing ecological land use and water demand of river systems: a case study in Luanhe River, North China. Hydrology and Earth System Sciences **16** 2469-2483.

YANG K, LEJEUNE J, ALSDORF D, LU B, SHUM CK and LIANG S (2012) Global distribution of outbreaks of water-associated infectious diseases. PLOS Neglected Tropical Diseases **6 (2)** e1483.

ZHUWAKINYU M (2012) *Water 2012: a review of South Africa's water sector.* http://pmg-assets.s3-website-eu-west-1.amazonaws.com/120904review.pdf. (Accessed 12th April 2019).

ZIERVOGEL G, JOHNSTON P, MATTHEW M and MUKHEIBIR P (2010) Using climate information for supporting climate change adaptation in water resource management in South Africa. Climatic Change **103 (3)** 537-554.

ZIERVOGEL G, JOHNSTON PA, WALAWEGE R, MATTHEW M and MUKHEIBIR P (2008) *Using enhanced knowledge of climate variability for the benefit of water resource management*. WRC report no. 1566/1/09, Water Research Commission, Pretoria, South Africa.

ZIERVOGEL G, NEW M, ARCHER VAN GARDEREN E, MIDGLEY G, TAYLOR A, HAMANN R, STUART-HILL S, MYERS J and WARBURTON M (2014) Climate change impacts and adaptation in South Africa. Wiley Interdisciplinary Reviews: Climate Change **5** (5) 605-620.

APPENDIX A: PHOTOGRAPHS OF THE NINE SAMPLING SITES ON THE NAMAHADI, METSI-MATSO, MPHUKOJWANE, AND ELANDS RIVERS

Plate 1: Namahadi River

Namahadi Upper site



Photo 1: Namahadi Upper: Google Earth image (4.39 km)

Namahadi Middle site



Photo 2: Namahadi Middle: Google Earth image (4.1 km)



Photo 3: Namahadi Upper: Upstream view



Photo 4: Namahadi Middle: Upstream view



Photo 5: Namahadi Upper: Downstream view



Photo 6: Namahadi Middle: Downstream view

Plate 2: Namahadi River

Namahadi Lower site



Photo 1: Namahadi Lower: Google Earth image (8.12 km)



Photo 2: Namahadi Lower: Upstream view



Photo 3: Namahadi Lower: Downstream view

Plate 3: Metsi-Matso River

Metsi-Matso Upper site



Photo 1: Metsi-Matso Upper: Google Earth image (6.6 km)

Metsi-Matso Middle site



Photo 2: Metsi-Matso Middle: Google Earth image (6.26 km)



Photo 3: Metsi-Matso Upper: Upstream view



Photo 4: Metsi-Matso Middle: Upstream view



Photo 5: Metsi-Matso Upper: Downstream view



Photo 6: Metsi-Matso Middle: Downstream view

Plate 4: Mphukojwane River

Mphukojwane Upper site



Photo 1: Mphukojwane Upper: Google Earth image (3.9 km)





Photo 2: Mphukojwane Middle: Google Earth image (3.2 km)



Photo 3: Mphukojwane Upper: Upstream view



Photo 4: Mphukojwane Middle: Upstream view



Photo 5: Mphukojwane Upper: Downstream view



Photo 6: Mphukojwane Middle: Downstream view

Plate 5: Elands River

Elands Upper site



Photo 1: Elands Upper: Google Earth image (11.79 km)

Elands Lower site



Photo 2: Elands Lower: Google Earth image (9.68 km)



Photo 3: Elands Upper: Upstream view



Photo 4: Elands Lower: Upstream view



Photo 5: Elands Upper: Downstream view



Photo 6: Elands Lower: Downstream view

APPENDIX B: LIST OF POTENTIAL POLLUTANTS, POTENTIAL SOURCES,

Micropollutants	Potential source(s)	Potential risk/effect
Acetochlor	Herbicide for weed control	 Potential health effects to humans and aquatic biota at high concentrations
Atenolol	Pharmaceutical	 Ecotoxicology test indicates toxicity to aquatic biota Atenolol is water soluble and shows rapid degradation in and aquatic environment
Atrazine	Herbicide for weed control	 Chronic exposure may increase cancer risk Consider in combination with pH and microbial activity
BPA	 Industrial chemical WWTP effluent 	 Potential endocrine disruptor Very low acute toxicity for humans Considered to be an environmental contaminant of emerging concern for a wide range of wildlife, especially aquatic invertebrates and amphibians
Benzophenone	 Industrial chemical ultraviolet (UV) blocker, photo initiator, packaging, etc.) 	 Potential negative impact on reproduction of some aquatic organisms
BP-4	 UV filter WWTP effluent; water recreation 	 Potential negative impact on the reproduction of some aquatic organisms
Caffeine	WWTP effluentUntreated sewage	 Could be seen as an indicator of untreated sewage Environmental contaminant – residues are detrimental to aquatic organisms
Carbamazepine	PharmaceuticalWWTP effluent	 Persistent contaminant in the environment Potential impact on embryonic and larval development of fish
DEET	 Insect repellent 	 Slightly toxic to birds, fish, and aquatic invertebrates
Diclofenac	Veterinary pharmaceuticalWWTP effluent	 Average toxicity and high environmental risk Synergistic effects between diclofenac and sulfamethoxazole
Efavirenz	PharmaceuticalWWTP effluent	Persistent environmental contaminant
Fipronil	Insecticide	Highly toxic to fish and aquatic invertebrates
Ibuprofen	 Pharmaceutical WWTP effluent 	 Environmental contaminant toxic to a range of aquatic organisms, influencing, for example, fish reproduction Metabolites from ibuprofen are more toxic than the parent molecule
Nevirapine	PharmaceuticalWWTP effluent	 Persistent environmental contaminant Potential human health effects Potential effect on growth of juvenile fish
Sulfamethoxazole	PharmaceuticalWWTP effluent	 Low toxicity with low environmental risk Synergistic effects between diclofenac and sulfamethoxazole

AND ASSOCIATED RISKS

Metal elements	Potential source(s)	Potential risk/effect
Al	Silicates of Al naturally	Chronic health issues at high concentrations
	widespread	• Consider in combination with pH and fluoride
	Runoff from catchment	concentration
	Present in water as	
	suspended Al minerals /	
	dissolved AI species	
	• Sewage/WWTPs – Al	
	phosphate is present in	
	sewage	
	 Industrial waste 	
Cr	• Cr(VI) is highly soluble and	• Long-term exposure increases risk of cancer
	mobile in the environment;	• Evidence of carcinogenic effects via the oral
	can move though soil profile	route
	into groundwater	
	 Industrial processes (e.g. 	
	paints, dyes, metal pickling,	
	ceramics, etc.)	
Cu	 Runoff from catchment 	 Generally no health effects, but high
		concentrations may cause gastrointestinal
		irritation
Fe	 Naturally abundant 	 Chronic health effects only at high
		concentrations
		Consider in combination with pH, redox
		potential, turbidity, suspended matter, Al
		concentration, and Mn
Mn	 Naturally abundant 	Health effects only at high concentrations
	Industrial processes	Consider in combination with pH, redox
		potential, turbidity, suspended matter, and Al
7	D (11)	
Zn	Runoff from catchment	• Fish highly susceptible to Zn poisoning
La citta ta ta	Industrial processes	
In vitro tests	Potential source(s)	Potential risk/effect
YES (Yeast	Birth control pills and other	• Estrogen is an endocrine disruptor with
estrogen	medication in wastewater	potential adverse effects on numans and
screen)	Natural oestrogens from soy	aquatic biota; e.g. fish
	and dairy products, as well	
		- Diavina ara hishlu tavia may agusa
Yeast Dioxin	Industrial processes	Dioxins are highly loxic; may cause repreductive and developmental problems
test)	Incineration of municipal	interfere with immune system and hormones
(631)	waste	and are carcinogenic
	Leaching from landilis	Negative health impact on aquatic hiota: e.g.
	• Stormwater runoii	in fish it may cause skin lesions and altered
		liver function
		Bio-accumulation in aquatic biota – trophic
		transfer in food web

Water quality	Potential source(s)	Potential risk/effect
Sulphates	 Naturally present in water, soil, and rocks Industrial processes 	• Human health effects; e.g. diarrhoea
Chloride	Anion of chlorine	 Accelerate corrosions rate in metals High concentrations may be detrimental to the health of babies
Nitrates	 Naturally present in water and soil Oxidation of organic material Effluent from WWTPs Agricultural and urban runoff 	 Increased algal growth and blooms Uptake of nitrates by plants regulated by temperature and pH Human health effects
Faecal coliforms and <i>E. coli</i>	 Faeces from humans and animals Untreated sewage 	 Waterborne diseases; e.g. diarrhoea Organic pollution – decomposition depletes oxygen in water Aesthetic effects; e.g. bad odours Reduces recreational and cultural uses of streams

 Streams

 Source: Adeola and Forbes (2022); Arnold *et al.* (2013); Arslan *et al.* (2017); Asadgol *et al.* (2019); Batucan *et al.* (2021); Chopra and Kumar (2020); Clara *et al.* (2004); Drzymala and Kalka (2020); Du *et al.* (2017); Joachim *et al.* (2021); Krahnstöver *et al.* (2022); Li *et al.* (2020); Ndlovu and Naidoo (2022); Nibamureke *et al.* (2019); Owen (2017); She *et al.* (2016); Van Vuren *et al.* (1994)

APPENDIX C: RESULTS FROM THE PHYSICO-CHEMICAL, ECOTOXICOLOGICAL AND RIVER HEALTH ASSESSMENTS OF THE NINE SAMPLING SITES ON THE NAMAHADI, METSI-MATSO, MPHUKOJWANE, AND ELANDS RIVERS

Site	ELL	ELU	MPM	MPU	NL	NM	NU	ММ	MU
Selected physical and chemical water quality parameters									
System variables	s (March 2	2022)							
Temp. (°C)	17.6	16.9	20.4	17.6	21.1	18.5	17.2	22.3	19.8
EC (mS/cm)	179.0	237.0	477.0	201.0	90.6	76.5	61	83.0	16.0
рН	7.7	7.6	6.9	8.4	7.6	7.2	7.8	6.8	6.9
COD	25.2	34.1	28.2	19.8	16.8	19.65	16.2	23.3	19.9
TOC	38.0	40.5	46.5	35.5	31.0	32.5	32.5	37.5	30
DOC (mg/L)	3.22	5.77	5.1	1.58	1.46	2.18	0.81	1.59	1.73
SAC ₂₅₄ (1/m)	12.02	18.32	14.4	5.79	4.19	8.96	2.47	5.6	5.00
Nutrients and an	ions (Mar	ch 2022)						-	
Phosphate (mg/L)	1.33	0.28	0.28	0.84	0.42	0.31	0.78	0.28	0.3
Carbonate (mg/L)	5.0	4.8	4.55	4.6	5.5	5.15	5.15	5.25	5.3
Chloride (mg/L)	12.7	18.2	41.2	0.85	2.95	5.65	0.95	1.0	0.9
Fluoride (mg/L)	0.055	0.076	0.074	0.058	0.031	0.035	<0.020	<0.020	<0.020
Chloride (mg/L)	12.70	18.20	41.20	0.85	2.95	5.65	0.95	1.00	0.85
Nitrate (mg/L)	25.4	31.7	25.9	25.3	23.6	30.8	43.6	25.9	25.9
Nitrite (mg/L)	<0.02	<0.02	0.037	<0.02	<0.02	<0.02	<0.02	<0.02	<0.003
Bromide (mg/L)	0.046	0.061	0.092	0.026	0.029	0.037	<0.026	0.028	0.026
Sulphate (mg/L)	12.05	18.37	39.4	9.15	4.83	5.58	3.01	2.21	1.44
Total	1.33	0.28	0.28	0.84	0.42	0.31	0.78	0.28	0.28
phosphorus (mg/L)									
Soluble reactive	<0.007	<0.007	<0.007	<0.007	<0.007	<0.007	<0.007	<0.007	<0.007
phosphorous									
(mg/L)									
Metal elements									
Al		0.148	0.194	0.172	0.169	0.105	-	-	
As		0.001	-	-	0.001	-	-	-	
Ca		26	23	20	18	16	8	7	
Cu		-	0.004	0.002	0.001	-	0.006	0.002	
Mg		9	7	10	8	6	3	3	
Mn		0.610	0.124	0.032	-	0.044	-	-	
Fe		2.140	0.491	0.295	0.462	0.320	0.063	0.428	
Zn		0.026	0.036	0.028	0.027	0.024	0.030	0.023	
Microbial data									
Faecal coliform		34 00	340	42	220	39	9	48	
bacteria (MPN/100 ml)		0							
E. coli		21 00	240	37	190	35	9	46	
(MPN/100 ml)		0							

	ELL	ELU	MPM	MPU	NL	NM	NU	MM	MU
OMPs									
BP (ng/L)	20	22	22	27	18	18	15	14	10
BP-4 (ng/L)	199	257	262	<10	<10	14	<10	<10	<10
Ibuprofen (ng/L)	71	90	29	<10	<10	<10	<10	<10	<10
BPA (ng/L)	68	<10	<10	<10	<10	<10	<10	<10	<10
Diclofenac (ng/L)	63	56	66	68	65	71	77	62	75
Fipronil (ng/L)	<10	45	<10	<10	<10	<10	<10	<10	<10
DEET (ng/L)	<10	11	<10	23	<10	<10	18	<10	<10
Caffeine (ng/L)	173	217	133	39	34	22	47	68	22
Carbamazepine	38	50	142	<10	<10	<10	<10	<10	<10
Sulfamethoxazole (ng/L)	62	60	97	<10	14	25	<10	<10	<10
Atenolol (ng/L)	20	30	18	<10	<10	<10	<10	<10	<10
Acetochlor (ng/L)	404	<10	<10	<10	<10	<10	<10	<10	<10
8-OH-efavirenz (ng/L)	135	195	125	<10	<10	<10	<10	<10	<10
Efavirenz (ng/L)	126	100	177	<10	<10	<10	<10	<10	<10
Nevirapine (ng/L)	10	<10	29	<10	<10	<10	<10	<10	<10
Atrazine (ng/L)	<10	<10	<10	<10	<10	<10	<10	<10	<10

	ELL	ELU	MPM	MPU	NL	NM	NU	MM	MU
Ecotoxicological tests									
Toxicity of river water									
P. subcapitata	146.6	N/A	34.0	ND	ND	N/A	ND	ND	ND
(LC50)									
D. magna	80.0	80.0	80.0	80.0	80.0	80.0	30.0	90.0	30.0
T. thermophila	13.2	13.9	31.5	31.1	31.2	100	33.8	34.5	27.8
Class score	51.1	47.0	47.0	55.6	55.6	90.0	13.9	62.3	28.9
Percentage effect	63.9	58.8	58.8	69.5	69.5	90.0	41.1	69.2	96.0
Toxicity class						IV	II		
In vitro tests									
LOQ YES in ng	-	0.6	0.6	0.6	0.6	0.6	0.6	0.6	0.6
E·L ⁻¹									
LOQ YAES in µg	-	5180	5180	5180	5180	5180	518	5180	5180
OHT L ⁻¹							0		
LOQ YAS ng T·L ⁻¹	-	3	3	3	3	3	3	3	3
YAAS µg Flu∙L ⁻¹	-	173	173	173	173	173	173	173	173
YDS µg ß-NF·L ⁻¹	-	0.1	0.1	0.1	0.1	0.1	0.1	0.1	0.1

	ELL	ELU	MPM	MPU	NL	NM	NU	MM	MU
River health									
HI Index score (%)		42.0	52.0	60.0	35.0	64.0	63.0	59.0	35.0
(Integrity category)		(D)	(D)	(C)	(E)	(C)	(C)	(D)	(E)
Riparian HI score		45.0	43.0	61.0	28.0	55.0	65.0	55.0	40.0
(%) (Integrity		(D)	(D)	(C)	(E)	(D)	(C)	(D)	(D)
category)									
FRAI score (%)		9.8 (F)	51.9	51.9	61.9	62.7	73.8	57.5	
(Present ecological			(D)	(D)	(C/D)	(C)	(C)	(C/D)	
status category)									

APPENDIX D: PHOTOGRAPHS OF PROMINENT ANTHROPOGENIC IMPACTS ON THE INSTREAM AND RIPARIAN ZONES AT THE SAMPLING SITES AS OBSERVED DURING THE APRIL 2021 FIELD VISIT

Instream and riparian impacts



Photo 1: Dam wall at the Upper Metsi-Matso causing serious modification to flow regime



Photo 2: River bank erosion at Metsi-Matso Middle



Photo 3: Exotic black wattle trees at Upper Namahadi



Photo 5: Road bridge at Namahadi Middle



Photo 4: River sand mining at Namahadi Middle (right bank)



Photo 6: Solid waste disposal (diapers) at Mphukojwane Upper

Instream and riparian impacts



Photo 7: Bank erosion at Mphukojwane Upper



Photo 8: Exotic vegetation encroachment into the channel, Mphukojwane Middle



Photo 9: Exotic vegetation encroachment into the channel, Mphukojwane Middle



Photo 10: Solid waste disposal upstream of Mphukojwane Middle



Photo 11: Road bridge at Elands Lower



Photo 12: Exotic willow trees at Elands Lower